

The effects of anthracyclines on calcium handling and contractility in sheep ventricular myocytes; role of oxidative stress

Amy Foster

School of Science, Engineering and Environment

University of Salford

Supervisor: Dr David Greensmith

Submitted in partial fulfilment of the requirements of the Degree of Doctor of Philosophy

2023

List of Contents

Table of figu	resx
Table of Tab	lesxii
Acknowledg	ements xiii
Confirmatio	n of ethical approval xiv
List of Abbre	viations xv
Abstract	xviii
Chapter 1 - C	Seneral Introduction1
1.1 Can	cer2
1.1.1	Childhood cancer2
1.1.2	Aetiology of cancer3
1.1.2.	1 Hallmarks of cancer
1.1.2.	2 Aetiology of childhood cancer4
1.1.3	Cancer diagnosis5
1.1.4	Treatment of childhood cancer6
1.2 Car	diac excitation-contraction coupling7
1.2.1	The human heart7
1.2.2	The role of calcium7
1.2.3	Excitation8
1.2.3.	1 The source of excitation8
1.2.3.	2 The cardiac cellular action potential9
1.2.4	Sarcolemma calcium entry10
1.2.4.	L-type voltage gated calcium channel activation10
1.2.4.	L-type voltage gated calcium channel inactivation
1.2.5	The sarcoplasmic reticulum13
1.2.6	Calcium-induced calcium release13
1.2.6.	1 The ryanodine receptor13
1.2.6.	2 The dyad14
1.2.6.	3 The process of calcium induced calcium release15
1.2.7	Contraction15
	i

1.2.7.	.1	Contractile apparatus	15
1.2.8	Rela	axation	17
1.2.8.	.1	The need for calcium removal	17
1.2.8.	.2	Sarco-endoplasmic reticulum calcium transport ATPase	18
1.2.8.	.3	Sodium- calcium exchanger	19
1.2.8.	.4	Other calcium removal processes	20
1.2.8.	.5	The calcium transient	21
1.2.9	Con	trol of inotropy	21
1.2.9.	.1	The dependence of contractile force on systolic Ca	22
1.2.9.	.2	The dependence of systolic Ca on sarcoplasmic reticulum Ca content	22
1.2.9.	.3	The dependence of systolic Ca on trigger Ca	23
1.2.9.	.4	Regulation of SR Ca content	24
1.2.9.	.5	β-adrenergic stimulation	24
1.2.10	Aut	oregulation	26
1.3 Sur	nmar	y of Excitation-contraction coupling	27
1.4 Rea	active	e oxygen species and oxidative stress	28
1.4.1	Rea	ctive oxygen species	28
1.4.1.	.1	Physiological sources of reactive oxygen species	28
1.4.1.	.2	The physiological fate of ROS	29
1.4.2	Oxi	dative stress and disease	30
1.4.3	The	effects of oxidative stress of excitation contraction coupling	31
1.4.4	Mea	asuring oxidative stress	32
1.5 Ant	thrac	yclines	34
1.5.1	Ant	hracycline chemical structure	34
1.5.2	Me	chanism of action of anthracyclines	34
1.5.2.	.1	Topoisomerase type 2 poisoning/inhibition	35
1.5.2.	.2	Free radical production	35
1.5.2.	.3	Altered intracellular signalling	36
1.6 Ant	thrac	ycline-induced cardiotoxicity	37
1.6.1	Clin	ical manifestation of anthracycline-induced cardiotoxicity	37

	163	,	Acute and long term anthracycling induced cardiotoxicity	20
	1.0.2	<u>×</u>		. 30
	1.6.5	3	Adult versus childhood anthracycline-induced cardiotoxicity	.39
	1.6.4	1	The mechanisms of anthracycline-induced cardiotoxicity	.40
	1.6.5	5	Prevention of anthracycline induced heart failure	.41
1.	7	Stuc	dy main aims	.43
Cha	pter	2 - G	eneral Methodology	.45
2.	1	The	use of animal models to study excitation-contraction coupling	.46
2.	2	Prin	nary myocyte isolation	.47
	2.2.1	L	Ethics	.47
	2.2.2	2	Tissue preparation and cell isolation	.47
2.	3	Expe	erimental solutions	.50
2.	4	Prep	paration of anthracyclines	.50
	2.4.1	L	Control of experimental cell environment / perfusion	.51
2.	5	Field	d stimulation of cardiomyocytes	.51
2.	6	Mea	asurement of intracellular calcium	.52
	2.6.1	L	Equipment configuration	.52
	2.6.2	2	Loading of cardiomyocytes with Ca fluorophore	.53
	2.6.3	3	Excitation and emission protocol	.53
	2.6.4	1	Epi-fluorescence microscope configuration for Ca measurement	.54
2.	7	Mea	asurement of sarcomere length	.56
2.	8	Data	a Analysis for calcium and sarcomere length	.57
	2.8.1	L	Diastolic and systolic Ca	.57
	2.8.2	2	Sarcomere length	.57
	2.8.3	3	Statistical Analysis for Ca handling and contractility	.57
2.	9	Mea	asurement of intracellular oxidative stress	.58
	2.9.1	L	Loading and preparation of cardiomyocytes for oxidative stress experiments	58
	2.9.2	2	Image capturing and analysis	.58
	2.9.3	3	Statistical analysis for oxidative stress data	.59
Chai	pter	3 - T	he effects of doxorubicin and daunorubicin on Ca handling and contractility	y in
shee	ep ve	ntri	cular myocytes	.60

3.1	Intr	oduction	.61
3.	1.1	Anthracyclines and excitation-contraction coupling	.61
3.	1.2	An experimentally relevant anthracycline concentration	.63
3.	1.3	Summary	.64
3.	1.4	Aims	.64
3.2	Me	thods	.64
3.	2.1	Cell preparation	.64
3.	2.2	Assessment of cell suitability	.65
3.	2.3	Excitation-contraction coupling measurements	.65
3.	2.4	Measurement of myofilament sensitivity	.65
3.	2.5	Measurement of sarcoplasmic reticulum Ca content	.67
3.	2.6	Measurement of the activity of the Ca removal mechanisms	.67
3.3	Res	ults	.67
3.	3.1	The effects of doxorubicin on cell suitability	.67
3.	3.2	The dose-dependent effect of doxorubicin on intracellular Ca and contraction 68	ility
3.	3.3	A detailed investigation of the effects of DOX on Ca and contractility	.70
3.	3.4	The effect of doxorubicin on SR Ca content and rate of Ca removal	.73
3.	3.5	A detailed investigation of the effects of daunorubicin on Ca and contraction 75	ility
3.	3.6	The effect of daunorubicin on SR Ca content	.79
3.	3.7	The apparent interaction of daunorubicin and caffeine	.80
3.	3.8	The effects of doxorubicin and daunorubicin on myofilament sensitivity	.82
3.4	Dise	cussion	.84
3.	4.1	What is an experimentally relevant anthracycline concentration?	.84
3. ca	4.2 Irdiom	Does 1 nM doxorubicin alter intracellular Ca handling and contractility in she yocytes?	eep 85
3.	4.3	Is SR Ca altered by 1 nM doxorubicin?	.87
3.	4.4	Does doxorubicin alter the activity of Ca removal mechanisms?	.87
3. ca	4.5 Irdiom	Does 1 nM daunorubicin alter intracellular Ca handling and contractility in she yocytes?	eep 89

3.4.6	Is SR Ca altered by 1 nM daunorubicin?89
3.4.7	Does daunorubicin alter the activity of Ca removal mechanisms?90
3.4.8	Do doxorubicin and daunorubicin alter myofilament sensitivity?91
3.5 Coi	ncluding remarks92
Chapter 4 -	Measurement of anthracycline-induced oxidative stress in sheep ventricular
myocytes	
4.1 Int	roduction94
4.1.1	ROS production in cardiac myocytes94
4.1.2	Anthracyclines and ROS95
4.1.3	Oxidative stress and calcium handling95
4.1.4	Antioxidants96
4.1.5	Aims97
4.2 Me	thodology
4.2.2	Cell preparation98
4.2.3	Oxidative stress measurements in cardiac myocytes98
4.2.3	1 Preparation of cells to measure oxidative stress
4.2.3	2 Measurement of oxidative stress in cardiac myocytes
4.2.4	Excitation-contraction coupling measurements99
4.2.5	Sarcoplasmic reticulum Ca content99
4.3 Res	sults
4.3.1	Validation of method sensitivity
4.3.2	The effect of DOX and DAUN on cytoplasmic oxidative stress101
4.3.3	The attenuation of DOX-induced OS by N-acetyl cysteine pre-treatment102
4.3.4 NAC	The effects of DOX on global Ca and contractility following pre-treatment with 103
4.3.5 oxidativ	The dependence of DOX-mediated changes to global Ca and contractility on ve stress
4.3.6 treatme	The effects of DOX on SR Ca content and cytoplasmic Ca removal following pre- ent with NAC108
4.3.7 remova	The dependence of DOX-mediated changes to SR Ca and cytoplasmic Ca l on oxidative stress

4.4 D	iscussion111
4.4.1	s this method sensitive enough to detect changes in oxidative stress?111
4.4.2	Does DOX increase levels of oxidative stress in sheep ventricular myocytes? 111
4.4.3	Does DAUN increase levels of oxidative stress in sheep ventricular myocytes? 112
4.4.4 deper	To what extent are the effects of DOX on Ca handling and contractility ident on oxidative stress?
4.4.5	To what extent are the effects of DOX on SR Ca dependent on oxidative stress? 114
4.4.6 on oxi	To what extent are the effects of DOX on Ca removal mechanisms dependent dative stress?114
4.5 C	oncluding remarks115
Chapter 5 oxidative	- The role of xanthine oxidase and NADPH oxidase in anthracycline-induced stress
5.1 lr	ntroduction117
5.1.1	The mechanisms of increased oxidative stress by anthracyclines117
5.1.2	ROS inhibition117
5.1.3	Xanthine oxidase inhibition118
5.1.4	NADPH oxidase inhibition119
5.1.5	Summary119
5.1.6	Aims119
5.2 N	1ethods120
5.2.1	Cell preparation and excitation-contraction coupling measurements120
5.2.2	Reactive oxygen species inhibitor preparation and treatment120
5.2.3 remov	Measurement of sarcoplasmic reticulum Ca content and the activity of Ca val mechanisms
5.2.4	Measurement of myofilament sensitivity121
5.3 R	esults
5.3.1 stress	The effect of inhibition of XO and NOX2 and NOX4 on DOX-induced oxidative 122

	5.3.2 contract	The effect of xanthine oxidase inhibition on the alterations to Ca handling ar ility produced by DOX	nd 23
	5.3.3 produce	The effect of NOX inhibition on the alterations to Ca handling and contractili d by DOX12	ty 26
	5.3.4 and con	The impact of NOX an XO inhibition on DOX-induced alterations to Ca handlin tractility12	ng 29
	5.3.5 myofilar	The effect of NOX and XOR inhibition on DOX-mediated alterations nent sensitivity	to 31
5	.4 Diso	cussion	33
	5.4.1	Does inhibition of XO and NOX2 and NOX4 reduce anthracycline-induced RO 133	S?
	5.4.2 contract	Does xanthine oxidase inhibition alter the effect of DOX on Ca handling ar ility?	nd 35
	5.4.3 contract	Does NOX2 and NOX4 inhibition alter the effect of DOX on Ca handling ar ility?	nd 36
	5.4.4	Does XO and NOX inhibition alter the effect of DOX on myofilament sensitivity 137	y?
5	.5 Cor	cluding remarks13	37
5 Cha she	.5 Cor pter 6 - 1 ep ventri	cluding remarks13 The effects of doxorubicin and daunorubicin on Ca handling and contractility cular myocytes from sheep with heart failure	37 in 38
5 Cha she 6	.5 Cor pter 6 - 1 ep ventri .1 Intr	cluding remarks13 The effects of doxorubicin and daunorubicin on Ca handling and contractility cular myocytes from sheep with heart failure	37 in 38 39
5 Cha she 6	.5 Cor pter 6 - 1 ep ventri .1 Intr 6.1.1	cluding remarks	37 in 38 39
5 Cha she 6	.5 Cor pter 6 - 1 ep ventri .1 Intr 6.1.1 6.1.2	cluding remarks	37 in 38 39 39 41
5 Cha she 6	.5 Cor pter 6 - 1 ep ventri .1 Intr 6.1.1 6.1.2 6.1.3	cluding remarks	37 in 38 39 39 41 41
5 Cha she 6	.5 Cor pter 6 - 1 ep ventri .1 Intr 6.1.1 6.1.2 6.1.3 .2 Me	cluding remarks	37 in 38 39 39 41 41 42
5 Cha she 6	.5 Cor pter 6 - 1 ep ventri .1 Intr 6.1.1 6.1.2 6.1.3 .2 Me ² 6.2.1	Including remarks 13 The effects of doxorubicin and daunorubicin on Ca handling and contractility cular myocytes from sheep with heart failure 13 oduction 13 oduction 13 Altered Ca handling and contractility in heart failure 13 Heart failure and anthracyclines 14 Aims 14 Heart failure model 14	37 in 38 39 39 41 41 42 42
5 Cha she 6	.5 Cor pter 6 - 1 ep ventri .1 Intr 6.1.1 6.1.2 6.1.3 .2 Me ² 6.2.1 6.2.2	cluding remarks 13 The effects of doxorubicin and daunorubicin on Ca handling and contractility 13 cular myocytes from sheep with heart failure 13 oduction 13 Altered Ca handling and contractility in heart failure 13 Heart failure and anthracyclines 14 Aims 14 thods 14 Cell preparation 14	37 in 38 39 41 41 42 42 43
5 Cha she 6	.5 Cor pter 6 - 1 ep ventri .1 Intr 6.1.1 6.1.2 6.1.3 .2 Me ² 6.2.1 6.2.2 6.2.3	cluding remarks 13 The effects of doxorubicin and daunorubicin on Ca handling and contractility 13 cular myocytes from sheep with heart failure 13 oduction 13 Altered Ca handling and contractility in heart failure 13 Heart failure and anthracyclines 14 Aims 14 thods 14 Cell preparation 14 Excitation-contraction coupling measurements 14	 37 in 38 39 39 41 42 42 43 43
5 Cha she 6	.5 Cor pter 6 - 1 ep ventri .1 Intr 6.1.1 6.1.2 6.1.3 .2 Me ² 6.2.1 6.2.2 6.2.3 6.2.4 mechan	cluding remarks. 13 The effects of doxorubicin and daunorubicin on Ca handling and contractility 13 cular myocytes from sheep with heart failure 13 oduction 13 Altered Ca handling and contractility in heart failure 13 Heart failure and anthracyclines 14 Aims 14 thods 14 Excitation-contraction coupling measurements 14 Measurement of sarcoplasmic reticulum Ca content and Ca remove 14 14 14 14 14 14 14 15 14 16 14 17 14 18 14 19 14 14 14 14 14 15 14 16 14 17 14 18 14 14 14 14 14 15 14 16 14 17 14 18 14 19 14 </td <td>37 in 38 39 39 41 42 42 43 43 43 43</td>	37 in 38 39 39 41 42 42 43 43 43 43
5 Cha she 6	.5 Cor pter 6 - 1 ep ventri .1 Intr 6.1.1 6.1.2 6.1.3 .2 Me 6.2.1 6.2.2 6.2.3 6.2.4 mechan 6.2.5	cluding remarks. 13 The effects of doxorubicin and daunorubicin on Ca handling and contractility 13 cular myocytes from sheep with heart failure 13 oduction 13 oduction 13 Altered Ca handling and contractility in heart failure 14 Heart failure and anthracyclines 14 Aims 14 Heart failure model 14 Cell preparation 14 Measurement of sarcoplasmic reticulum Ca content and Ca removisms 14 Myofilament sensitivity measurements 14	37 in 38 39 41 42 42 43 43 43 43

	6.3.1 failure n	A detailed investigation of the effects of DOX on Ca and contractility in a heart nodel144
	6.3.2 failure n	The effect of doxorubicin on SR Ca content and rate of Ca removal in a heart nodel
	6.3.3 a heart f	A detailed investigation of the effects of daunorubicin on Ca and contractility in failure model
	6.3.4 heart fa	The effects of doxorubicin and daunorubicin on myofilament sensitivity in a ilure model
	6.3.5 healthy	A comparison of the effects of DOX on global Ca and contractility in cells from and HF sheep
	6.3.6 healthy	A comparison of the effects of DOX on SR Ca and Ca removal in cells from and HF sheep
	6.3.7 betweer	Alterations in response to daunorubicin in Ca handling and contractility na healthy and heart failure model159
6	.4 Disc	cussion
	6.4.1 cardiom	Does 1 nM doxorubicin alter intracellular Ca and contractility in sheep yocytes isolated from a heart failure model?161
	6.4.2 doxorub	Is SR Ca and the activity of the Ca removal mechanisms altered by 1 nM icin in a heart failure model?
	6.4.3 cardiom	Does 1 nM daunorubicin alter intracellular Ca and contractility in sheep yocytes isolated from a heart failure model?
	6.4.4 model?	Do doxorubicin and daunorubicin alter myofilament sensitivity in a heart failure 163
	6.4.5 healthy	Do the effects of doxorubicin on Ca handling and contractility differ between a and heart failure model?
	6.4.6 a contro	Do the effects of daunorubicin on Ca handling and contractility differ between I and heart failure model?165
6	.5 Cor	cluding remarks165
Cha	pter 7 - 0	General Discussion
7	.1 The	effects of doxorubicin and daunorubicin on excitation- contraction coupling167
	7.1.1	The effect of doxorubicin and daunorubicin on intracellular Ca and contractility 167
	7.1.2	The effect of doxorubicin and daunorubicin on SR Ca content168

7.1.3	The effect of doxorubicin and daunorubicin on the mechanisms of Ca removal 169
7.1.4	The effect of doxorubicin and daunorubicin on myofilament sensitivity to Ca 169
7.1.5	An integrated model
7.2 contrac	The role of oxidative stress in the anthracycline-mediated changes to excitation- tion coupling
7.2.1	Do doxorubicin and daunorubicin increase oxidative stress?171
7.2.2	What is the source of DOX-induced oxidative stress?
7.2.3	The role of xanthine oxidase in DOX-induced oxidative stress
7.2.4	The role of NADPH oxidase in DOX-induced oxidative stress172
7.2.5 conti	Can DOX-induced oxidative stress account for the alterations to excitation- raction coupling?
7.2.6 conti	Can DAUN-induced oxidative stress account for the alterations to excitation- raction coupling?
7.3	The effects of doxorubicin and daunorubicin in heart failure
7.3.1 in a r	The effect of doxorubicin and daunorubicin on excitation-contraction coupling nodel of heart failure
7.3.2 to th	Are the nature of the effects of doxorubicin and daunorubicin in health different ose in heart failure?
7.4	Clinical implications in anthracycline-induced cardiotoxicity178
7.4.1 dysfu	Do the cellular effects of doxorubicin and daunorubicin account for clinical unction?
7.4.2 high	How may the response to anthracycline treatment differ in patients considered risk of cardiac complications?178
7.5 I	mplications for alternative treatment strategies180
7.5.1	Compensatory treatment strategies for anthracycline-induced cardiotoxicity 180
7.5.2	Prevention treatment strategies for anthracycline-induced cardiotoxicity181
7.6	Limitations
7.7	Future work
7.8	Summary
Chapter 8	3 - Bibliography

Table of figures

 Figure 1.1: The Hallmarks of cancer and associated therapies Figure 1.2: Anatomy of the cardiac conduction system Figure 1.3: Phases of the cardiac action potential. Figure 1.4: Structure of the L-type calcium channel Cav1.2 Figure 1.5: Voltage dependence of L-type Ca current; the current – voltage relationship 	4 9 10 11 12
Figure 1.6: Contractile apparatus that comprises the sarcomeres in cardiac myocytes Figure 1.7: Na/Ca exchanger structure Figure 1.8: Relationship between calcium transient amplitude and force of sarcomere contraction in cardiac myocytes	17 19 22
Figure 1.9: Cubic relationship between sarcoplasmic reticulum calcium content and calcium transient amplitude	23
 Figure 1.10: β-adrenergic signalling cascade Figure 1.11: Schematic of excitation-contraction coupling Figure 1.12: Enzymatic processes that result in neutralisation of ROS Figure 1.13: The Fenton reaction resulting in ROS Figure 1.14: Chemical structure of daunorubicin and doxorubicin Figure 1.15: The effect of doxorubicin on left ventricular contractility and wall thickness Figure 2.1: Langendorff apparatus used for cardiac myocyte isolation Figure 2.3: Excitation spectra of Fura-2 Figure 2.4: Configuration of epi-fluorescent microscope illustrating filters and light 	25 27 29 30 34 39 48 49 54 56
Figure 3.1: Phase plane loops highlighting indications of change in myofilament sensitivity	66
Figure 3.2: Cell suitability of left ventricular cardiac myocytes isolated from control sheep when treated with 0.001 μ M- 10 μ M doxorubicin and washout Figure 3.3: The effects of DOX (0.001 – 10 μ M) on intracellular Ca and contractility in sheep ventricular myocytes Figure 3.5: The effect of 1 nM DOX on cytoplasmic Ca and sarcomere shortening Figure 3.5: The effect of 1 nM DOX on intracellular Ca and contractility in sheep ventricular myocytes Figure 3.5: The effect of 1 nM DOX on SR Ca content	68 69 71 72 74
Figure 1.9: Cubic relationship between sarcoplasmic reticulum calcium content and calcium transient amplitude Figure 1.10: β-adrenergic signalling cascade Figure 1.11: Schematic of excitation-contraction coupling Figure 1.12: Enzymatic processes that result in neutralisation of ROS Figure 1.13: The Fenton reaction resulting in ROS Figure 1.14: Chemical structure of daunorubicin and doxorubicin Figure 1.15: The effect of doxorubicin on left ventricular contractility and wall thickness Figure 2.1: Langendorff apparatus used for cardiac myocyte isolation Figure 2.3: Excitation spectra of Fura-2 Figure 2.4: Configuration of epi-fluorescent microscope illustrating filters and light pathways for use with Fura-2 Figure 3.1: Phase plane loops highlighting indications of change in myofilament sensitivity Figure 3.2: Cell suitability of left ventricular cardiac myocytes isolated from control sheep when treated with 0.001 μM- 10 μM doxorubicin and washout Figure 3.5: The effect of 1 nM DOX on cytoplasmic Ca and sarcomere shortening Figure 3.5: The effect of 1 nM DOX on SR Ca content	23 25 27 29 30 34 39 48 49 54 56 66 66 68 69 71 72 74

Figure 3.7: The effect of 1 nM DOX on Ca removal in sheep ventricular myocytes Figure 3.8: The effect of 1 nM DAUN on cytoplasmic Ca and sarcomere shortening Figure 3.9: The effect of 1 nM DAUN on intracellular Ca and contractility in sheep ventricular myocytes	75 77 78
Figure 3.10: The effect of 1 nM DAUN on SR Ca content Figure 3.11: The effects of 1 nM DAUN on Ca removal in sheep ventricular myocytes Figure 3.12: The effect of DOX and DAUN on myofilament sensitivity in sheep left ventricular myocytes	79 81 83
Figure 3.13: The effect of 1 nM DOX and DAUN on myofilament sensitivity Figure 4.1: ROS detected from sheep ventricular cardiac myocytes treated with various concentrations of hydrogen peroxide (10, 20, 50, 100 and 200 µM)	84 100
Figure 4.2: The effect of DOX and DAUN on OS production in sheep ventricular myocytes	101
Figure 4.3: The time dependent effect of 0.1 μ M DOX on ROS production in sheep ventricular myocytes with and without 1 mM NAC pre-treatment	103
Figure 4.4: Specimen traces of the effect of 1 mM NAC pre-treatment along with 1 nM DOX treatment on cytoplasmic Ca and contractility in sheep ventricular myocytes	104
Figure 4.5: The effect of 1 Mm NAC pre-treatment on Ca handling and contractility in sheep ventricular myocytes when treated with 1 nM DOX	105
Figure 4.6: The percentage change compared to control in cardiac myocytes treated with 1 nM DOX only and those treated with 1 mM NAC prior to DOX treatment	107
Figure 4.7: The effect of 1 nM DOX on Ca removal in sheep ventricular myocytes were pre-treated with 1 mM NAC	109
Figure 4.8: The percentage change compared to control in cardiac myocytes treated with 1 nM DOX and those pre-treated with 1 mM NAC prior to DOX treatment	110
Figure 5.1: Production of ROS via xanthine oxidase	118
Figure 5.2: The time dependent effect of 0.1 μ M DOX on OS production in sheep ventricular myocytes with and without XO and NOX inhibition	122
Figure 5.3: The effect of xanthine oxidase inhibition and 1 nM DOX on cytoplasmic Ca and contractility in sheep ventricular myocytes	124
Figure 5.4: The effect of xanthine oxidase inhibition and 1 nM DOX on intracellular Ca and contractility in sheep ventricular myocytes	125
Figure 5.5: Specimen traces of the effect of NOX2 and NOX4 inhibition and DOX on cytoplasmic Ca and contractility in sheep ventricular myocyte	127
Figure 5.6: The effect of NOX2 and NOX4 inhibition and 1 nM DOX on intracellular Ca and contractility in sheep ventricular myocytes	128
Figure 5.7: The percentage change compared to control in cardiac myocytes treated with 1 nM DOX only and those treated with FE or VAS3947 prior to DOX treatment	130
Figure 5.8: The effect of 1 nM DOX on myofilament sensitivity in sheep ventricular cardiac myocytes pre-treated with antioxidants	132
Figure 5.9: Mean normalised cytoplasmic Ca at 50 % sarcomere length relaxation (EC_{50})	133

Figure 6.1: The effect of 1 nM DOX on cytoplasmic Ca and sarcomere shortening in145cardiac myocytes from a heart failure model145

Figure 6.2: The effect of 1 nM DOX on intracellular Ca and contractility in sheep 146 ventricular myocytes from a heart failure model.

Figure 6.3: The effect of 1 nM DOX on SR Ca content in a heart failure model

Figure 6.4: The effect of 1 nM DOX on Ca removal in sheep ventricular myocytes from 149 a heart failure model

Figure 6.5: The effect of 1 nM DAUN on cytoplasmic Ca and sarcomere shortening in 151 cardiac myocytes from a heart failure model

Figure 6.6: The effect of 1 nM DAUN on intracellular Ca and contractility in sheep 152 ventricular myocytes from a heart failure model

Figure 6.7: The effect of DOX and DAUN on myofilament sensitivity in sheep left 154 ventricular myocytes from a heart failure model

Figure 6.8: The effect of 1 nM DOX on Ca handling and contractility in healthy versus 156 heart failure animal model.

Figure 6.9: The effect of 1 nM DOX on Ca removal in sheep ventricular myocytes 158 isolated from healthy versus heart failure animal model

Figure 6.10: The effect of 1 nM DAUN on Ca handling and contractility in healthy 160 versus heart failure animal model

Figure 7.1: The effects of DOX and DAUN on Ca handling and contractility in sheep 170 ventricular myocytes

Figure 7.2: The role of DOX-induced OS on Ca handling and contractility in sheep 175 ventricular myocytes

Table of Tables

Table 2.1: Calcium free/ Isolation solution	48
Table 2.2: Taurine solution	49
Table 2.3: Normal Tyrodes solution	50
Table 2.4: Equipment to measure intracellular Ca and contractility	52

148

Acknowledgements

I am grateful for all the guidance and continuous support provided by my supervisor, Dr David Greensmith.

Many thanks to the Trafford / Dibb laboratories of the University of Manchester, for the provision of sheep ventricular myocytes and use of the experimental models. I would also like to thank Kidscan Children's cancer research for supporting my PhD studentship.

A special thank you to my parents, Joanne and Iain Foster for their constant love and belief in my abilities. My academic journey so far would not have been possible without their unwavering support.

I'd also like to thank my sisters, Zoe Robinson and Rebecca Hopkins and my nephews, Isaac and Harry for always putting a smile on my face.

Lastly, I would like to thank my colleagues at the University of Salford that helped to make this experience enjoyable.

Confirmation of ethical approval



Research, Innovation and Academic Engagement Ethical Approval Panel

Doctoral & Research Support Research and Knowledge Exchange, Room 827, Maxwell Building University of Salford Manchester M5 4WT

T +44(0)161 295 5278

www.salford.ac.uk/

6 December 2019

Amy Foster

Dear Amy,

<u>**RE: ETHICS APPLICATION STR1920-05**</u> — The cellular basis and prevention of anthracycline-induced late cardiac toxicity following childhood cancer treatment

Based on the information you provided, I am pleased to inform you that your application STR1920-05 has been approved.

If there are any changes to the project and/ or its methodology, please inform the Panel as soon as possible by contacting <u>S&T-ResearchEthics@salford.ac.uk</u>

Yours sincerely,

presed.

List of Abbreviations

AC	Adenylyl cyclase
ALL	Acute lymphoblastic leukaemia
ATP	Adenosine triphosphate
AVN	Atrioventricular node
BDM	2,3 – Butanedione monoxime
BPM	Beats per minute
BSA	Bovine serum albumin
Са	Calcium
CaCl ₂	Calcium chloride
CaMKII	Calmodulin-dependent protein kinase II
CaM	Calmodulin
cAMP	Cyclic AMP
CDI	Calcium-dependent activation
CICR	Calcium-induced calcium release
CRUs	Calcium release units
CSQ2	Calsequestrin 2
DAUN	Daunorubicin
DCFDA	2,7-dichlordihydrofluorescein
DEX	Dexrazoxane
DMSO	Dimethyl sulfoxide
DOX	Doxorubicin
DPHs	Dihydropyridines
ECC	Excitation-contraction coupling
FC	Flow cytometry
GPx	Glutathione peroxidase
GSH	Glutathione
H_2O_2	Hydrogen peroxide

HF	Heart failure
HIF – 1	Hypoxia-inducible factor 1
HR	Heart rate
JPH2	Junctophilin - 2
К	Potassium
KCI	Potassium chloride
LTCC	L-type calcium channels
LVEF	Left ventricular ejection fraction
Mg	Magnesium
MgCl ₂	Magnesium chloride
MgSO ₄	Magnesium sulfate
Na	Sodium
NaCl	Sodium chloride
NADPH	Nicotinamide adenine dinucleotide phosphate
NaH ₂ PO ₄	Monosodium phosphate
NCX	Sodium-calcium exchanger
NETs	Neutrophil extracellular traps
NO	Nitric oxide
NOX	NADPH oxidase
0 ₂ -	Superoxide anion
ОН	Hydroxyl radicals
OS	Oxidative stress
PLB	Phospholamban
PMCA	Plasma membrane calcium ATPase
ROS	Reactive oxygen species
RyR	Ryanodine receptor
SACPs	Sinoatrial conduction pathways
SAN	Sinoatrial node

- SERCA Sarcoendoplasmic reticulum calcium-ATPase
- SOD Superoxide dismutase
- SR Sarcoplasmic reticulum
- SV Stroke volume
- Tm Tropomyosin
- Tn Troponin
- Tn1 Troponin 1
- TnC Troponin-C
- TnT Troponin-T
- VDI Voltage-dependent inactivation

Abstract

Anthracyclines such as doxorubicin (DOX) and daunorubicin (DAUN) are effective chemotherapeutics and contribute to improved cancer survival rates in children and adults. However, anthracyclines exhibit acute and chronic cardiotoxicity which can produce heart failure in cancer survivors. While the cellular basis remains unclear, limited previous studies show DOX perturbs certain aspects of excitation-contraction coupling and increases production of reactive oxygen species (ROS). Fewer studies have investigated the effects of DAUN and the effects of either anthracycline on excitation-contraction coupling (ECC) in a large animal model has yet to be demonstrated. Furthermore, the extent to which altered ECC is dependent on anthracycline-induced ROS production remains ambiguous. This is compounded by the fact that no studies have investigated whether elevated ROS production produces oxidative stress in cardiac myocytes. To address these gaps in our understanding, we performed the first integrative investigation of the effects of DOX and DAUN in sheep ventricular myocytes. We also measured the effect of DOX and DAUN on oxidative stress in these cells and further elucidated the underlying sources of ROS. Furthermore, we investigated the dependence of perturbed ECC on ROS thence oxidative stress elevations.

Sheep ventricular myocytes were enzymatically isolated in accordance with the Animals (Scientific Procedures) Act, UK, 1986 and used for all experiments. Intracellular calcium and contractility dynamics were measured using epi-fluorescent photometry and video sarcomere detection simultaneously. Cells were field stimulated at 0.5 Hz then acutely exposed to 1 nM DOX or DAUN. Rapid application of 10 mM caffeine was used to measure SR Ca content. Oxidative

stress was measured using CellROX red. Fluorescent images were captured using the cytation imaging system and cell fluorescence determined using ImageJ software.

DOX reduced the activity of SERCA and increased the activity of NCX resulting in a reduction in SR Ca content. DAUN also reduced SR Ca content however due to an interaction with

xviii

caffeine the mechanism could not be fully elucidated. The decrease in SR Ca content accounted for a decrease in systolic Ca which underpinned a decrease in systolic shortening. Both DOX and DAUN increased myofilament sensitivity to Ca, potentially offsetting the effect on contractility. DOX increased oxidative stress in a concentration and time-dependent manner. DAUN also increased oxidative stress, but only at relatively high concentrations (10 mM). Removal of oxidative stress by n-acetylcysteine (NAC) attenuated the effects of DOX on the majority of Ca handling and contractility parameters. For example, the effect of DOX on SR Ca content and Ca transient amplitude were reduced by approximately 50 %. Inhibition of the ROS producing enzymes NADPH oxidase (NOX) and xanthine oxidase (XO) reduced DOX-mediated oxidative stress by ~50 % and ~20 % respectively and attenuated the effects on ECC. In cells from sheep with heart failure, DOX reduced SR Ca content thence systolic Ca and contractility but had no effect on SERCA and NCX.

These findings suggest that in a large animal model, DOX and DAUN decrease SR Ca content leading to a reduction in systolic Ca thence contractility. In the case of DOX, decreased SERCA and increased NCX activity likely contribute to the decrease of SR Ca content. However, that this isn't the case in heart failure suggests a role for other mechanisms. These findings are also the first to show that DOX and DAUN increase intracellular oxidative stress in the heart and that NOX and XO are key enzymatic sources of ROS. Furthermore, these findings show this increase in oxidative stress is pathologically important as it accounts for approximately half of the effects of DOX on ECC. Collectively, these findings further elucidate the effects of anthracyclines on ECC and make important contributions to the understanding of the cellular basis of anthracycline-mediated cardiotoxicity. Furthermore, the dependence on and sources of oxidative stress reveal clinically relevant therapeutic targets.

Chapter 1 General Introduction

1.1 Cancer

Cancer is the uncontrollable growth and division of cells within the body. There are more than 100 types of cancer and subtypes may then be found within certain organs (Hanahan & Weinberg, 2011) with incident rates of the disease predicted to increase each year in the UK (Smittenaar *et al.*, 2016). Treatment for cancer has improved due to advances in research, however improvements are still required, not only to reduce lives lost to this disease but to improve the quality of life following treatment.

1.1.1 Childhood cancer

The probability of being diagnosed with a malignant tumour increases with age. This is likely due to prolonged exposure to carcinogens and other risk factors discussed in 1.1.2. However, certain cancer types are predominantly seen in children and include acute lymphoblastic leukaemia (Yamamoto & Goodman, 2008), cancers of the brain and spinal cord and lymphomas (Pesola *et al.*, 2017).

Between 1980 and 2006 almost 30,000 cases of childhood cancer were recorded in Great Britain, which equates to approximately 1150 cases per year, with leukaemia being the most common (Kendall *et al.*, 2013). Data from between 2016 and 2018 shows that incidence increased to around 1800 cases of childhood cancer per year (Cancer Research UK, 2022). 27 % of these cancer cases are leukaemia (Grabow *et al.*, 2018).

The type of cancer prevalent in children often differs to that suffered by adults. In adults, colon and rectum cancer are among the most prevalent whereas this is not the case for children (NHS digital, 2019). The aetiology of the cancer is probably an influencing factor in this difference.

1.1.2 Aetiology of cancer

1.1.2.1 Hallmarks of cancer

In 2000, Hanahan & Weinberg defined 6 hallmarks of cancer which are required in terms of cell physiology - for a malignant growth to develop, these are: self-sufficiency in growth signals, insensitivity to anti-growth signals, evading apoptosis, tissue invasion & metastasis, limitless replicative potential, and sustained angiogenesis. In essence, this means that cancer cells grow without an external stimulus and can avoid death via apoptosis. Furthermore, cancer cells can invade tissues and alter angiogenesis which aids metastasis, and insensitivity to antigrowth signals allow the cells to replicate indefinitely. The hallmarks have since been expanded to include characteristics that enable the other characteristics specified (see figure 1.1). These include genomic instability and tumour-promoting inflammation which aid the other hallmarks microenvironment (Hanahan & Weinberg, 2011). These hallmarks of cancer are universal regardless of age at diagnosis.



Figure 1.1: The Hallmarks of cancer and associated therapies (Hanahan & Weinberg, 2011).

1.1.2.2 Aetiology of childhood cancer

The aetiology of childhood cancer is often unclear, but it is commonly accepted that genetic predisposition is a major contributing factor. Different types of childhood cancer are underpinned by genetic irregularities. For example, *TEL-AML1* chromosome translocation is the most common fusion gene (hybrid of two independent genes) present in acute lymphoblastic leukaemia (ALL). However, presence of this genetic translocation does not mean leukaemia is inevitable, therefore other factors must influence the progression of disease (Zuna *et al.*, 2011). This is where the other hallmarks of cancer such as sustaining proliferative signalling and resisting cell death may play a role.

This theory is also described in the two-hit model (also known as the two-stage model) which predicts if a person has a genetic mutation this alone is not enough to result in cancer. However, if a second genetic or environmental mutation occurs the risk of developing cancer increases considerably. This model was originally developed by Knudson when studying a retinal cancer, retinoblastoma (Knudson, 1971).

Identifying environmental and genetic risk factors for cancer is vital to reduce the number of people who develop the disease. However, it is also important to improve screening and treatment strategies.

1.1.3 Cancer diagnosis

Symptoms of cancer vary depending on the type and age of onset. In children these symptoms may not be specific and may present as more common or relatively benign illnesses. This often means that correct diagnosis takes longer by which time the disease has progressed to a more advanced stage (Shanmugavadivel *et al.*, 2022). Campaigns exist to increase general practitioners' awareness of childhood cancer in the UK. For example, the Grace Kelly childhood cancer trust has a diagnostic support tool which highlights some symptoms such as abdomen distension, recurring viral illnesses, headaches and vomiting (Walker, 2021). If a child presents to a health care professional with these symptoms at least three times in three months, a more thorough investigation may be conducted to ascertain if cancer is the cause. Symptoms such as lumps/swellings, pallor and fatigue further increase the likelihood that cancer may be present (Dommett *et al.*, 2013).

Once cancer is suspected various tests may be conducted. For example, in suspected cases of ALL bone marrow cytology including identifying the distribution of lymphoblasts and morphological examination of lymphocytes may be conducted (Brown *et al.*, 2020; Inaba *et al.*, 2013). Patients may then undergo further tests if cancer is detected to establish if any genetic abnormalities are present as this may alter treatment regimens (Nogrady, 2020).

1.1.4 Treatment of childhood cancer

Various treatment strategies exist depending on the type of cancer and age of the patient. Fundamental approaches often involve the use of anti-cancer drugs, radiation therapy and surgery, with treatment for children being given at lower doses due to the increased risk of long-term effects (Kattner *et al.*, 2019).

Anti-cancer drugs with different modes of action are often used in combination to reduce the risk of resistance and limit possible side effects (Tew, 2012). Examples of classes of anti-cancer drugs used in combination include steroids, anti-mitotic agents and anthracyclines (Schmiegelow *et al.*, 2010).

It is important to balance the concentration of the drug used to gain therapeutic effect without resulting in severe side effects, especially in children given the lifetime following treatment. Treatments for childhood cancer have improved considerably with 5-year survival reaching 80 %, however these survivors may then face various off-target effects (Armenian *et al.*, 2015). Cardiac complications are the most common cause of severe health issues in these survivors (Trachtenberg *et al.*, 2011). Deaths among childhood cancer survivors are 11 times more than that of the general population with 13 % of these attributed to cardiac causes (Reulen *et al.*, 2010). In particular, the anthracycline class of drugs have been associated with causing injury to cardiomyocytes which can induce heart failure as well as other complications (Henriksen, 2018). These drugs are commonly used to treat acute lymphoblastic leukaemia, acute myeloid leukaemia, Hodgkin lymphoma, Ewing sarcoma and osteosarcoma in children (Armenian & Bhatia, 2018). Anthracyclines are highly effective, however the cardiac implications need to be addressed. Therefore, it is important to understand how anthracyclines impact cardiac function on a cellular level. To do this we must first understand the cellular basis of cardiac function under normal physiological conditions.

1.2 Cardiac excitation-contraction coupling

1.2.1 The human heart

The human heart weighs between 250 - 350g and beats approximately 100,000 times a day (Shaffer *et al.*, 2014). The heart consists of 4 chambers: right atrium and ventricle and the left atrium and ventricle as shown in figure 1.2 (Whitaker, 2018). When functioning normally, the heart powers the circulation of enough oxygenated blood to meet metabolic needs of the body. The basic essential characteristics required for this are contractility and rhythmicity (Wilcken, 2012).

The fundamental cardiac cycle (a functional heart beat) consists of systole and diastole; during systole the ventricles contract increasing blood pressure while in diastole the ventricles relax allowing the chambers to refill with blood (Shaffer *et al.*, 2014). Within the chambers, heart muscle (the myocardium) is responsible for the generation of force required for pump function. Myocardial tissue is comprised mainly of contractile cells known as cardiac myocytes (Gupta & Poss, 2012). The generation of force at cell level is dependent on a process known as excitation-contraction coupling (ECC) which is a sequence of events from the electrical excitation of a given cardiac myocyte to that cardiac myocyte's contraction.

1.2.2 The role of calcium

Calcium (Ca) is essential in cardiac contraction (Ringer, 1883). As discussed below, at the cellular level Ca is involved in all stages of ECC; it contributes to the cellular action potential, links excitation to contraction (and controls inotropy), then acts as the ligand that activates the contraction itself.

Other signalling molecules exist, but unlike molecules such as phosphate, Ca is able to bind to and disassociate from sub-cellular structures rapidly. This is essential given the time course of one heart beat (Eisner, 2018). Ca is also ubiquitously involved in other cellular processes, such as gene expression and neurotransmitter release (Pantazis *et al.*, 2014). Ca is also a regulatory co-factor for many intracellular enzymes, including those involved in the production of reactive oxygen species (ROS). Examples include xanthine oxidase and NADPH oxidases (Görlach *et al.*, 2015) which are discussed in 1.4.1.1.

1.2.3 Excitation

1.2.3.1 The source of excitation

Contraction of cardiac muscle is initiated by a wave of excitation originating from the sinoatrial node (SAN), the cells of which are capable of spontaneous depolarisation (Shih, 1994). In mammals the SAN is located within the wall of the right atrium (Baruscotti & Robinson, 2007) and is comprised of pacemaker cells which are surrounded by collagen and elastin fibres (Mitrofanova *et al.*, 2018). The SAN structure is extremely complex and has at least four sinoatrial conduction pathways (SACPs) which conduct electrical impulses to the myocardium of the atria (Fedorov *et al.*, 2012). In non-disease states excitation begins within one of the intranodal pacemakers in one of the SACP which then propagates through the other SACPs (Fedorov *et al.*, 2012). The internodal pathways connect thus excite the atrioventricular node (AVN). The AVN delays excitation permitting discrete atrial excitation before transducing excitation to the His-Purkinje system allowing the wave of excitation to spread across the ventricles (Hucker *et al.*, 2007). This wave of excitation results in contraction of the ventricles.



Figure 1.2: Anatomy of the cardiac conduction system. SA (sinoatrial), AV (atrioventricular). Adapted from Munshi (2012).

1.2.3.2 The cardiac cellular action potential

The wave of excitation described in 1.2.3.1 is produced by the sequential excitation thus depolarisation of each cell in the myocardium. This depolarisation manifests as the cardiac cellular action potential. The ventricular action potential has 4 distinct phases as shown in figure 1.3. A wave of excitation activates voltage gated sodium (Na) channels resulting in Na influx that depolarises the sarcolemma producing phase 0 of the action potential, the Na channels inactivate within a few milliseconds and result in the phase 1 notch (Catterall, 2018). As with neuronal tissue, potassium (K) channels open allowing K efflux thus repolarisation resulting in phase 3 (Catterall, 2018). However, cardiomyocytes (CMs) have longer duration of action potential depolarisation due to transient Ca influx that offsets K efflux resulting in a plateau phase (phase 2). As K efflux dominates repolarisation occurs producing phase 4 (Cooper *et al.*, 2010).



Figure 1.3: Phases of the cardiac action potential. Phase 0 is depolarisation followed by phase 1 which is termed the notch. Phase 2 is known as the plateau phase. Repolarisation is shown in phase 3. Phase 4 is resting phase. Figure modified from Nerbonne & Kass (2005).

1.2.4 Sarcolemma calcium entry

1.2.4.1 L-type voltage gated calcium channel activation

The primary pathway by which Ca enters the cardiac myocyte from the extracellular fluid via the L-type calcium channels (Lederer *et al.*, 1989). There are four subtypes of L-type calcium channels (LTCC) with Ca_v1.2 being mainly expressed in cardiac muscle (Lipscombe *et al.*, 2004). Ca_v1.2 are selective to Ca and are mainly located at the transverse tubules in close proximity to ryanodine receptors (RyR) forming a dyad as shown in figure 1.11 (Bers, 2001b).

LTCC consist of 3 different subunits that are tissue specific. Ca_v1.2 channels consist of α_1 , $\alpha_2\delta_1$ and β sub-units as shown in figure 1.4 (Hu *et al.*, 2017; Wang *et al.*, 2004). The most common isoforms of α_1 and β within cardiac tissue are α_1 c and β_2 respectively (Bers, 2001b). The α_1 subunit consists of twenty-four segments, four repeats of six transmembrane segments (S1-S6), four of which are responsible for the voltage-sensing domain while S5-S6 are associated with the Ca conductive pore (Pantazis *et al.*, 2014). α_1 has binding sites for calcium channel blockers such as most dihydropyridines (DPHs) while, $\alpha 2\delta_1$ and β regulate the electrophysiological and pharmacological features of the LTCC (Hullin *et al.*, 2003).



Figure 1.4 – Structure of the L-type calcium channel Ca_v1.2 from Hu *et al* (2017).

L-type calcium channels are voltage dependent, so are activated by the depolarisation of the membrane during the action potential (see 1.2.3.2.) (Sato *et al.*, 2018). In the α_1 subunit S4-S6 are involved in the voltage dependent activation of the channel, and the S3-S4 linker has been shown to have a role in the kinetics of the activation (Nakai *et al.*, 1994). LTCC activate at around -40 mV with peak activation being at approximately +10 mV, taking in the region of 5 msec (Pelzmann *et al.*, 1998). Once activated, Ca channels open allowing an influx of Ca. This Ca influx is due to the extracellular Ca concentration being greater than intracellular.



Figure 1.5: Voltage dependence of L-type Ca current; the current – voltage relationship. Adapted from Bers (2001).

1.2.4.2 L-type voltage gated calcium channel inactivation

Inactivation of LTCC is important in normal cardiac function. LTCC can be inactivated by calcium-dependent and voltage-dependent mechanisms. 80 % of LTCC inactivation is calcium dependent (Lacinová & Hofmann, 2005). It is well understood that calcium-dependent inactivation (CDI) is initiated when Ca binds to the α_1 c subunit of the LTCC (Peterson *et al.*, 1999). This inactivation process involves amino acid sequences in the cytoplasmic segments of the α_1 c subunit (Zühlke & Reuter, 1998). An EF-hand motif and an IQ calmodulin binding motif are present in the α_1 c subunit that have been identified as being important in calcium-dependent inactivation (Zühlke & Reuter, 1998).

Calmodulin (CaM) is a protein involved in altering protein-protein interaction when there are changes in [Ca²⁺] (Chazin & Johnson, 2020). CaM is able to both inactivate and facilitate LTCC (Zühlke *et al.*, 1999). The influx of Ca, and CaM interacting with the IQ motif in LTCC leads to inactivation (Pitt *et al.*, 2001).

Voltage-dependent inactivation (VDI) is a relatively slower process compared to CDI (Findlay, 2002) and is facilitated by the current decay. During phase 3 of the ventricular action potential the efflux of K drives repolarisation leading to a decrease in the membrane potential which results in more LTCC being inactivated.

The influx of Ca due to activation of LTCC does not raise intracellular Ca to levels required for contraction. Rather it results in a mass release of Ca from the sarcoplasmic reticulum in a process is known as calcium-induced calcium release (CICR). To understand this, we must first consider the sarcoplasmic reticulum and associated structures.

1.2.5 The sarcoplasmic reticulum

The sarcoplasmic reticulum (SR) is a specialised form of endoplasmic reticulum and in the heart cells acts as an intracellular Ca store (Rossi *et al.*, 2008). In mammals, most of the Ca that activates contraction (see 1.2.7) is released from the SR in a process known as calcium induced calcium release.

1.2.6 Calcium-induced calcium release

Influx of Ca via the LTCC does not elevate cytoplasmic Ca to levels required for the activation of contraction. Rather, in a process known as calcium induced calcium release (CICR) it triggers mass Ca release from the SR which then allows contraction to occur (Hanna *et al.*, 2014). To fully understand this process, first, the structure and arrangement of proteins involved need to be considered.

1.2.6.1 The ryanodine receptor

On the surface of the SR are specialist Ca release channels. As these channels have a high affinity to ryanodine, they are termed ryanodine receptors (RyR) (Bers, 2001b).

With a molecular mass of more than 2MDa RyRs are the largest known ion channels and have a homotetrameric structure (Peng *et al.*, 2016). Three RyR isoforms are found in mammals,

though the isoform primarily expressed in the heart is RyR2 (Peng *et al.*, 2016). RyRs are found in clusters on the surface of the SR of cardiac myocytes near the plasma membrane (Sato & Bers, 2011). These clusters are known as Ca release units (CRUs), and release discrete units of Ca known as Ca sparks (Núñez-Acosta & Sobie, 2014). Approximately 14 RyR are in each cluster and each CRU is made up of 3 clusters at a minimum (Baddeley *et al.*, 2009; Zahradníková & Zahradník, 2012). However, this number can vary from a few to several hundred (Sato *et al.*, 2016).

RyR spontaneously and stochastically open in diastole however the open probability is low resulting in a small, short lived release of cytoplasmic Ca termed the Ca spark (Dulhunty *et al.*, 2012). However, certain factors regulate the open probability of RyR such as pH, magnesium (Mg), adenosine triphosphate (ATP) and Ca (hence CICR). A decrease in pH to 6.6 reduces the open probability of RyR by more than 50 % (Rousseau & Pinkos, 1990). Mg has also been shown to reduce the open probability of RyR and increase the decline in open probability when there is a rapid increase in [Ca] such as in CICR (Valdivia *et al.*, 1995). Due to the fact ATP and Mg do not change over the time course of a single cycle of excitation-contraction coupling (i.e. heart beat) it is thought that these molecules regulate the sensitivity of RyR to Ca; the ligand that gates RyR during ECC (Bers, 2001b).

This means in the context of ECC, *the* most important influence of RyR activity is the concentration of Ca in the cytosol (specifically, the dyadic cleft, see below) and SR [Ca] (Bers, 2001b).

1.2.6.2 The dyad

The intracellular junction between the sarcoplasmic reticulum cisterna and the sarcolemma is termed the dyadic junction (Franzini-Armstrong *et al.*, 1999). The dyadic junction is between 12-15 nM wide (Pinali *et al.*, 2013). The dyad is formed by a cluster of RyR in close proximity

to a given LTCC across the dyad which increases the open probability of the channels when Ca enters via LTCC.

Junctophilin-2 (JPH2) is a protein present in the dyad that contributes to the stabilisation of the junction (Takeshima *et al.*, 2000). Loss of this protein in the dyad can lead to suppression of CICR and therefore impacts ECC, this is likely due to the changes in dyadic junctional distances and loss of t-tubules (Oort *et al.*, 2011). The dysregulation of JPH2 has been implicated in various cardiac diseases which highlights the importance of the dyad in ECC (Beavers *et al.*, 2014).

1.2.6.3 The process of calcium induced calcium release

The local dyadic Ca regulates CICR (Greenstein *et al.*, 2006). While influx of Ca via LTCC is relatively small the local increase of Ca in this microdomain is high and given that cytosolic Ca is the primary determinant of RyR gating (see section 1.2.6.1) this leads to increased open probability of RyR. The stochastic opening of RyR is increased therefore spark frequency increases considerably and these sparks summate to produce mass Ca release (Winslow *et al.*, 2006), this is termed CICR.

1.2.7 Contraction

Mechanically, for cardiac myocytes to contract Ca must trigger the alteration of the conformation of myofilaments within the sarcomeres (Schwan & Campbell, 2015).

1.2.7.1 Contractile apparatus

Myofilaments are the contractile proteins involved in cardiac cellular contraction. The structure of the sarcomere and myofilaments are depicted in figure 1.6. The contractile apparatus is comprised of thick (myosin) and thin (actin) filaments (Solaro & Rarick, 1998). These myofilaments are arranged in an overlapping structure that form contractile units termed sarcomeres. Using X-ray diffraction the organisation of the filaments in the sarcomere were identified with alternating A and I bands forming a striated appearance (Hanson &

Huxley, 1953). The Z-line depicts the end of each sarcomere and anchors the structures together. Actin protrudes from the Z-line towards the centre of the sarcomere (see figure 1.6). Actin filaments are double stranded helical structures that overlap with myosin in regions except the I band (Huxley & Niedergerke, 1954).

The interaction between myosin and actin is the fundamental basis of sarcomere shortening and is regulated by troponin and tropomyosin (Vanburen & Palmer, 2010) as illustrated in figure 1.6. Troponin (Tn) is a complex of three proteins: troponin-C (TnC), troponin-T (TnT) and cardiac troponin-1 (Tn1) (de Tombe, 2003). Tropomyosin (Tm) is a protein that is located between actin subunits, and every seventh troponin is bound to Tm which prevents myosin from binding to actin and therefore prevents contraction (Bers, 2001b). Tm dissociates from Tn when Ca binds to TnC subunit of Tn, this binding strengthens the interaction of TnC to Tn1 which is the subunit bound to actin (Bers, 2001b). Therefore, the interaction between Tn1 and actin is less stable and allows myosin to interact with actin.

The attachment of myosin to actin is termed the cross bridge. In the presence of ATP and Ca cross bridge cycling occurs and the actin slides alongside myosin (towards the z-line) shortening the sarcomere, thus myofibril, thus myocytes. Upon removal of Ca these contractions cease.


Figure 1.6: Contractile apparatus that comprises the sarcomeres in cardiac myocytes adapted from Schwan & Campbell (2015).

1.2.8 Relaxation

For the chambers of the heart to be filled before the next contraction, relaxation must occur, if this is impaired then the cardiac output would be reduced, and the oxygenated blood circulated would be insufficient (Biesiadecki *et al.*, 2014). The combined activity of the Ca removal mechanisms allows restoration of diastolic Ca. Here, Ca disassociates from troponin and thus the myocytes therefore the heart relaxes.

1.2.8.1 The need for calcium removal

As Ca is integral in cardiac contraction, for relaxation to occur the free Ca must be removed from cytosol. Four transport systems compete for cytoplasmic Ca during relaxation; the sarcoendoplasmic reticulum calcium transport ATPase (SERCA), sarcolemmal sodium-calcium exchanger (NCX), sarcolemmal Ca-ATPase (PMCA) and mitochondrial Ca uniport system (Bassani *et al.*, 1992). By blocking each channel separately Bassani *et al* (1992) were able to determine the relative contribution of each transport system to Ca removal. In mammals SERCA and NCX account for 95 % of calcium removal (Bers, 1997).

1.2.8.2 Sarco-endoplasmic reticulum calcium transport ATPase

SERCA belongs to the P-type ion transporting ATPase family (Bers, 2001b). Three genes encode SERCA. *SERCA1* is expressed in fast twitch muscle while *SERCA2* is expressed in slowtwitch skeletal and cardiac muscle, the third gene *SERCA3* is present in various tissues that are not muscles (Lytton *et al.*, 1992). The transmembrane region of SERCA contains ten α helices (M1-M10), with M4-M6 being thought to be responsible for the Ca binding sites (Toyoshima *et al.*, 2000). In the heart, SERCA is responsible for 70 – 90 % of systolic Ca removal (Bers, 2001b). It also replenishes SR Ca in readiness for the next beat.

Transport of Ca back into the SR via SERCA is always against a concentration gradient thus an active process. Two molecules of Ca bind to the cytoplasmic side of SERCA along with a molecule of ATP (Bers, 2001b). The transmembrane site M4-M6 have higher Ca affinity than those in the luminal side (M1-M2) due to the amino acid sequence (Clarke *et al.*, 1989). Phosphorylation leads to a change in conformation that prevents the bound Ca re-entering the cytoplasm and reduces the binding affinity, meaning the calcium molecules are released into the lumen of the SR (Bers, 2001b).

SERCA activity is regulated by phospholamban. Phospholamban (PLB) is a 6KDa pentamer protein that is found within the SR of cardiac, smooth and slow twitch skeletal muscles (Schmidt *et al.*, 2001). In the dephosphorylated basal state, PLB inhibits SERCA activity by decreasing the ability of SERCA to bind Ca (Akin *et al.*, 2013). When phosphorylated the effects of PLB are relieved and an increase in SERCA activity is observed (Kranias & Hajjar, 2012).

1.2.8.3 Sodium- calcium exchanger

The Sodium-calcium exchanger (NCX) is a membrane bound exchanger that removes 10-30% of systolic Ca (Bassani *et al.*, 1992) and is responsible for the majority of Ca removal out of the cell. The exchange mechanism has been found to occur at a ratio of approximately 3:1, in that for every 3 Na brought in to the cell, 1 Ca is removed (Pitts, 1979). Structure of the NCX can be seen in figure 1.7.



Figure 1.7: Na/Ca exchanger structure. The exchanger contains 9 transmembrane domains along with a large cytoplasmic loop from Philipson & Nicoll (2000).

The concentration of the substrate at which the speed of reaction is half its maximum is termed K_m. The K_m for intracellular Na is around 20 mM for the NCX, while for intracellular calcium it is 0.6 μ M (Miura & Kimura, 1989). Miura & Kimura (1989) identified that external Ca is not required for NCX current however intracellular Ca is essential in activating this channel with the exchanger being fully activated within an intracellular Ca concentration range that is present physiologically at resting.

NCX is not only sensitive to the intracellular and extracellular concentrations of Na and Ca but also to the membrane potential (Bers, 2001b). Depolarisation of the membrane leads to Ca entry as described in 1.2.4.1, which enhances the activity of NCX.

NCX activity is also influenced by pH as a low pH reduces the exchange. Maximal exchange occurs at an extracellular pH of 7.6, with activity reducing to 70 % at pH 6 and a further reduction resulting in 10 % activity at pH 4 (Marcel & Ernst, 2000).

1.2.8.4 Other calcium removal processes

As mentioned in 1.2.8.1 the majority of Ca removal in mammals in attributed to SERCA and NCX with PMCA and mitochondrial Ca uptake accounting for less than 5 % of removal (Bers, 1997).

The PMCA is a P-type ATPase like SERCA (Bers, 2001b). This means that PMCA uses energy produced when ATP is converted to ADP and phosphate to transport ions. Structurally, PMCA has a long cytosolic C-terminal tail which binds calmodulin (Brini *et al.*, 2013). There are 4 isoforms of PMCA with isoforms 1 and 4 being found in multiple tissue types and have been found to be an important regulator within the heart (Mohamed *et al.*, 2013). At resting [Ca] the calmodulin binding motif in the C terminal interacts with the catalytic site which inhibits the activity of PMCA (Bruce, 2018). When the concentration of intracellular Ca increases it binds to the autoinhibitory motif in the C terminal along with calmodulin and causes a conformational change which increases the pump's ability to transport Ca out of the cell (Carafoli, 1994).

The mitochondria takes up Ca via a uniport system, meaning it travels down a concentration gradient caused by removal of a positively charged particle from the mitochondria (proton) (Bers, 2001b). This uptake is predominantly controlled by a protein complex known as the mitochondrial Ca uniporter (Nickel *et al.*, 2020).

1.2.8.5 The calcium transient

Cellular contraction causes an increase in cytoplasmic Ca concentration and produces a systolic Ca transient (Eisner, 2014). The Ca transient is a ratio of fluorescence emissions at 340 nm which increase with a rise in intracellular Ca concentration to fluorescence emissions at 380 nm which decrease with a rise in intracellular Ca concentration. The rising phase of the Ca transient indicates the increase in cytoplasmic Ca concentration while the decay phase of the Ca transient illustrates the decrease in intracellular Ca concentration (removal of Ca).

1.2.9 Control of inotropy

Previous sections described the fundamental process of ECC, however the heart is a dynamic organ and cardiac output is constantly modulated to meet changing metabolic demand. Cardiac output is the product of stroke volume (SV) x heart rate (HR) (Vincent, 2008). Therefore, to increase cardiac output the heart SV and HR must also increase (and *vice versa* for decrease). For example, during exercise the cardiac output is increased to meet the increased rate of oxygen consumption.

Changes in SV have a cellular basis as the greater the cell contracts, the greater the chamber contracts and therefore the greater SV will be. To understand how SV is regulated by the cells, we must understand the dependence of force on systolic Ca.

1.2.9.1 The dependence of contractile force on systolic Ca

The force of contraction of the sarcomere is determined by systolic Ca. A concentration of approximately 70 μ mol of intracellular calcium is required to achieve half the maximum activation of contraction (Bers, 2001a). As the concentration of calcium increases (and therefore the Ca transient amplitude) the force of contraction increases up to a given point as shown in figure 1.8.



Figure 1.8: Relationship between calcium transient amplitude and force of sarcomere contraction in cardiac myocytes. Adapted from Sun & Irving (2009).

We must now consider what controls the levels of systolic Ca which results in a change of force to meet metabolic demand.

1.2.9.2 The dependence of systolic Ca on sarcoplasmic reticulum Ca content

The Ca content of the SR is directly related to systolic Ca (Trafford *et al.*, 1997). The amplitude of the Ca transient is dependent on the cube of SR Ca content as shown in figure 1.9 (Trafford *et al.*, 2000).

The steepness of this relationship is due to several factors. As per 1.2.6.1 RyR open probability is dependent on luminal/SR Ca content therefore, an increase in SR Ca content results in an increase in the basal open probability of RyR meaning for a given Ca trigger via LTCC the open probability of RyR increases further (Eisner, 2014). This means more Ca sparks occur and greater Ca release. The steepness of this relationship means that modest changes in SR Ca content will have a greater impact on systolic Ca. For example, a minor decrease in SR Ca content will result in a larger decrease in systolic Ca. To maintain Ca amplitude with each contraction the SR Ca content of the cell has to be precisely regulated. This makes SR Ca content a powerful physiological regulator of inotropy.



Figure 1.9: Cubic relationship between sarcoplasmic reticulum calcium content and calcium transient amplitude. Adapted from Trafford *et al* (2000).

1.2.9.3 The dependence of systolic Ca on trigger Ca

A rise may occur due to an increase in SR Ca release as discussed in 1.2.9.2, however may also occur via Ca influx (via LTCC). The two are not mutually exclusive as an increase in Ca influx triggers a greater release in SR Ca release. As discussed in 1.2.6.1 the open probability of RyR is also dependent on cytosolic [Ca] in the dyadic junction. Therefore, for all things being equal the greater the influx of Ca, the greater the open probability of RyR meaning spark frequency increases resulting in greater summation and a greater fractional SR Ca release.

1.2.9.4 Regulation of SR Ca content

SERCA is a powerful regulator of SR Ca content as evidenced in heart failure as decreased activity results in a decrease in SR Ca content (Eisner, 2014). NCX is also important in regulating SR Ca content and recent evidence suggest that is a far more important regulator than originally thought with this contributing to SR Ca content to a greater extent than SERCA (Bode *et al.*, 2011). We must now consider the signalling mechanisms that influence the aforementioned regulator of inotropy.

1.2.9.5 β -adrenergic stimulation

 β -adrenergic stimulation is one of the main physiological mechanisms to modulate cardiac function (Bers, 2001b). Three β -adrenergic receptors have been identified with β_1 and β_2 receptors involved in positive inotropic reactions (Lee *et al.*, 2010). The dominant β receptor in the heart is β_1 .

As shown in figure 1.10 stimulation of β -adrenergic receptors then stimulates adenylyl cyclase (AC) via Gs-coupled proteins which leads to an increase in production of cyclic AMP (cAMP) (Bers, 2001b). This then activates cAMP protein kinase A which phosphorylates various targets such as; troponin I, sarcolemmal Ca channels, SR Ca release channels and phospholamban (Bers, 2001b).



Figure 1.10: β-adrenergic signalling cascade adapted from Johnson & Antoons (2018).

Phosphorylation of Tn1 reduces the affinity of Ca to bind to the Ca specific site on TnC and increases the rate of removal of Ca from the troponin complex, therefore accelerating the time taken for relaxation (Robertson *et al.*, 1982). Furthermore, β -adrenergic stimulation leads to phosphorylation of protein kinase A (Bryant *et al.*, 2014). This then alters the voltage required for activation and inactivation of LTCC to a more negative membrane potential (Chase *et al.*, 2010). Therefore, increasing the Ca current which results in more Ca entering the cell at each excitation.

As previously discussed, phosphorylation of phospholamban allows SERCA to be stimulated (see section 1.2.8.2). β -adrenergic stimulation leads to phosphorylation of phospholamban which increases SERCA activity and thus SR Ca content. To summarise β -adrenergic stimulation provides an inotropic set point. However, this set point must be tightly regulated to prevent arrhythmogenic consequences.

1.2.10 Autoregulation

Autoregulation ensures stability in Ca handling through a feedback mechanism. As discussed in section 1.2.9.2 regulation of SR Ca content is required to ensure Ca transient amplitude and therefore inotropy are kept consistent beat to beat. Autoregulation ensures this stability. A change in activity of a single component such as RyR will not produce a prolonged effect on SR Ca release. Though not physiologically applicable a depletion in SR Ca content resulted in Ca transient progressively returning to steady state where Ca influx balanced efflux (Trafford *et al.*, 1997). This demonstrates how through the feedback autoregulation regulates demand. To further explain, an increase in the open probability of RyR will result in a brief increase in SR Ca release therefore increasing systolic Ca. On the next beat LTCC inactivation is enhanced due to the dependence on Ca. The increase in systolic Ca will increase the forward-mode of NCX activity resulting in an increase in Ca efflux. Ca efflux exceeds influx which leads to reduced intracellular Ca for subsequent beats, therefore less Ca is taken up into the SR via SERCA and SR Ca content is reduced. A reduction in SR Ca content and reduced Ca trigger compensate for the increase in open probability of RyR and restores the Ca transient amplitude (Eisner *et al.*, 2000; Greensmith *et al.*, 2014).

In disease states autoregulation may be impacted which can lead to variation between beats because of the imbalance in Ca influx/efflux. A change in a single component of Ca handling will not lead to a long-term effect on inotropy and the effect may be acute due to the other mechanisms in place. However, if more than one of these components are impacted issues with autoregulation will occur resulting in heart problems. The metabolic demand of the cell may vary in different conditions which will therefore alter the threshold for autoregulation. Cellular inotropy is important in setting these thresholds.

1.3 Summary of Excitation-contraction coupling

The action potential results in depolarisation of the sarcolemma which activates LTCC and a small amount of Ca influx occurs. This then increases the open probability of RyR leading to a large magnitude of Ca release from the SR in a process termed CICR. The release of Ca from the SR binds to the myofilaments and facilitates contraction. For relaxation to occur this Ca must be removed. In mammals this is mainly achieved by NCX and SERCA. SERCA actively uptakes the Ca back into the SR, while NCX is responsible for sarcolemmal efflux of Ca. This is summarised in figure 1.11.



1.11: Schematic of excitation-contraction coupling from Bers (2002).

1.4 Reactive oxygen species and oxidative stress

1.4.1 Reactive oxygen species

Reactive oxygen species (ROS) include oxygen free radicals such as superoxide anion (O_2^-) and hydroxyl radicals (OH[•]) as well as oxidising agents such as hydrogen peroxide (H_2O_2) , they are formed due to partial reduction of oxygen (Ray *et al.*, 2012). While ROS are implicated in a variety of disease states there is also evidence that they are involved in normal physiological functions, such as intracellular signalling pathways (Finkel, 2011) and the antimicrobial action of neutrophils (Winterbourn *et al.*, 2016).

1.4.1.1 Physiological sources of reactive oxygen species

Enzymatic and non-enzymatic processes are involved in ROS generation including the mitochondrial electron transport chain, nicotinamide adenine dinucleotide phosphate (NADPH) oxidases, xanthine oxidase (XO) and myeloperoxidase (Sugamura & Keaney, 2011b). The mitochondria produce ATP and are involved in maintaining homeostasis for example, cardiac myocytes rely on the mitochondria to produce 95 % of the ATP (Ventura-Clapier *et al.*, 2011). However, they also have a role in generating ROS as a by-product of the respiration chain (Jensen, 1966). Approximately 2 - 3 % electron leakage occurs to form O^{•-} (Chance *et al.*, 1979). ROS may then leak out of the mitochondria and cause ROS-induced ROS release from neighbouring mitochondria (Zorov *et al.*, 2006).

Enzymatically, ROS may be produced via NADPH oxidase (NOX) that use NADPH as an electron donor (García *et al.*, 2023). There are seven isoforms of NOX and these enzymes are transmembrane oxidoreductases with 6 α helices (Chen *et al.*, 2012a; Sugamura & Keaney, 2011a). NOX2 is the most-studied and is primarily expressed in phagocytic cells but is also present in cardiac myocytes (Noreng *et al.*, 2022). NOX2 activation involves phosphorylation of an adaptor protein which leads to ROS production (Brandes *et al.*, 2014). This ROS production aids in killing microorganisms as part of the immune defence (Bayir, 2005).

Furthermore, xanthine oxidoreductase produces ROS as a biproduct of uric acid production as part of hypoxanthine metabolism (Kang & Ha, 2014). This enzyme exists in various mammalian tissues, with particularly high expression in capillary endothelium (Jarasch *et al.*, 1981). The enzyme exists in two interconvertible forms in mammals; xanthine dehydrogenase and xanthine oxidase (Chen & Meng, 2022). Both forms of the enzyme convert hypoxanthine to xanthine and xanthine to uric acid, however XO has an important role when cells are under stress as it generates O^{•-} and H₂O₂ (Harrison, 2002).

1.4.1.2 The physiological fate of ROS

Various antioxidants exist naturally within the human body to ensure a balance in ROS and avoid oxidative stress. These may be enzymatic such as catalase and glutathione peroxidase (GPx) or external molecules such as certain vitamins, flavonoids and carotenoids which are ingested as part of the diet (He *et al.*, 2017). Antioxidant enzymes scavenge ROS, without this ROS production may lead to oxidative stress. Superoxide species produced via enzymes and the mitochondrial electron transport chain are converted to H_2O_2 by superoxide dismutase (SOD), H_2O_2 is then neutralised to $H_2O + O_2$ by catalase and GPx as shown in figure 1.12 (Bayir, 2005).



Figure 1.12: Enzymatic processes that result in neutralisation of ROS adapted from Bayir (2005).

Oxidative stress occurs when there is an imbalance in the generation of reactive oxygen species (ROS) and the antioxidant defences (Burton & Jauniaux, 2011). ROS can be produced in many tissues including the heart by various mechanisms such as the uncoupling of NO synthase, the action of cytokines, or by NADPH oxidases (Giordano, 2005). To prevent the

overproduction of ROS certain systems are in place that may be either enzymatic or nonenzymatic. Antioxidant enzyme systems to combat ROS include superoxide dismutase (SOD), catalase and glutathione (GSH), with GSH being the most abundant intracellular thiolbased antioxidant (Nordberg & Arnér, 2001). Issues occur if there is an over production of ROS or under activity of the systems to remove ROS.

If superoxide and therefore H_2O_2 production overwhelms SOD and GPx/catalase this may result in further ROS production by reacting with transition metals (see figure 1.13). The reaction of H_2O_2 with iron (II) results in OH[•] and further reactions may then occur with iron (III) to generate $O_2^{\bullet-}$ and reduce iron (III) to iron (II) to allow self-propagation (Bayir, 2005). When ROS production exceeds the antioxidant capacity this results in oxidative stress (OS).



Figure 1.13: The Fenton reaction resulting in ROS production from Bayer (2005).

1.4.2 Oxidative stress and disease

As previously stated, OS is implicated in a range of diseases. One area of interest is the implication of OS in heart failure and cardiovascular events. OS is correlated to increased endothelial dysfunction which can be linked to coronary artery disease (Heitzer *et al.*, 2001). Endothelial dysfunction results in the endothelium losing its anti-inflammatory properties and its ability to maintain vascular homeostasis (Ana María *et al.*, 2015) which may then give rise to states such as coronary artery disease.

Furthermore, patients with congenital heart failure have a higher level of oxygen free radicals (Keith *et al.*, 1998). One theory to explain this is that the chronic activation of renin-

angiotensin-aldosterone systems is associated with heart failure (Münzel *et al.*, 2017). Research has shown the vasoconstrictor angiotensin II, activates monoamine oxidase which is an enzyme that produces H₂O₂ while deaminating compounds such as serotonin (Manni *et al.*, 2013). As previously mentioned, hydrogen peroxide is a ROS. Other sources of ROS in heart failure include the enzymes xanthine oxidase and uncoupled nitric oxide synthases (Borchi *et al.*, 2010). Angiotensin II also activates NADH/NADPH oxidase which stimulates the production of superoxides (Ushio-Fukai *et al.*, 1996).

It is evident that cardiovascular disease is associated with OS, this cardiac dysfunction is often underpinned by perturbed ECC.

1.4.3 The effects of oxidative stress of excitation contraction coupling

ROS have a role in normal signalling as discussed in 1.4.1 however the heart has lower levels of antioxidant enzymes in comparison to other organs (Costa *et al.*, 2013) with catalase activity being particularly lower in the heart (Thayer, 1986). This means that while under normal physiological conditions the heart has the coping mechanisms to neutralise ROS, in states of pathology that result in increased ROS production these mechanisms may not be sufficient and reactions such as those discussed in 1.4.1.2 may occur. This will result in a state of OS.

ROS are known to perturb ECC through a wide range of effects on ECC components (D'Oria *et al.*, 2020). The Ca transient amplitude and contractility in cardiac myocytes have been shown to be reduced by ROS (Goldhaber & Liu, 1994; Greensmith *et al.*, 2010; Kuster *et al.*, 2010) and may be explained by reduction in SR Ca content (Boraso & Williams, 1994; Greensmith *et al.*, 2010; Kuster *et al.*, 2010). As discussed in 1.2.9.2 Ca content of the SR is directly related to systolic Ca (Trafford *et al.*, 1997) and a small change in SR Ca content can result in a significant change to Ca amplitude given the steep relationship.

The reduction in SR Ca content may be attributed to a reduction in SERCA activity due to ROS (Greensmith *et al.*, 2010; Kuster *et al.*, 2010; Morris & Sulakhe, 1997; Reeves *et al.*, 1986; Scherer & Deamer, 1986) which would mean less Ca taken back up into the SR. Furthermore, ROS have been shown to increase NCX activity (Eigel *et al.*, 2004; Goldhaber, 1996; Kuster *et al.*, 2010; Reeves *et al.*, 1986) and as discussed in 1.2.9.4 recent evidence suggests NCX is a far more important regulator than originally thought with this contributing to SR Ca content to a greater extent than SERCA (Bode *et al.*, 2011). In addition to this ROS may also impact the LTCC and have been shown to reduce the voltage gated Ca current therefore Ca influx (Anzai *et al.*, 1998; Boraso & Williams, 1994; Kawakami & Okabe, 1998). The combined effect on both these transporters may account for the reduction in SR Ca content due to ROS which in turn can explain the decrease in Ca transient amplitude and reduced contractility. To identify the presence of oxidative stress in disease states various methods exist.

1.4.4 Measuring oxidative stress

Given the implications of ROS on ECC as discussed in 1.4.3 it is important to ascertain if ROS are increased in cells and if this increase in ROS exceeds the cells coping mechanisms. The impact on a cellular basis in turn effects whole organ function.

In vivo, one method that may be used on a clinical sample is the OXY-adsorbent test. This test measures the total antioxidant capacity by determining the ability of the sample to inactivate the oxidant solution which is added in excess, the results are then compared to the standard calibration curve (Vassalle *et al.*, 2012). This means that by measuring the excess oxidant solution you can indirectly measure the antioxidant capacity of the sample, in principle the higher the antioxidant capacity the lower the OS (Bonanni *et al.*, 2007). However, it can be argued that measuring the antioxidant capacity of a sample is not sufficient to make an indirect conclusion about OS. Various factors such as the consideration on non-enzymatic antioxidants and species variation suggest this method alone does not suffice and a method to determine oxidative damage would also be required (Costantini & Verhulst, 2009).

Methods to detect OS in *vitro* include detecting antioxidant enzyme levels as well as ROS using various techniques. For example, detection of lactate dehydrogenase may be used as an indicator of oxidative stress as leakage of LDH indicates damage to the cell membrane by free radicals (Zhou *et al.*, 2010). The detection of ROS may be achieved through fluorescent probes along with the flow cytometry and fluorescence microscopy (Shen *et al.*, 2013). For example, hydroethidine and 1,3-diphenylisobenzfuran can be used to detect superoxide radicals while 2,7-dichlordihydrofluorescein (DCFDA) can be used for detection of hydrogen peroxide (Gomes *et al.*, 2005). While these probes are commonly used due to their high sensitivity the detection of specific ROS may not be useful in the context of measuring overall OS.

While various methods exist for detecting OS and ROS, measuring these in cardiac myocytes poses new challenges. Isolation of primary cardiac myocytes yield varies, and identification of viable cells must be made based on visual characteristics meaning many methods may include data from unviable cells which will impact findings. To remove the non-viable cells is not always feasible as cardiac myocytes remain in suspension and would require the use of a micromanipulator which would be extremely time consuming (Kosloski *et al.*, 2009). While flow cytometry (FC) may seem a useful tool to sort and detect OS in cardiac myocytes it's use is limited in viable ventricular cardiac myocytes, and many experiments using FC with this cell type require them to be fixed (Diez & Simm, 1998). The major limitation in using FC with live primary ventricular cardiac myocytes is the variety in the cells shapes and sizes as this can impact validity and reproducibility (López *et al.*, 2017). There is the option to wash and filter the cardiac myocytes to select smaller cells as demonstrated by López *et al* (2017) as this would allow the cells to fit through the filter without causing equipment issues but this then presents its own problems with bias. Therefore, for the purposes of this project an alternative method was used as described in section 2.9.

1.5 Anthracyclines

As discussed in 1.1.4 anthracyclines are highly effective anti-cancer drugs with a main caveat of their use being cardiotoxicity.

1.5.1 Anthracycline chemical structure

In 1957 anthracyclines were first derived from the bacteria *Streptomyces*, in south-eastern Italy, with daunorubicin being the first antibiotic discovered (Weiss *et al.*, 1986). Doxorubicin (adriamycin) was isolated in 1967 and has the same anthraquinone and glycoside structure as daunorubicin (Arcamone *et al.*, 1969). The only chemical difference between these two anthracyclines is the hydroxyl group in doxorubicin compared to the methyl group in daunorubicin (DAUN) as highlighted in figure 1.14.



Figure 1.14: Chemical structure of daunorubicin (left) and doxorubicin (right) highlighting the hydroxyl group in doxorubicin compared to the methyl group in daunorubicin from Alves *et al* (2017).

1.5.2 Mechanism of action of anthracyclines

Research into anthracyclines suggest various mechanisms of action for their anti-cancer effects including cellular damage via free radical production, altered intracellular signalling and inhibiting topoisomerase type 2 (Fanous & Dillon, 2016).

1.5.2.1 Topoisomerase type 2 poisoning/inhibition

Topoisomerase II cleaves and relegates double stranded DNA to solve issues such as supercoiling and knotting (Hangas *et al.*, 2018). Studies have shown anthracyclines cause DNA cleavage via activation of topoisomerase II. Different anthracyclines have different DNA binding capabilities, DOX and DAUN are two anthracyclines that are known to bind strongly which is linked to inducing topoisomerase II cleavage (Capranico *et al.*, 1990). It is also important to note that anthracyclines have a strong binding capacity to topoisomerase 2 cleavage complex this is the intermediate that is present when the DNA is broken and can lead to apoptosis of the cell if the cuts are irreversible (Marinello *et al.*, 2018). It has been found that at low concentrations anthracyclines elicit poisoning activity on topoisomerase 2 compared to high concentrations which cause inhibition of activity. This means anthracyclines can intercalate with DNA base pairs as well as stabilising the topoisomerase 2 cleavage complex after DNA cleavage. To summarise, anthracyclines can increase the breaks in the DNA but also prevent ligation and therefore DNA synthesis.

1.5.2.2 Free radical production

Another proposed mechanism of action for anthracyclines is the generation of free radicals. The quinone structure in anthracyclines acts as an electron acceptor in reactions such as those mediated by NADH dehydrogenase, the free electrons then convert quinones in to a semiquinone and subsequently the formation of oxygen free radicals (Goodman & Hochstein, 1977). Anthracyclines can also produce ROS through increasing intracellular free iron (Fe²⁺ and Fe³⁺) which leads to an increase in iron-mediated ROS production (Gammella *et al.*, 2014).

It has been shown that high levels of exogenous hydrogen peroxide cause DNA damage and severe cytotoxicity while low levels of exogenous H_2O_2 lead to a reduction in ATP, a delay in toxicity and the cell is seen to arrest in mitosis and can lead the cell to stop dividing known as senescence (Panieri *et al.*, 2013). This shows that by increasing the production of ROS, anthracyclines can stop division of cancer cells as well as induce DNA damage.

1.5.2.3 Altered intracellular signalling

Anthracyclines may also alter intracellular signalling either as a direct result of their ability to inhibit topoisomerase II or via another mechanism.

Formation of neutrophil extracellular traps are part of the innate immunity in response to factors such as foreign bodies and an increase in pH (Khan *et al.*, 2018). Neutrophil extracellular traps (NETs) are DNA-protein complexes that as well as responding to foreign bodies such as infection also combat conditions such as sepsis and cancer (Khan *et al.*, 2019). However, studies also show that NETs have an advantageous effect on cancer cells as they have been found in close proximity to tumour cells such as Lewis lung carcinoma and entrap these cells to aid survival as well as inhibiting immune cells (Demers & Wagner, 2014). Research is still ongoing into the effect of NETs on cancerous cells. Anthracyclines can suppress pathways that release NETs such as NADPH oxidase (NOX)-dependent and independent pathways (Khan *et al.*, 2019). If the cancer aiding properties of NETs outweigh the protection they provide from cancer, the ability of anthracyclines to suppress NETs would prove beneficial in preventing the survival of cancer cells.

Hypoxia-inducible factor 1 (HIF-1) overexpression is linked to the increased growth of tumours as well as vascularisation which aids metastasis (Semenza, 2007). Research has shown anthracyclines such as DOX and DAUN inhibit HIF-1 (Lee *et al.*, 2009). Anthracycline achieves this is by preventing HIF-1 from binding to DNA as well as inhibiting the transcriptional activity of HIF-1, this reduction in HIF-1 transcription can lead to a downstream reduction in other proteins that promote vessel formation such as vascular endothelial growth factor (Lee *et al.*, 2009). By reducing the expression of HIF-1, anthracyclines may prevent the growth of blood vessels around cancerous tumours and thereby reduce tumour growth and spread.

1.6 Anthracycline-induced cardiotoxicity

As discussed in 1.5, anthracyclines are effective anti-cancer agents and are commonly used in treatment regimes, however the off-target effect, particularly cardiotoxicity needs to be addressed. Cardiac complications are the most common cause of severe health issues in childhood cancer survivors (Trachtenberg *et al.*, 2011).

1.6.1 Clinical manifestation of anthracycline-induced cardiotoxicity

As discussed in section 1.5.2 anthracyclines have multiple mechanisms of action. This makes them effective anti-cancer drugs however, the effects are not exclusive to cancer cells. A major limitation of these drugs is the negative effect on the heart. Childhood cancer survivors treated with anthracyclines have a 15-fold increased risk of heart failure (Oeffinger *et al.*, 2006). The identification of cardiotoxicity is subcategorised. Occurrence of clinical heart failure in those treated with a low cumulative dose of anthracyclines is between 2 -4 % while cardiac injury has been found in 30 - 35 % of patients (McGowan *et al.*, 2017).

Clinically, left ventricular ejection fraction (LVEF) is used to quantify ventricular function in the heart with a poorer prognosis associated with a lower percentage (Wehner *et al.*, 2019). Cardiotoxicity is recognised by a decrease of 10 % or more in LVEF (Nakayama *et al.*, 2020). In childhood cancer survivors with subclinical cardiac dysfunction a decrease in left ventricular mass and wall thickness was detected as well as decreased contractility (Lipshultz *et al.*, 1991). This study also found 57 % of patients had an increase in afterload due to reduced wall thickness. The afterload is the pressure in the vasculature that the heart must exceed to successfully eject blood from the ventricles and is determined by the systolic pressure; an increase in afterload results in a decrease in stroke volume (LaCombe *et al.*, 2022). As well as left ventricular systolic dysfunction patients may suffer from myocardial ischaemia and hypertension (Galán-Arriola *et al.*, 2021). Myocardial ischemia may occur due to microvascular dysfunction, stenosis and mitochondrial dysfunction (Marzilli *et al.*, 2020). The

clinical manifestations though similar vary depending on the cumulative dose and age at treatment.

1.6.2 Acute and long-term anthracycline-induced cardiotoxicity

Anthracyclines are known to elicit acute and long term cardiotoxicity, with acute symptoms being present during or less than a year after treatment has ended and this increases the risk of the patient suffering from late cardiotoxicity (Grenier & Lipshultz, 1998). Chemotherapy such as DOX and DAUN are linked to adverse cardiac events including arrhythmias, myocarditis and heart failure (Lipshultz *et al.*, 2013). A longitudinal study recorded various cardiac parameters following DOX treatment in 115 childhood survivors of acute lymphoblastic leukaemia and some of the findings are illustrated in figure 1.15 (Lipshultz *et al.*, 2005). Lipshultz *et al* (2005) identified an initial reduction in left ventricular contractility which recovered over the next 6 years only to reduce again (fig 1.15A), this pattern of improvement followed by a reduction was evident regardless of cumulative dose of anthracycline. Following completion of DOX treatment, the left ventricular wall thickness was also reduced and became thinner over the course of the 15 years of the study with measurements after 6 years being significantly less than normal (Lipshultz *et al.*, 2005).

It stands to reason that although the cardiotoxic effects seen acutely during and shortly after anthracycline treatment improve over time they then decline again and can be a good indicator of future heart problems as a result of the treatment.



Figure 1.15: The effect of doxorubicin on left ventricular contractility and wall thickness. (A) mean left ventricular cardiac measurements from 115 survivors of ALL treated with doxorubicin with the upper and lower 95 % confidence interval. (B) mean left ventricular wall thickness for the same cohort. Adapted from Lipshultz *et al* (2005).

1.6.3 Adult versus childhood anthracycline-induced cardiotoxicity

The clinical features of cardiotoxicity following anthracycline treatment may present differently in children compared to adults. Adults are more likely to present with dilated cardiomyopathy whereas children may present with dilated cardiomyopathy which then develops into restrictive cardiomyopathy, however both are known to suffer with reduction of contractility (Giantris *et al.*, 1998). Dilated cardiomyopathy is characterised by dilated left ventricle with normal or thinning of the wall whereas restrictive cardiomyopathy is due to failure of the myocardium to relax (Davies, 2000). The difference in how the toxicity presents is likely due to the survival time following treatment as well as continuing growth. The proliferation of cardiomyocytes slows down as humans age, however these myocytes will increase in size meaning that in adults this growth will have already occurred whereas in children anthracycline treatment may impact this growth and proliferation (Mollova *et al.*, 2013). Not only may anthracyclines reduce cardiomyocyte number, but they may alter their function via various mechanisms.

1.6.4 The mechanisms of anthracycline-induced cardiotoxicity

The alteration to calcium handling in cardiomyocytes during and following anthracycline treatment may be due to a variety of mechanisms. As previously discussed, these drugs lead to an increase in ROS which have been theorised to activate Ca/calmodulin dependent protein kinase II (CaMKII) which then leads to SR Ca leakage (Sag *et al.*, 2011). CaMKII is expressed abundantly in the heart and phosphorylates various proteins involved in ECC (Bers & Grandi, 2009). If CaMKII is inhibited a leak of SR Ca has been found to be reduced indicating the importance of this protein in this process (Curran *et al.*, 2007). A calcium leak from the SR can contribute to a decrease in systolic Ca transient and lead to an abnormal heartbeat, this is termed an arrhythmia (Curran *et al.*, 2007).

Alternatively, the anthracycline may bind directly to a channel within the cardiac myocyte. Doxorubicin and its metabolite, doxorubicinol can bind to the RyR and at low concentrations (0.5 and 2.5 μ M) can activate the channels as well as causing secondary inhibition of the receptor, the activation response has been found to be reversible however the inhibition has not (Hanna *et al.*, 2014). Hanna *et al* (2014) found when treated with 2.5 μ M DOX the maximal RyR activation was seen after 2.5 minutes, then after 16 minutes inhibition was observed, they believed the inhibition to be due to RyR thiol modification. Alternatively, this inhibition could be due to the RyR becoming insensitive to the changes in luminal [Ca] as doxorubicinol disrupts calsequestrin 2 (CSQ2) which is a binding protein in the SR capable of altering the activity of the RyR (Hanna *et al.*, 2017).

Hanna *et al* (2014) also found low concentrations of doxorubicinol reduced SERCA2a activity but not DOX suggesting that the metabolite acts with greater efficacy than the original compound. Autoregulation may compensate for an increase in open probability in the RyR however this is assuming only one component has been affected, if SERCA is also affected the cell may not be able to compensate and this can lead to major consequences. While one of the mechanisms of action for anthracyclines may be the production of ROS this may also be linked to how they cause cardiotoxicity. When rat cardiomyocytes were treated with DOX, ROS were produced immediately and increased over time and through the use of various channel blockers it was found that ROS production due to DOX opens RyR and causes an increase in systolic [Ca] which in turn generates more ROS (Kim *et al.*, 2006a). As discussed in 1.3.1 these free radicals have been linked to heart failure. This highlights some of the information research has discovered regarding anthracycline induced heart failure however these studies are often conducted on individual channels rather than the intact cardiac myocyte.

1.6.5 Prevention of anthracycline induced heart failure

Current research to prevent cardiac complication due to anthracycline treatment involves modification of anthracycline doses, use/production of less cardiotoxic analogues of anthracyclines and the use of cardioprotective agents alongside anthracyclines (Nathan *et al.*, 2016).

There is ongoing research into altering the chemical structure of certain anthracyclines via the addition of polyethylene glycol (PEG) or encapsulation to make delivery targeted and reduce concentrations of circulating anthracyclines such as DOX (Volkova & Russell, 2011). PEG is a polymer that can be added to the surface of liposomes which are used as nanocarriers to decrease the drugs interactions with blood components (Chen *et al.*, 2020). Pegylated-liposomal DOX is associated with a lower risk of DOX induced HF when looking at a retrospective group of patients who have received large cumulative doses (Skubitz *et al.*, 2017). However, issues may arise with PEGylated liposomal DOX if the patient has anti-PEG antibodies as this can lead to activation of the complement immune system and cause rapid release of DOX which will alter the safety for use in patients (Chen *et al.*, 2020). By creating a

targeted drug delivery, it is thought the toxic effect of anthracyclines on the heart will be minimised.

Dexrazoxane (DEX) is a drug used clinically in combination with certain anthracyclines to try and combat the cardiotoxic effects. DEX is an iron chelator which may work by preventing iron-dependent oxidative stress and/or via degrading topoisomerase II in heart tissues which prevents DNA damage (Deng et al., 2014). In the presence of iron DOX may undergo redox cycling resulting in ROS production, as an iron chelator DEX reduces mitochondrial iron accumulation and therefore prevent iron mediated ROS production (Ichikawa et al., 2014a). Use of DEX with DOX in children with acute lymphoblastic leukaemia showed their left ventricular function to be closer to normal than those treated with DOX alone indicating that dexrazoxane provides some form of long-term protection from the cardiotoxic effects of DOX (Lipshultz et al., 2010). Impaired left ventricular function is often used as an indicator of heart failure (Harbo et al., 2020). Another study conducted on children with solid tumours who received DEX along with DOX showed fewer cardiac events related to cardiotoxicity than those treated with DOX alone (Choi et al., 2010). Choi et al (2010) found that 13 cardiotoxicities were found in the DEX treated group compared to 22 cardiotoxicities in patients treated with DOX alone, however the follow-ups were at 54 months and 86 months respectively meaning there may have been more cardiotoxicities reported after 86 months following DEX treatment. Furthermore, a risk of secondary malignancies (acute myeloid leukaemia and myelodysplastic syndrome) was identified in children who received DEX (Tebbi et al., 2007). The co-treatment of DEX with anthracyclines appears to be beneficial though the exact mechanism of action is unclear. The effects in children are of greater concern due to the risk of secondary malignancies.

Various theories exist and studies have expanded knowledge on how the cardiotoxic effects of anthracyclines can be limited. However, few (if any) look at the effects of these combination or altered treatments on a cardiac cellular basis. Given the cardiac implications of anthracycline treatment the use of positive inotropic drugs may be of interest. These drugs 42 increase the force of contraction by increasing free intracellular Ca (Scholz, 1984) and provide an avenue of combating anthracycline cardiotoxicity.

1.7 Study main aims

The role of anthracycline-induced cardiotoxicity is not a new concept and various studies have sought to investigate how this occurs. However, many of these studies use small animal models or isolated ion channels which have limitations as previously mentioned. It therefore stands to reason that the impact on the components of excitation-contraction coupling may be altered to a different extent in an intact large animal model. Furthermore, the majority of these studies focus solely on doxorubicin with the assumption other anthracyclines will behave in the same way. This study will address the effects of another commonly used anthracycline that is believed to be less cardiotoxic; daunorubicin.

While previous research has highlighted the increase in ROS due to anthracyclines and implicated the role of ROS in altered cardiac cellular function there are limited studies on the impact of anthracycline-induced ROS on excitation-contraction coupling components in intact cardiac myocytes and the pathways involved.

This project aimed to:

- Characterise the effects of doxorubicin and daunorubicin on excitation-contraction coupling and DAUN on ECC in the integrated model of the intact sheep ventricular myocytes.
- 2. Determine whether DOX and DAUN increase oxidative stress in sheep ventricular myocytes.
- 3. Determine to what extent changes to excitation-contraction coupling are dependent on oxidative stress.
- 4. Determine to what extent elevations in oxidative stress are dependent on NADPH oxidase and xanthine oxidase.

5. Measure the effects of DOX and DAUN on excitation-contraction coupling in myocytes isolated from a sheep with heart failure.

Chapter 2 General Methodology

2.1 The use of animal models to study excitation-contraction coupling

One of the main issues in studying heart disease states or the effect of drugs on cardiac function is the limitation of research methods. Due to ethical and practical constraints experiments on the human cardiac system are only possible in the rare cases of donated organs which poses its own limitations in terms of control data. Animal models offer a useful alternative.

Heart rates vary between species and generally the larger the mammal the slower the heart rate (Zhang & Zhang, 2009). Humans have a heart rate of around 80 beats per minute (bpm) compared to rats which are approximately 250 bpm and sheep which are 103 bpm (Noujaim *et al.*, 2004). There is variation in calcium handling between species and so there is a difference in the time of contraction and relaxation (Janssen & Periasamy, 2007).

While small rodents such as rats are easier to handle and house than larger mammals such as sheep, they have very high heart rates which means their hearts are different on a cellular level to facilitate the rapid contraction. For example, the action potentials of ventricular myocytes differ in rodents compared to humans as the outward potassium current and density of potassium channels differs meaning rodents such as rats have a large initial repolarisation phase and short plateau phase at lower membrane potentials compared to humans (Joukar, 2021).

Large animal models such as sheep share cardiac similarities to humans. For example, the removal of Ca via SERCA activity in sheep is similar to that of humans at 81.5 % (Dibb *et al.*, 2004) which as discussed in section 1.2.8.2 is responsible for 70 – 90 % of calcium removal. This means sheep serve as a good animal model for cardiovascular research particularly in terms of investigating excitation-contraction coupling.

46

2.2 Primary myocyte isolation

2.2.1 Ethics

All procedures were in accordance with the Animals (Scientific Procedures) Act, UK, 1986 and Directive 2010/63/EU of the European Parliament. All procedures carried out at the University of Manchester have ethical approval from the University of Manchester Welfare and Ethical Review board and the experiments have ethical approval from the University of Salford ethical review board.

2.2.2 Tissue preparation and cell isolation

Cells were isolated from the left ventricle of young (approx. 18 month), healthy sheep or sheep with tachypacing induced heart failure by collaborators at the University of Manchester. The sheep received heparin intravenously (20 ml) to prevent blood clots and then received a lethal injection of sodium pentobarbitone (200 mg/kg). Incisions were made to access and remove the heart to allow for cell isolation.

Langendorff apparatus was prepared as shown in figure 2.1, and all solutions were incubated in a water bath at 37°C. Ca free solution (see table 2.1) was used to rinse the heart and remove excess blood prior to dissection of the ventricle and atria. The left ventricle was cannulated via the coronary artery onto the langendorff apparatus (see figure 2.1). Ca free solution was perfused through the left ventricle using a peristaltic pump for at least 10 minutes to clear intravenous blood and prevent blood clots forming in the tissue. Collagenase type 1 (Worthington, 0.096 mg/ml) and protease (Sigma, 0.018 mg/ml) were then added to the perfusing solution to commence enzymatic digestion of the tissue. Collagenase digests the extracellular matrix and collagen, allowing the dissociation of cells. Protease facilitates this while digesting the membrane. Digestion ranged from 5-15 minutes and was judged by visual observation of pericardium viscosity to achieve optimal cell quality.



Figure 2.1: Langendorff apparatus used for cardiac myocyte isolation. (A) taurine solution. (B) Ca free solution. (C) water bath. (D) multi-way taps. (E) heat exchanger. (F) peristaltic pump. (G) cannula leading to sheep heart into the coronary artery.

Table 2.1: Calcium free/ Isolation solution		
Constituent (Fisher or Sigma	Concentration (mM)	
Aldrich)		
Bovine Serum Albumin (BSA)	1%	
2,3- Butanedione Monoxime	10	
(BDM)		
Sodium Chloride (NaCl)	134	
HEPES	10	
Glucose	11	
Potassium chloride (KCl)	4	
Magnesium sulfate (MgSO ₄)	1.2	
Monosodium Phosphate	1.2	
(NaH ₂ PO ₄)		
pH adjusted to 7.34 using 1M NaOH or 1M HCl prior to		
volumizing.		

After digestion, taurine solution (see table 2.2) was perfused through the tissue for 10 minutes to reintroduce Ca and wash out any enzyme from the tissue. The endocardium and epicardium were then removed, and the tissue was cut into small pieces and transferred to fresh taurine solution. The tissue was agitated using a pasteur pipette and filtered through 200 μ M pored mesh. The cell suspension was then inspected under an inverted microscope to determine experimental suitability using visual characteristics such as distinctive rod shape and striations as shown in figure 2.2.



Figure 2.2: Images of experimentally suitable and non-experimentally suitable cells. Suitable cells are distinguished by the rod shape and striations.

Table 2.2: Taurine solution		
Constituent (Fisher or Sigma Aldrich)	Concentration (mM)	
BSA	1 %	
BDM	10	
NaCl	108	
HEPES	10	
Glucose	11	
KCI	4	
MgSO ₄	1.2	
NaH ₂ PO ₄	1.2	
Taurine	50	
Calcium chloride (CaCl ₂)	0.1	
pH adjusted to 7.34 using 1M NaOH or 1M HCl prior to volumizing.		

2.3 Experimental solutions

Sydney Ringer developed the first form of solution used in cardiac experiments and noted that when a frog heart was perfused with water contraction became more rapid and died quickly. From this observation Ringer added sodium chloride (NaCl), potassium chloride (KCl) and calcium chloride (CaCl₂) to produce physiologically normal contractions (Ringer, 1885). These contractions lasted for ~1.5 hours however the addition of sodium bicarbonate strengthened the contractions due to altering the pH of the solution (Ringer, 1885). The solution was further developed by Locke who observed adding glucose as an energy source for metabolism extended the survival of cardiac myocytes (Locke, 1895). The current solution used experimentally is called Normal Tyrode's (NT) solution and has a similar composition to the Ringer-Locke solution with the addition of magnesium chloride (MgCl₂) and sodium phosphates (Lee, 1981). Probenecid was added to reduce fluorophore leaking out of the cell. Addition of taurine has been found to retain Ca (Dolara *et al.*, 1973). Taurine containing solution is used during cell isolation and provides an intermediate calcium concentration.

All solutions were kept for a maximum of 4 days prior to experiments at 4°C. Glucose and calcium were added on the day of use to limit contaminant growth.

Table 2.3: Normal Tyrodes solution		
Constituent (Fisher or Sigma Aldrich)	Concentration (mM)	
NaCl	140	
HEPES	10	
Glucose	10	
Probenecid	2	
KCI	4	
MgCl ₂	1	
CaCl ₂	1.8	
Probenecid dissolved in 1M NaOH before addition to solution.		
pH adjusted to 7.34 using 1M NaOH or 1M HCl prior to volumizing.		

2.4 Preparation of anthracyclines

Doxorubicin hydrochloride (Fisher) was dissolved in DMSO (Fisher) to create a stock concentration of 2 mM and aliquots were stored at -20°C. All further dilutions of DOX were achieved using NT and ensured the concentration of DMSO present was less than 0.5 % 50

initially, most experiments were carried out at 0.001 μ M which contained < 0.001% DMSO. The percentage of DMSO would not alter cell osmolality, pH, viability, action potential or field potential (Hyun *et al.*, 2017). Daunorubicin hydrochloride (Fisher) was prepared in the same manner.

2.4.1 Control of experimental cell environment / perfusion

Preparation of the experimental apparatus involved flushing the tubing with experimental solution to ensure no bubbles were present, as these can impact cell function and solution flow.

Solutions were preheated to 37° C using a BubbleStop syringe heater. ValveLink was used to control which solution perfused on to the bath and allowed a seamless switch between solutions. Flow rate of solutions was controlled by solution level present in the syringe and aperture of the tip (250 µm). Experiments were conducted under perfusion conditions.

After cells were loaded with Fura-2, a small volume of cells (~0.5 ml) was pipetted on to the cell bath and allowed to settle for 5-10 minutes. Cells were brought into focus using the epi-fluorescent inverted microscope at x 40 magnification with the use of immersion oil. Cells that appeared rod shaped, striated and non-granulated were classified as viable. NT was perfused into the bath and field stimulation was used to stimulate cell contraction.

2.5 Field stimulation of cardiomyocytes

A circuit was formed between the electrodes on the cell bath, a DS2A isolated voltage stimulator and a DG2A train delay generator. The DS2A isolated voltage stimulator was set to the lowest voltage required to stimulate cells, typically between 55 – 75 V. The train delay generator controlled the frequency at which the cells were paced, at the voltage set on the isolated voltage stimulator. The pacing frequency used was 0.5 Hz with a duration of 100 ms.

2.6 Measurement of intracellular calcium

2.6.1 Equipment configuration

Various pieces of equipment are required to measure intracellular Ca and contractility simultaneously, the roles of each piece of equipment are described in table 2.4.

Table 2.4: Equipment to measure intracellular Ca and contractility	
Equipment	Purpose
DG2A train delay generator	Controls the frequency of electrical
	stimulation
DS2A isolated voltage stimulator	Outputs electrical stimulation to the cell
	bath
Valvelink system	Allows seamless switch between solutions
	perfusing into the cell bath
ThermoClamp [™] -1	Enables heated perfusion of solutions into
	the cell bath
Photomultiplier tube	Receives and amplifies fluorescence signal
	from the cell
Multistream pro	Switches LED excitation between 340 nm
	and 380 nm at a set frequency
Data system interface	Communicates between the hardware and
	computer
Dual OptoLED power supply	Allows two LEDs independently
Myocam-S power supply	Powers the Myocam-S
Epi-fluorescent microscope	A series of filters and mirrors that enables
	excitation/emission of Fura-2 as illustrated
	in figure 2.4
BubbleStop syringe heater	Heats the solutions for perfusion and
	prevents bubbles in the tubing
Myocam-S high speed camera	Video detection of sarcomere which allows
----------------------------	---
	measurement of sarcomere length in
	combination with SarcLen Sarcomere
	Length Acquisition Module (IonOptix)
Anti-vibration table	Reduces vibrations and disturbance
Blackout case	Limits other light sources

2.6.2 Loading of cardiomyocytes with Ca fluorophore

Primary sheep cardiomyocytes were stored at room temperature. Cells were acclimatised to experimental concentrations of calcium gradually by storing in a solution of half NT (see table 2.3) and half taurine solution (see table 2.2) to produce an intermediary Ca concentration of 0.95 mM. Experiments were carried out in NT at a Ca concentration of 1.8 mM (see table 2.2). The acetoxymethyl ester (AM) form of the fluorophore Fura-2 (Invitrogen) was used to measure intracellular Ca. The AM group allows the otherwise negatively charged compound to enter through the membrane of the cell, where the cell esterases then cleave the AM groups leading to strong fluorescent indicators to be retained within the cell (Grynkiewicz *et al.*, 1985). Cells were incubated with Fura-2-AM at 0.1 μ M for 10 minutes at room temperature. Loading of cells ceased after 10 minutes when an excess of solution was added. The cells were then kept in the dark at room temperature for a further 30 minutes to allow for de-esterification of Fura-2. After 30 minutes cells were ready for experimental use.

2.6.3 Excitation and emission protocol

Fura-2 is a dual excitation (340/380 nm) indicator. At 340 nm the fluorescence of Fura-2 increases with an increasing Ca concentration. At 380 nm, the fluorescence of Fura-2 decreases with an increasing Ca concentration (Grynkiewicz *et al.*, 1985). This is illustrated in figure 2.2. Fura-2 emits at 510 nm in response to excitation at both wavelengths. The ratio between these signals is not affected by loss of signal or photobleaching unlike non-

ratiometric fluorophores. Therefore, it provides a reflection of real changes in intracellular Ca.

Fura-2 was excited sequentially at 200 Hz with the multistream pro at 340 and 380 nm using OptoLED light sources. Emitted fluorescence was measured via a 515 nm long-pass filter using a photomultiplier tube coupled to a photometry system.



Figure 2.3: Excitation spectra of Fura-2 from Trabalza et al (2021).

2.6.4 Epi-fluorescence microscope configuration for Ca measurement The configuration of the epi-fluorescence microscope is illustrated in figure 2.4. The use of Fura-2 to detect relative changes in intracellular [Ca] means it is important to limit exposure to light from outside sources. A Faraday cage surrounds the microscope, and a black curtain covers the front to prevent outside light. As an extra precaution during experiments the lights in the room are switched off and to protect the photomultiplier tube.

As mentioned in section 2.6.3 Fura-2 excites at 340/380 nm and this is achieved by placing band pass filters in front of the LED light sources and alternating at a frequency of 200 Hz by the Multistream Pro (Cairn). The 370 nm long pass dichroic mirror allows the 380 nm light beam to pass, and this then converges with the 340 nm light beam before being directed to the cell bath by the 425 nm and 500 nm long pass dichroic mirrors. After excitation, the light emitted due to Fura-2 in the cells is deflected by standard mirrors and the 600 nm long pass dichroic mirror into the PMT. The PMT amplifies this fluorescent signal from the cell before it is integrated in the IonWizard (IonOptix) software for analysis as described in section 2.4. The amplification from the PMT was kept consistent throughout experiments while the intensity of LED excitation of Fura-2 was altered on a cell-by-cell basis due to variation in fluorophore uptake.



Figure 2.4: Configuration of epi-fluorescence microscope illustrating filters and light pathways for use with Fura-2.

2.7 Measurement of sarcomere length

An Ion Optix MyoCam-S high speed camera was utilised to detect changes in sarcomere length. Data regarding sarcomere length was acquired by the SarcLen Sarcomere Length Acquisition Module along in combination with the IonWizard 6.5 system. Once a cardiac myocyte was positioned in the field of view with the striations aligned vertically a box of appropriate size was positioned around the striations (sarcomeres). This determined the sarcomere length that was measured. A density trace was produced, and the peak produced converted in the software to pixels per sarcomere.

2.8 Data Analysis for calcium and sarcomere length

2.8.1 Diastolic and systolic Ca

Analysis of diastolic and systolic Ca was conducted by exporting 5 averaged Ca transients from IonWizard software (Ion Optix) into Ca Multi which is an excel program to analyse transients obtained using fluorescent indicators (Greensmith, 2014a). This was done for each steady state condition (e.g. steady state control, steady state drug etc). The program allows the identification of parameters such as peak Ca and diastolic Ca using cursors, the program then uses this information to determine Ca amplitude. The background (no cell) for each cell was also taken after each experiment by moving the cell out of field of view. Both signals obtained for the no cell from the 340 nm and 380 nm were averaged and a ratio was determined by dividing 380 by 340. The program subtracts the background fluorescence (no cell) from the raw fluorescence, so fluorescence analysed is that from the cell.

2.8.2 Sarcomere length

The average of 5 sarcomere lengths at each steady state were exported from IonWizard software to the Ca Multi program described in section 2.8.1 and diastolic and systolic length was determined. These values also allow the program to calculate the peak amplitude of contraction.

The rate of sarcomere relaxation and shortening was analysed using the excel program APD single 2010 v2. The relaxation of the sarcomere was determined by calculating the time taken for 90 % rate of decay minus 10 % rate of decay.

2.8.3 Statistical Analysis for Ca handling and contractility

The raw data for each cell was extracted from the Ca Multi program and input into a new excel spreadsheet. This data was then normalised against each cells own control state. Statistical analysis was carried out on IBM SPSS. Unless otherwise stated a one-way ANOVA

with Tukey post-hoc test was carried out and a statistically significant result was identified if P < 0.05.

2.9 Measurement of intracellular oxidative stress

2.9.1 Loading and preparation of cardiomyocytes for oxidative stress experiments

Throughout the oxidative stress experiments cells were stored at room temperature in half NT (see table 2.3) and half taurine solution (see table 2.2) at Ca concentration of 0.95 mM. Cardiac myocytes were split into tubes corresponding to treatment conditions and treated with anthracyclines of varying concentrations for a set period. After incubation the drug solution was removed from the cells. Cells were then loaded with 5 μ M CellROX Deep Red (Invitrogen) (Ex 640/Em 665) for 30 minutes. CellROX deep red is a cell permeable fluorophore that becomes fluorescent upon oxidation and is localised to the cytoplasm (Manoil & Bouillaguet, 2018). CellROX Deep Red has the advantage of being used with viable cells and detects increases in oxidative stress (Habibalahi *et al.*, 2020). Cells pelleted during this period allowing the solution containing fluorophore to be removed before being resuspended in fresh solution.

Following resuspension in fresh solution 100 μ l of cell suspension was pipetted into wells on a 96 well plate and allowed to settle before images were captured using the Biotek cytation imaging system.

2.9.2 Image capturing and analysis

The Biotek cytation imaging system 3 was set to 25° C (room temperature) and focused to capture images of cells using both the brightfield and Texas red filter cube at x 10 magnification. Gain and intensity were kept consistent for all experiments.

Analysis was conducted using imageJ software. Brightfield images were overlayed on the corresponding Texas red image and viable cells were identified as previously mentioned by shape, striation, and granularity. The integrated density and area of the cells were measured and areas surrounding the cells selected to subtract the background. The total corrected cellular fluorescence (TCCF) was calculated by subtracting the background fluorescence reading ((integrated density – (area of cell x mean fluorescence of background reading)) (McCloy *et al.*, 2014).

2.9.3 Statistical analysis for oxidative stress data

For these experiments each plate (animal) determined the n. The average fluorescence for each treatment condition was calculated and normalised against the untreated average for each plate. Statistical analysis was carried out on IBM SPSS. Unless otherwise stated a one-way ANOVA with Tukey post-hoc test was carried out and a statistically significant result was identified if P < 0.05.

Chapter 3

The effects of doxorubicin and daunorubicin on Ca handling and contractility in sheep ventricular myocytes

3.1 Introduction

As discussed in detail in 1.1.4 and 1.5, anthracyclines are highly effective against many cancer types, so frequently form part of treatment strategies. The most commonly used anthracyclines are doxorubicin (DOX), daunorubicin (DAUN), idarubicin and epirubicin (Armenian & Bhatia, 2018). Of these, epirubicin is predominantly used for the treatment of breast cancer (Levine, 2000). While doxorubicin, daunorubicin and idarubicin are used for the treatment of childhood cancer – the focus of this project – the use of idarubicin is relatively less common (Feijen *et al.*, 2019). For this reason, doxorubicin and daunorubicin were selected for this investigation. As per 1.1.4, the anti-cancer effectiveness of these drugs is not disputed however, early and late cardiotoxic effects remain a considerable limitation. The extent to which doxorubicin and daunorubicin-mediated *cell* dysfunction contributes remains unclear.

3.1.1 Anthracyclines and excitation-contraction coupling

As discussed in 1.2.1 whole heart function is dependent on that of the single cardiac myocyte. As such, if we are to fully understand how anthracyclines cause cardiotoxicity, it is essential to elucidate their effects on ECC. Limited, previous studies provide an insight into the effects of anthracyclines on the many components of ECC.

In terms of global cell function, several studies agree that doxorubicin reduces the Ca transient amplitude and contractility in a variety of models (Maeda *et al.*, 1997; Sag *et al.*, 2011; Temma *et al.*, 1996; Zheng *et al.*, 2011). However, the *direct* effects of daunorubicin on these inotropic indicators have yet to be investigated. For doxorubicin, the reduction in Ca transient amplitude and contractility may be explained by alterations to the components of ECC, though again, previous studies are remarkably limited.

As discussed in section 1.6.4 calcium-calmodulin protein kinase (CaMKII) regulates phosphorylation of various proteins of ECC. DOX increases phosphorylation of CaMKII which

is implicated in cardiomyocyte apoptosis. However, this is suppressed by blocking LTCC (Ikeda *et al.*, 2019a). CaMKII activity is regulated by intracellular Ca, these findings suggest that DOX increases intracellular Ca via increased LTCC activity thereby increasing activation of CaMKII pathways. Others show DOX directly increases L-type Ca channel activity leading to an increase in intracellular Ca (Keung *et al.*, 1991). This is surprising given negatively inotropic effects of DOX, observed by others, at cell level. This inconsistency may be explained by effects of anthracyclines on SR Ca content and the RyR. Evidence shows DOX causes a reduction in SR Ca content (Ondrias *et al.*, 1990; Sag *et al.*, 2011). This is also the case for DAUN (Shadle *et al.*, 2000).

Studies in planar membranes reveal a concentration-dependent biphasic effect on the RyR (Hanna *et al.*, 2014). Here, low concentrations of DOX (<2.5 μ M) increase the open probability (activate) the RyR whereas higher concentrations produced an irreversible inhibition. *In vivo* studies show DOX to increase (Llach *et al.*, 2019) and decrease (Gambliel *et al.*, 2002; Olson *et al.*, 2005) RyR expression. DAUN produced a time-dependent biphasic effect on RyR activity (Hanna *et al.*, 2011b). Here, short-term exposure (10 minutes) to concentrations between 0.5 and 10 μ M increased RyR activity. However, at higher concentrations (\geq 2.5 μ M), an inhibitory effect developed after 20 minutes. These studies highlight a controversy surrounding the effects of anthracyclines on RyR.

The effects on the SR Ca content and RyR may be explained by alterations to the Ca removal pathways (see 1.2.9 for review). The expression of SERCA has been found to be reduced following DOX treatment (Arai *et al.*, 1998). Furthermore, a modulatory effect was observed by Hanna *et al* (2014) who show low concentrations of doxorubicinol (a metabolite of DOX) reduce SERCA2a activity. However, this was not the case for DOX itself suggesting an important role for DOX metabolic products.

A shift in the relationship between systolic Ca and force of cell contraction may also explain the apparent negatively inotropic effects of DOX and DAUN. This may occur due to a change in myofilament sensitivity. While the impact of DOX on myofilament sensitivity has been 62 studied in small animal models, whether this sensitivity is increased, decreased or unaltered remains controversial. In one study, DOX did not alter myofilament sensitivity (Boucek *et al.*, 1993) while in others, a decrease was observed (Chakouri *et al.*, 2020). Whether DAUN alters myofilament remains unknown.

While these previous studies provide certain insight into anthracycline cardiotoxicity there are many limitations. For instance, most studies use doxorubicin on the assumption that other anthracyclines will produce similar effects. Clearly, this isn't the case yet few studies have investigated the impact of daunorubicin on calcium handling. Furthermore, many of these studies use artificial planar membranes or small animal models and look at components individually. This makes integration of discrete findings into a unified model problematic. Our experiments overcome this by using cardiomyocytes from a large animal model. This integrative approach allows all aspects of ECC to be studied in the dynamic setting of a physiologically intact cell.

3.1.2 An experimentally relevant anthracycline concentration

In vitro studies are conducted on individual intact cardiac myocytes so the experimental concentrations of anthracyclines must not alter cell viability during experiments designed to elucidate mechanistic effects. Clinically, plasma concentrations of DOX following treatment in female patients reached levels of 630.4 ± 22.1 ng/ml to 39.8 ± 15.3 ng/ml after 0.66 and 24.66 hours respectively (Barpe *et al.*, 2010). We do not know if the cells of the hearts in patients are exposed to these concentrations, nonetheless, this information provides a useful indication.

In cultured cardiac myocytes (both primary and iPS-derived) a considerable range of concentrations have been used (0.01 to 10 μ M) (Barpe *et al.*, 2010; Burridge *et al.*, 2016; Chaudhari *et al.*, 2016; Hanna *et al.*, 2014; Hanna *et al.*, 2017; Kim *et al.*, 2006a; Vavrova *et*

al., 2013; Zhao & Zhang, 2017). In iPS derived cardiomyocytes, DOX-dependent cytotoxicity became evident at 0.5 μ M with an IC₅₀ of 3.5 μ M (Zhao & Zhang, 2017).

While the cellular effects of anthracyclines may differ, it remains useful to consider DOX as a reference compound to develop working concentrations for experimental methodology (Chaudhari *et al.*, 2016).

3.1.3 Summary

Given that anthracyclines are unquestionably linked to cardiac toxicity, there remains significant gaps in the knowledge surrounding their effects on cardiac myocytes. This is critically important if we are to (1) fully understand anthracycline mediated cardiotoxicity and (2) design new therapeutic strategies that allow the continued use of anthracyclines with mitigation of cardiotoxicity.

3.1.4 Aims

- Establish a suitable working AC concentration.
- Use integrative analysis to measure the effects of DOX and DAUN on ECC in the dynamic setting of intact sheep ventricular myocyte.
- Investigate the effect of doxorubicin and daunorubicin on myofilament sensitivity.

3.2 Methods

3.2.1 Cell preparation

Primary left ventricular cardiomyocytes were isolated from healthy, ~18-month-old sheep (see section 2.2.2). Cells were loaded with fura-2 as described in section 2.6.2 then stored in a solution produced by mixing NT and taurine (see 2.6.2) solution in equal parts to produce an intermediate Ca concentration of 0.95 mM.

3.2.2 Assessment of cell suitability

Cell suitability was assessed using visual characteristics of cardiac myocytes as described in 2.2.2. Cells were viewed in NT solution prior to acute exposure to DOX (0.001 μ M – 10 μ M) followed by a wash out period. Cell experimental suitability was assessed in each condition. Overall experiments lasted 3-10 minutes. Cells that remained striated, rod shaped and non-granulated throughout the experiment were classed as suitable.

Percentage suitability of cells for each animal at each concentration was determined and an average calculated. Statistical significance between treatment groups was established using a one-way ANOVA.

3.2.3 Excitation-contraction coupling measurements

Loaded cells were paced at 0.5 Hz using field stimulation (see section 2.5) in NT. Once Ca and sarcomere length transients reached steady state, the perfusing solution was switched to DOX or DAUN at concentrations of 0.001, 0.01, 0.1, 1 and 10 μ M. Once a new steady state was reached, the reversibility of any effect was determined with a washout (NT). Original recordings were calibrated and analysed offline using custom software (see 2.8).

3.2.4 Measurement of myofilament sensitivity

Alterations to myofilament sensitivity were investigated by plotting phase-plane loops of cytoplasmic Ca versus sarcomere length of each experiment. The phase-plane loops proceed counter-clockwise with cell shortening denoted by the ascending limb (see fig 3.1). The relaxation phase of the cardiac myocyte defines a quasi-equilibrium of systolic Ca, myofilament Ca binding and force, hence cell length (Spurgeon *et al.*, 1992). Therefore, in each plot the cytoplasmic Ca at 50 % relaxation (EC₅₀) was used to quantify the change in myofilament sensitivity and statistical significance was calculated using a paired T-test.



Figure 3.1: Phase plane loops highlighting indications of change in myofilament sensitivity.

Further investigation of myofilament sensitivity was carried out by permeabilising cells using saponin (25 μ g/ml) in a Ca free version of the NT solution outlined in table 2.3. Cardiac myocytes were exposed to saponin for 1 – 5 minutes depending on the visual integrity of the cell. Once permeabilised, resting sarcomere length was measured in the absence of Ca then incremental Ca concentrations (0.35, 3.5 and 35 μ M). This manoeuvre was then repeated in the presence of DOX and DAUN. Mean sarcomere lengths were measured at each Ca concentration and normalised to the length in the absence of Ca. The gradient of the relationship between [Ca] and sarcomere length was compared to determine if DOX or DAUN altered myofilament sensitivity. Statistical analysis was carried out using a 2-way ANOVA and a Holm-Sidak post-hoc test was used to allow comparison between groups.

3.2.5 Measurement of sarcoplasmic reticulum Ca content

Changes in SR Ca were estimated by comparison of Ca transients evoked by rapid application of caffeine (10 mM). This works on the principle that caffeine increases the open probability of the RyR to the extent that mass SR Ca release occurs (Eisner *et al.*, 2017).

3.2.6 Measurement of the activity of the Ca removal mechanisms

As noted in section 1.2.9.1 the main mechanisms of Ca removal in mammalian cardiomyocytes are SERCA and NCX. The rate constant (RC) of decay of systolic Ca transient (k_{sys}) therefore represents the combined activity of SERCA and NCX. Caffeine increases RyR open probability (see 3.2.5) to the extent that SERCA activity is rendered futile, thus the RC of decay of the caffeine-evoked Ca transient (k_{caff}) represents the activity of NCX alone. Subtraction of k_{caff} from k_{sys} provides k_{SERCA} and so the activity of SERCA alone.

3.3 Results

3.3.1 The effects of doxorubicin on cell suitability

To establish a suitable anthracycline experimental concentration, the effects of DOX on cell suitability were assessed. A concentration that produced a graded effect on ECC without reducing cell suitability was required.

Exposure to 10 μ M DOX (fig 3.2), reduced cell suitability to 54.17 % ± 17.58 %. Cell suitability was reduced to 80.56 % ± 10 %, 90 % ± 10 and 96.67 ± 3.33 % by 1, 0.1 and 0.01 μ M DOX respectively. Cell suitability was unaltered during the time course of our experiments at 0.001 μ M DOX. While no statistically significant difference was detected between the suitability of cells from each animal (n = 3, p < 0.066) there is a clear correlation between an increase in concentration and a reduction in cell suitability as shown in figure 3.2.



Figure 3.2: Cell suitability of left ventricular cardiac myocytes isolated from control sheep when treated with 0.001 μ M- 10 μ M doxorubicin and washout (n = 3, p = 0.066).

3.3.2 The dose-dependent effect of doxorubicin on intracellular Ca and contractility

At the concentrations used, no dose dependent effect of DOX was observed (fig 3.3A-F). This was the case for diastolic Ca (n = 13, p = 0.38; fig 3.3A) and diastolic sarcomere length (n = 12, p = 0.599; fig 3.3B). The effect on Ca transient amplitude was not dose dependent (n = 13, p = 0.086; fig 3.3C) and was accompanied by no dose dependent effect on sarcomere shortening (n =10, p = 0.156; fig 3.3D). This was also the case for peak intracellular Ca (n = 13, p = 0.637; fig 3.3E) and rate of sarcomere relaxation (n = 10, p = 0.156; fig 3.3F).



Figure 3.3: The effects of DOX (0.001 – 10 μ M) on intracellular Ca and contractility in sheep ventricular myocytes. (A) mean normalised diastolic Ca. (B) mean normalised diastolic sarcomere length. (E) mean normalised peak sarcomere length. (C) mean normalised Ca

transient amplitude. (D) mean normalised sarcomere length shortening. (E) mean normalised peak Ca. (F) mean normalised 90 % - 10 % decay time of sarcomere length. n = 10 - 21 (from 3 animals).

3.3.3 A detailed investigation of the effects of DOX on Ca and contractility

A working concentration of 1 nM was selected for subsequent detailed investigations given (1) no concentration-dependence of effect was observed in 3.3.2 and (2) this concentration produced no significant effect on cell suitability (3.3.1).

The specimen traces in figure 3.4 show the effect of 1 nM DOX on intracellular Ca and contractility and are representative of the means shown in fig 3.5. On average, no significant effect on diastolic Ca (n = 42, p = 0.213; fig 3.5A) or diastolic sarcomere length (n = 40, p = 0.06; fig 3.5B) was observed. However, DOX reduced the Ca transient amplitude by 39.13 % \pm 3 % (n = 43, p < 0.001; fig 3.5C) which partially reversed upon washout (n = 43, p < 0.001; control vs wash and p = 0.004; drug vs wash). This effect on Ca was accompanied by a 42.46 % \pm 3.79 % (n = 36, p < 0.001) decrease in systolic sarcomere shortening (fig 3.5D) which reversed fully on washout (n = 38, p = 0.159; control vs wash and p < 0.001; drug vs wash). On average, peak Ca decreased by 19.11 % \pm 2.13 % (n = 43, p < 0.001) which partially reversed upon washout (n = 43, p < 0.001; control vs wash and p = 0.011; drug vs wash). The rate of sarcomere relaxation was increased on average by 39.46 \pm 9.21 % (n = 43, p < 0.001) which recovered in washout (n = 36, p = 0.895; control vs wash and p = 0.002; drug vs wash; fig 3.5F).



Figure 3.4: The effect of 1 nM DOX on cytoplasmic Ca and sarcomere shortening. (A) long time base record that shows specimen Ca transients. (B) specimen contractility transients. (C) Ca transient overlays. (D) contractility transient overlays. Long time base records represent steady state from which five Ca transients were averaged to produce the overlays. Data from this single cell is representative of the mean data shown in fig 3.5.



Figure 3.5: The effect of 1 nM DOX on intracellular Ca and contractility in sheep ventricular myocytes. (A) mean normalised diastolic Ca. (B) mean normalised diastolic sarcomere length. (C) mean normalised Ca transient amplitude. (D) mean normalised sarcomere shortening. (E)

mean normalised peak Ca. (F) mean normalised 90-10 % decay time of sarcomere length. n = 36 - 43 (from 9 animals), * denotes statistical significance.

3.3.4 The effect of doxorubicin on SR Ca content and rate of Ca removal

As described in section 3.2 caffeine allows the investigation of alterations to SR Ca content. Specimen shown in figure 3.6A shows the effect of DOX on the amplitude of the caffeine evoked Ca transient. On average this was reduced by 33 % ± 5.16 % (n = 24, p < 0.001) and did not recover in washout (n = 20, p < 0.001; fig 3.7A). Specimen figure 3.6B illustrates the effect of DOX on the Ca transient rate of constant decay (k_{sys}). On average, k_{sys} is reduced by 16.56 % ± 3.02 % (n = 46, p < 0.001) which then recovers in wash (n = 40, p = 0.059; fig 3.7B). The caffeine-evoked Ca transient rate constant of decay (K_{caff}) shown in figure 3.6C increased on average by 64 % ± 25.52 % (n = 20, p = 0.017) and recovered in wash (n = 20, p = 0.107; fig 3.7C). On average, k_{SERCA} was significantly reduced by DOX by 34.07 % ± 8.49 % (n = 20, p = 0.002) and recovered in wash (n = 19, p = 0.588; fig 3.7D).



Figure 3.6: The effect of 1 nM DOX on SR Ca content. (A) specimen Ca transients in the presence of caffeine. (B) specimen normalised Ca transients to permit direct comparison of the rate of Ca decay. (C) specimen normalised caffeine-evoked Ca transients to permit direct comparison of the rate of Ca decay in the presence of caffeine.



Figure 3.7: The effect of 1 nM DOX on Ca removal in sheep ventricular myocytes. (A) mean normalised amplitude of caffeine evoked Ca transient. (B) mean normalised k_{sys} . (C) mean normalised k_{caff} . (D) mean normalised k_{SERCA} . n = 19 - 24 (from 5 animals), * denotes statistical significance.

3.3.5 A detailed investigation of the effects of daunorubicin on Ca and contractility

The specimen traces in figure 3.8 show the effect of 1 nM DAUN on intracellular Ca and contractility. On average, no significant effect was seen on diastolic Ca (n = 20, p = 0.776; fig 3.9A) however the change to diastolic sarcomere length although minor at an increase of

0.32 % ± 0.078 was significant (n = 18, p = 0.003; fig 3.9B). DAUN decreased Ca transient amplitude by 45.31 % ± 4.51 % (n = 20, p < 0.001) which reversed upon washout (n = 20, p = 0.138; fig 3.9C). This was accompanied by a decrease in sarcomere shortening of 64.38 % ± 4.51 % (n = 19, p < 0.001) that recovered upon washout (n = 18, p = 0.313; fig 3.9D). On average, peak Ca was decreased by 18.42 % ± 2.19 % (n = 20, p < 0.001) which partially reversed upon washout (n = 20, p = 0.045; control vs wash and p < 0.001; drug vs wash; fig 3.9E). Sarcomere relaxation was increased on average by 88.5 % ± 26.21 % (n = 18, p < 0.001) which recovered upon washout (n = 16, p = 0.509; fig 3.9F).



Figure 3.8: The effect of 1 nM DAUN on cytoplasmic Ca and sarcomere shortening. (A) long time base record that shows specimen Ca transients. (B) specimen contractility transients. (C) Ca transient overlays. (D) contractility transient overlays. Long time base records represent steady state from which five Ca transients were averaged to produce the overlays. Data from this single cell is representative of the mean data shown in fig 3.9.



Figure 3.9: The effect of 1 nM DAUN on intracellular Ca and contractility in sheep ventricular myocytes. (A) mean normalised diastolic Ca. (B) mean normalised diastolic sarcomere length.

(C) mean normalised Ca transient amplitude. (D) mean normalised sarcomere shortening. (E) mean normalised peak Ca. (F) mean normalised 90-10 % decay time of sarcomere length. n = 16 - 20 (from 6 animals), * denotes statistical significance.

3.3.6 The effect of daunorubicin on SR Ca content

As described in section 3.2 caffeine allows the investigation of alterations to SR Ca content. Specimen shown in figure 3.10A shows the effect of DAUN on the amplitude of the caffeine evoked Ca transient. On average this was reduced by 14 % \pm 6.29 % (n = 23, p = 0.035) and did not recover in washout (n = 20, p < 0.001; fig 3.10B).



Figure 3.10: The effect of 1 nM DAUN on SR Ca content. (A) specimen Ca transients in the presence of caffeine. (B) mean normalised caffeine evoked Ca transient amplitude. n = 20 - 23 (from 5 animals), * denotes statistical significance.

3.3.7 The apparent interaction of daunorubicin and caffeine

In the presence of caffeine, the rate constant of decay was not significantly altered following exposure to DAUN treatment (n = 19, p = 0.06) as shown in figure 3.11. Without caffeine DAUN caused a decrease in k_{sys} of 23.12 % ± 4.51 % (n = 18, p < 0.001) whereas this change was not detected in the k_{sys} after cells had been exposed to caffeine. Due to this apparent interaction of DAUN and caffeine the removal of Ca could not be measured in the presence of DAUN.



Figure 3.11: The effects of 1 nM DAUN on Ca removal in sheep ventricular myocytes. (A) mean normalised rate constant of decay in the presence and absence of caffeine. (B) specimen normalised Ca transients to permit direct comparison of the rate of Ca decay. (C) specimen normalised Ca transients to permit direct comparison of the rate of Ca decay in the presence of caffeine. n = 18 - 19 (from 6 animals), * denotes statistical significance.

3.3.8 The effects of doxorubicin and daunorubicin on myofilament sensitivity Certain factors may reduce myofilament Ca sensitivity such as; acidosis, an increase in the concentration of free magnesium ([Mg]) and β -adrenergic stimulation (Bers, 2001b). Drugs have also been found to impact myofilament Ca sensitivity and alter the force-Ca relationship and so it is important to establish if anthracyclines alter the force-Ca relationship.

Figure 3.12A and C show representative phase-plane loops for cells treated with 1 nM DOX and DAUN. Phase-plane loops proceed counter-clockwise and plot the change of cytoplasmic Ca vs the change of sarcomere length in the absence and presence of DOX and DAUN. Both DOX and DAUN caused the loops to shift leftwards. This shift was quantified by measuring the intracellular Ca concentration required for half relaxation (EC₅₀). On average, DOX reduced EC₅₀ by 6.12 % \pm 1.9 % (n = 20, p = 0.026). DAUN reduced EC₅₀ on average by 8.5 % \pm 2.4 % (n = 20, p < 0.001). Furthermore, due to systolic effects reported in 3.3.2 and 3.3.4 both loops are compressed (fig 3.12 A&C). These observations suggest increased myofilament sensitivity due to anthracyclines.



Figure 3.12: The effect of DOX and DAUN on myofilament sensitivity in sheep left ventricular myocytes. (A) phase-plane loop of sarcomere length versus cytoplasmic Ca when treated with 1nM DOX. (B) EC_{50} (cytoplasmic Ca at 50 % relaxation) of cells in NT compared to 1nM DOX treatment. (C) phase-plane loop of sarcomere length versus cytoplasmic Ca when treated with 1nM DAUN. (D) EC_{50} of cells in NT compared to 1 nM DAUN treatment. n = 20 (from 3 – 6 animals), * denotes statistical significance.

Further experiments were conducted to verify changes to myofilament sensitivity due to anthracycline treatment. As Ca concentration increases the resting sarcomere length reduces as shown in figure 3.13, with a significant shortening at 35 μ M (n = 16, p = 0.036). 1 nM DAUN

resulted a significant reduction in sarcomere shortening at 3.5 and 35 μ M (n = 18, p < 0.001) and was significantly shorter than the control at these concentrations (n = 16, p < 0.001) indicating an increase in myofilament sensitivity. 1 nM DOX did not cause a significant change.



Figure 3.13: The effect of 1 nM DOX and DAUN on myofilament sensitivity. n = 15 (from 5 animals), * denotes statistical significance.

3.4 Discussion

3.4.1 What is an experimentally relevant anthracycline concentration? Using clinically relevant concentrations for reductionist experiments like these is not useful. The concentrations administered clinically will not be the concentration present in the heart or in each cardiac myocyte. These experiments aim to elucidate mechanistic effects on anthracyclines at a cellular level. Therefore, a concentration that does not impact cell experimental suitability while allowing mechanisms to be clarified is required. As shown in figure 3.2 as the concentration of DOX increases (0.001 μ M to 10 μ M), suitability reduces in primary cardiac myocytes. DOX elicits death in human cardiac myocytes within hours of administration (Unverferth *et al.*, 1983) supporting the rapid response seen in our experiments. In cultured human pluripotent stem cell derived cardiomyocytes cell viability was unaltered after a single 3-hour treatment with 150 nM and 300 nM DOX, however a reduction in viability was seen if this treatment was followed by washout (Louisse *et al.*, 2017). Our calcium handling experiments are short term (3 – 10 minutes) and include a wash out period. Maintaining viability and therefore suitability for the length of the experiment is important. A concentration that allows the identification of subtle mechanisms and alterations to Ca handling and contractility while maintaining viability is ideal.

To assist in choosing an appropriate concentration the dose-response of Ca handling and contractility was established. All aspects of calcium dynamics measured (diastolic, peak, transient amplitude, and rate constant of decay) were not altered when cells were treated with the different concentrations of DOX ($0.001 - 10 \mu$ M; as shown in figure 3.3).

From this information we can deduce that since no dose-dependent effect was evident the lowest concentration that did not alter cell experimental suitability is suitable for experimental use (0.001 μ M). While we cannot claim this concentration is clinically appropriate, given the reductionist nature of the experiments we can be confident that this concentration will elucidate mechanistic effects of anthracyclines that will aid in the knowledge of how these drugs impact the heart on a cellular level. Thereby providing valuable insight into the potential implications of these drugs on global Ca handling and contractility.

3.4.2 Does 1 nM doxorubicin alter intracellular Ca handling and contractility in sheep cardiomyocytes?

1 nM DOX did not alter diastolic Ca or sarcomere length in sheep ventricular myocytes (fig 3.4). In rat cardiomyocytes following a 30 minute treatment with 10 μ M DOX an increase in

diastolic Ca was detected (Sag *et al.*, 2011) which was also found to be the case in acute exposure (1 minute) (Mijares & López, 2001). However, this acute study also found no change in diastolic Ca at 1 μ M. In a mouse atrial cardiac myocyte cell line (HL-1) diastolic Ca was not altered by 5 μ M DOX after 30 minutes, however after 60 minutes an increase was evident (Asensio-López *et al.*, 2016). These findings suggest that at higher concentrations of DOX or longer treatment times than those used in our studies an increase diastolic Ca occurs.

DOX resulted in a reduction in Ca amplitude which was accompanied by a decrease in sarcomere shortening, these findings are supported by other experiments conducted on alternative models using various methods. Sag *et al* (2011) saw a reduction in Ca amplitude due to DOX in rat cardiac myocytes which was also found to be the case by Zheng *et al* (2011). Zheng *et al* (2011) measured cell shortening separately and found this was also reduced. It is important to note that these experiments incubated the cardiomyocytes with DOX for a longer period; 30 minutes and 12 hours respectively and they did not investigate recovery after removal of the anthracycline. The time taken for Ca amplitude to reduce in sheep ventricular myocytes was rapid whereas experiments conducted on rats showed variation in the time taken to reduce Ca amplitude with some studies taking up to 3 hours (Timolati *et al.*, 2006). However, once reduced this observation was maintained for the period of exposure (18 hours) along with sarcomere shortening. In our experiments Ca amplitude partially recovered in wash and sarcomere shortening fully recovered. From these data it is evident that DOX reduces Ca amplitude and sarcomere shortening.

Peak Ca was also found to be reduced and partially recover in wash (fig 3.4). While Ca amplitude represents the relative change in Ca it is not an absolute level. A decrease in peak Ca confirms absolute systolic Ca levels – that which activates the myofilaments - are lower. The rate of sarcomere relaxation (90 % - 10 %) was increased in the presence of DOX as shown in figure 3.5F, indicating a decrease in the rate of removal of Ca. This was investigated by quantifying the rate of Ca removal illustrated in figure 3.7B which was found to be reduced. These findings suggest that DOX reduces the rate of removal of Ca which in turn increases the time taken for the sarcomere to relax (Bers, 2001a).

As discussed in 1.2.8.2, a major regulator of systolic Ca Is SR Ca content (Trafford *et al.*, 2000). Therefore, it is logical that SR Ca content may be impacted by DOX. The reduction in the rate of Ca removal and increase in rate of sarcomere relaxation indicates an alteration in the function of the main removal mechanisms, SERCA and NCX (see 1.2.9.4). Therefore, it is important to investigate whether SR Ca content and ion transporter activity are altered due to DOX.

3.4.3 Is SR Ca altered by 1 nM doxorubicin?

In our experiments, DOX decreased SR Ca content (fig 3.6A). The Ca transient amplitude is proportional to the third power of SR Ca (Dibb *et al.*, 2007). DOX reduced SR Ca content to 67 % of control. This would be expected to decrease the Ca transient amplitude to $(0.67)^3 = 30$ %. The Ca transient amplitude only decreased to 61 %, indicating that the decreased SR Ca can account for the change in Ca transient amplitude. However, we do not know if DOX alters Ca buffering and so we cannot quantify the absolute change in SR Ca content (Trafford *et al.*, 1999b). DOX was found to decrease SR Ca content in canine SR vesicle and primary rat cardiac myocytes (Ondrias *et al.*, 1990; Sag *et al.*, 2011), supporting the findings in sheep ventricular myocytes.

3.4.4 Does doxorubicin alter the activity of Ca removal mechanisms?

We next investigated the cause of reduction in SR Ca content. As shown in figure 3.7 DOX increased NCX activity (k_{caff}) and decreased SERCA activity (k_{SERCA}) which can account for the decrease in SR Ca content (Eisner, 2014). While SERCA activity is restored upon washout, NCX recovery is partial. This, and that recent work highlights a relatively small dependence of SR Ca on SERCA activity (Bode *et al.*, 2011) may account for the lack of recovery of SR Ca content during washout. It also supports the suggestion that NCX plays a key role in the regulation of SR Ca content, which is apparently the case here. SERCA and NCX activity are integral to SR regulation but have not been directly quantified.

Long term studies found a downregulation in SERCA expression as well as SR leak leading to a decrease in SR Ca content (Altomare *et al.*, 2021; Timolati *et al.*, 2006). Our findings show the effect of DOX on SERCA and NCX to be rapid (3- 10 minutes) indicating a direct effect of DOX on these channels as protein expression would not be altered. Studies report that DOX activates Ca/calmodulin protein kinase II (CaMKII) (Ikeda *et al.*, 2019b; Tscheschner *et al.*, 2019) which phosphorylates phospholamban and promotes reuptake of SR Ca via SERCA, while also phosphorylating RyR and promoting SR Ca leakage (Yuchi *et al.*, 2012). The roles of CaMKII are vast but its activation would suggest an increase in SERCA activity which is not the case in our study.

With regards to direct interaction with SERCA Hanna *et al* (2014) found SERCA to be inhibited by the metabolite of DOX but not DOX itself. However, DOX was able to bind directly to SERCA. As these findings were not carried out in intact cardiomyocytes it stands to reason that this direct interaction may alter SERCA activity, and result in a decrease in Ca uptake into the SR via SERCA. It is also possible that anthracyclines do not impact SERCA through direct binding but by the production of reactive oxygen species. As mentioned in section 1.5.2.2 anthracyclines have been shown to increase OS. Greensmith *et al* (2010) have shown hydrogen peroxide which is a ROS results in a reduction in SERCA activity which supports this theory.

DOX treatment results in an increase in NCX activity (fig 3.7C). While most studies focus on the impact of DOX on SERCA and RyR few investigate the impact on NCX. The expression of NCX is dependent on CaMKII via stimulation of β -adrenergic receptor (Mani *et al.*, 2010). A study in mice found correlation between increased activity of CaMKII and NCX current in heart failure (Xu *et al.*, 2012). In turn if CaMKII is activated via DOX, this may then result in an increase in NCX current as found by Xu *et al* (2012). This however would only occur after long term exposure. It may be the effect on NCX is due to ROS production as ROS has been shown to increase NCX activity (Goldhaber, 1996; Reeves *et al.*, 1986).
In summary, we are the first to provide mechanistic explanation for decreased SR Ca content by showing that DOX decreased SERCA activity while increasing NCX activity.

3.4.5 Does 1 nM daunorubicin alter intracellular Ca handling and contractility in sheep cardiomyocytes?

There is very little literature regarding the overall impact of daunorubicin on Ca handling and contractility in cardiac myocytes. As mentioned in 3.4.5 there is information regarding the anthracyclines impact on individual channels and the effect on SR Ca, but we are the first to explore the effect of DAUN on global intracellular Ca and contractility.

DAUN did not alter diastolic Ca (fig 3.9A) however the change to diastolic sarcomere length although minor at an increase of 0.32 % was significant. This increase may not be physiologically important as it is < 1 %. A reversible decrease in Ca amplitude was accompanied by a reversible decrease in sarcomere shortening. This is as expected as explained by the force-calcium relationship in section 1.2.9.1. DOX caused an increase in sarcomere relaxation indicating a decrease Ca removal.

3.4.6 Is SR Ca altered by 1 nM daunorubicin?

The caffeine-evoked Ca transient amplitude was decreased by DAUN (fig 3.10). Ca transient amplitude is proportional to the third power of SR Ca content (Dibb *et al.*, 2007). DAUN reduced SR Ca content to 86 % of control. This would be expected to decrease Ca transient to $(0.86)^3 = 64$ %. The Ca transient decreased to 55 % indicating that a decrease in SR Ca content can account for this reduction.

However, it was noted that k_{sys} was altered in the presence of caffeine and DAUN suggesting overall Ca removal is impacted by this interaction therefore impacting the caffeine-evoked Ca amplitude detected. Given the magnitude of the change in the caffeine-evoked Ca transient amplitude it is highly likely that SR Ca content is in fact reduced however this is a relative change. In canine cardiac SR vesicle nanomolar ranges of DAUN inhibited SR Ca release and prevented caffeine-induced SR Ca release (Olson *et al.*, 2000). This suggests DAUN has an inhibitory effect on RyR2. Reduction in the open probability of RyR2 (activity) was identified by treatment with DAUN at 2.5 μ M for 10 minutes (Hanna *et al.*, 2011a). Within the same study a concentration < 2.5 μ M for less than 10 minutes increased the open probability of RyR (activity).

3.4.7 Does daunorubicin alter the activity of Ca removal mechanisms?

It was noted that caffeine exposure in combination with DAUN treatment resulted in no change in the rate constant of as shown in figure 3.11. However, in experiments conducted without caffeine DAUN resulted in a decrease in k_{sys} . Due to this apparent interaction the removal of Ca was unable to be investigated for DAUN.

Interestingly, a study conducted using rabbit skeletal SR vesicles identified that anthraquinones can bind to ryanodine receptors at a different site to caffeine and showed non-competitive interaction (Abramson *et al.*, 1988). This study investigated daunorubicin-ryanodine binding and found binding to be reduced in the presence of caffeine. Caffeine was also shown to inhibit daunorubicin-stimulated SR Ca release but resolves upon washout. Given that our results only found an apparent interaction between DAUN and caffeine and not DOX and caffeine it may be that a different interaction occurs with DOX. More chemical research is required to understand this.

The activity of SERCA needs to be considered when investigating SR Ca content. If the mechanism of action is similar to that of DOX, we would expect a decrease in SERCA activity. This theory is supported by a study in cardiac myocytes isolated from rabbits that had been treated with DAUN for 2 weeks as an inhibition in Ca uptake into the SR was observed (Cusack *et al.,* 1993) indicating a reduction in SERCA activity. While a decrease in SERCA may contribute to a reduction in SR Ca content, the effect of DAUN on NCX is unknown. Given both

SERCA and NCX massively contribute to Ca removal in mammalians both would need to be considered to account for a decrease in SR Ca content.

3.4.8 Do doxorubicin and daunorubicin alter myofilament sensitivity?

By plotting phase-plane loops it was determined that the EC_{50} for cytoplasmic Ca at 50 % relaxation for DOX and DAUN was significantly decreased. This infers that less calcium is required to cause the same change in sarcomere length indicating an increase in myofilament sensitivity highlighted by a shift of the force versus Ca amplitude relationship shown in figure 1.8 to the left.

Further experiments shown in figure 3.11 confirmed that DAUN increased myofilament sensitivity (n = 16, p < 0.001) whereas the results for DOX were inconclusive given no change in sarcomere length at different Ca concentrations was detected. This increase in myofilament sensitivity would (for all things being equal) likely offsetting the effect of the decrease in Ca amplitude on sarcomere shortening. It would be expected that sarcomere shortening would be altered to a greater degree for any given change in Ca. An increase in myofilament sensitivity leads to a rise in force of contraction due to an increase in sensitivity to Ca. This may be due to an increased affinity for troponin C to Ca or increase in the effectiveness of this attachment (Lee & Allen, 1997). For example, the Ca sensitiser levosimendan increases myofilament sensitivity by binding to both the N-terminal and C-terminal of cardiac troponin C, therefore stabilising the complex and increasing the force of contraction (Haikala *et al.*, 1995).

Permeabilised rabbit cardiac myocytes treated with DOX did not alter the Ca concentrationtension relationship (Boucek *et al.*, 1993), which suggests DOX does not alter myofilament sensitivity. Whereas, in male rats exposed to DOX over 6 weeks a decrease in myofilament sensitivity was detected (Chakouri *et al.*, 2020). Interestingly, there may be a gender difference in the effect of DOX on myofilament sensitivity. An increase in myofilament sensitivity was detected in female rats treated with DOX over 10 weeks however no change was found in male rats (Rattanasopa *et al.*, 2019). The same study showed that in the absence of testosterone myofilament sensitivity was increased.

To the best of our knowledge there is no literature identifying the impact of DAUN on myofilament sensitivity.

As discussed in 1.5.2.2 anthracyclines increase ROS. This production of ROS may impact myofilament sensitivity. Hydroxyl radicals lead to a leftward shift in the force-calcium relationship indicating an increase in myofilament sensitivity in a rabbit model (Haizlip *et al.*, 2012). These data along with the data shown in figure 3.13 suggest that DAUN and DOX increase myofilament sensitivity.

3.5 Concluding remarks

In conclusion, this study has elucidated the mechanistic effects of DOX and DAUN on intracellular Ca and contractility in cardiac myocytes. The autoregulation of the cardiomyocyte appears to be altered as a reduction in SR Ca content would normally result in an increase in SERCA activity and reduction in NCX, however the opposite is shown in our findings with DOX. Our findings also highlight differences between DOX and DAUN with DAUN showing a potential caffeine interaction suggesting changes in protein interactions between the two anthracyclines. Furthermore, myofilament sensitivity is increased by both DOX and DAUN.

It is unknown if the impact of DOX and DAUN on Ca handling and contractility is a direct interaction or due to the production of reactive oxygen species (potentially both). This requires further investigation.

Chapter 4

Measurement of anthracycline-induced oxidative stress in sheep ventricular myocytes

4.1 Introduction

As discussed in 1.4 oxidative stress is a state in which the production of ROS outweighs the cells antioxidant mechanisms and is implicated in various cardiovascular diseases (Burton & Jauniaux, 2011). Anthracyclines are known to induce ROS in cancer cells as a mechanism of cytotoxicity as discussed in 1.5.2.2 (Attanasio *et al.*, 2021; Doroshow, 2019; Sheibani *et al.*, 2022). The increase in OS in the heart may contribute to cardiotoxicity.

4.1.1 ROS production in cardiac myocytes

ROS play a role in normal signalling within the heart and are involved in normal excitationcontraction coupling, however they are also implicated in cardiovascular disease (Burgoyne *et al.*, 2012). There are several sources of ROS within cardiomyocytes and examples include; the mitochondrial electron transport chain, NADPH oxidases (NOX) (specifically NOX2 and NOX4) and xanthine oxidoreductase (XO) (Akki *et al.*, 2009).

The mitochondrial electron transport chain consists of four enzymatic complexes and is a main source of physiological ROS (Peoples *et al.*, 2019). This is due to a leakage of electrons (2 -5 %) into the cytoplasm resulting into the formation of superoxides (Kagan *et al.*, 2009). ROS may then leak out of the mitochondria and lead to ROS-induced ROS release from neighbouring mitochondria (Zorov *et al.*, 2006).

Activation of enzymes may induce ROS production. NADPH oxidases are one of the main sources of ROS in the heart (Gray *et al.*, 2019) with NOX2 and NOX4 being the main isoforms expressed (Nabeebaccus *et al.*, 2011). NOX4 is primarily expressed in the mitochondria in cardiac myocytes (Kuroda *et al.*, 2010) whereas NOX2 is localised in the plasma membrane and cytosol (Krijnen *et al.*, 2003). Another enzyme involved in the production of super oxide species and hydrogen peroxide are xanthine oxidases. ROS are produced through a series of reactions in the process of converting xanthine to uric acid (Berry & Hare, 2004a). The

production of ROS via xanthine oxidase is particularly implicated in heart failure (Hajjar & Leopold, 2006).

All these processes contribute to ROS production in cardiac myocytes and this production may be altered in disease states or by drugs.

4.1.2 Anthracyclines and ROS

As discussed in 1.5.2.2 anthracyclines increase ROS in cancer cells and this contributes to their mechanism of cytotoxicity therefore it is plausible that this also occurs in cardiac cells. Anthracyclines such as DOX may lead to semi-quinone production by reduction via xanthine oxidase (Bates & Winterbourn, 1982). This redox cycling of anthracyclines produces various ROS such as hydrogen peroxide (Menna *et al.*, 2010) which may then create an imbalance resulting in oxidative stress in cardiac myocytes. This reaction may also occur due to NOX and by the mitochondrial electron transport chain (Huang *et al.*, 2022). In isolated cardiomyocytes DOX was shown to accumulate in the mitochondria (Ichikawa *et al.*, 2014b) suggesting that NOX4 and/or the mitochondrial transport chain may be involved in the reduction reaction of DOX that produces ROS given the localisation of both the drug and enzyme.

An imbalance in ROS and antioxidant properties of a cell results in oxidative stress and may impact calcium handling within cardiac cells.

4.1.3 Oxidative stress and calcium handling

The modulation of ECC by OS underpins many cardiovascular diseases (Dubois-Deruy *et al.*, 2020; Moris *et al.*, 2017b; Peoples *et al.*, 2019). ROS are known to perturb Ca handling and contractility (D'Oria *et al.*, 2020). This is due a wide range of effects on ECC components.

ROS have been shown to reduce Ca transient amplitude and contractility (Goldhaber & Liu, 1994; Greensmith *et al.*, 2010; Kuster *et al.*, 2010). This decrease in amplitude can be explained by a reduction in SR Ca content (Boraso & Williams, 1994; Greensmith *et al.*, 2010;

Kuster *et al.*, 2010). ROS leads to a decrease in SERCA activity (Greensmith *et al.*, 2010; Kuster *et al.*, 2010; Morris & Sulakhe, 1997; Reeves *et al.*, 1986; Scherer & Deamer, 1986) and an increase in NCX activity (Eigel *et al.*, 2004; Goldhaber, 1996; Kuster *et al.*, 2010; Reeves *et al.*, 1986) which can account for the reduction in SR Ca content. Furthermore, the open probability of RyR has been shown to be increased by ROS (Anzai *et al.*, 1998; Boraso & Williams, 1994; Kawakami & Okabe, 1998) and the voltage gated Ca current reduced (Anzai *et al.*, 1998; Boraso & Williams, 1994; Kawakami & Okabe, 1998). Combined these findings identify the impacts of ROS on ECC components in a wide range of models and provide vital insight into the mechanisms by which Ca amplitude and contractility are reduced.

ROS clearly has implications on calcium handling and contractility. Therefore, it is logical to suggest that the anthracycline-induced ROS may directly lead to Ca handling and contractility dysfunction.

4.1.4 Antioxidants

Cardiac myocytes are highly metabolically active cells and so contain many mitochondria biogenesis (Bartz *et al.,* 2015). As discussed in 4.1.1 the mitochondrial electron transport chain produces superoxides and NOX4 is expressed in the mitochondria and contribute to mitochondrial ROS. Intracellular antioxidant defence mechanisms exist to cope with ROS production such as glutathione peroxidases and catalase as discussed in 1.4.1.2.

The heart has lower levels of antioxidant enzymes compared to other organs (Costa *et al.*, 2013) with catalase activity being particularly lower in the heart (Thayer, 1986). Under normal physiological conditions the coping mechanisms within the heart are adequate to manage ROS production, however in states of pathology such as increased ROS production the antioxidant defences may not be sufficient.

96

The susceptibility of cardiac myocytes to pathological ROS production may account for anthracycline-induced cardiotoxicity if ROS is increased. This increase in ROS is more likely to result in oxidative stress in the heart and result in dysregulation of Ca handling.

The imbalance in ROS to coping mechanisms of a cell may be altered by introducing antioxidants. This is also a useful tool to highlight the role of ROS on cellular dysfunction such as that seen in Ca handling and contractility.

N-acetyl cysteine (NAC) has long been considered a broad antioxidant as it maintains total antioxidant capacity (ROS scavenging) and increases glutathione levels (Mahmoudinezhad *et al.*, 2023). Furthermore, emerging evidence suggests that NAC may work by increasing sulfane sulphur inside cells leading to an increase in the oxidant scavenging capacity of the cell, as well as providing protection against irreversible oxidative damage (Pedre *et al.*, 2021). NAC has been used in various studies as a tool to reduce ROS in primary and cultured cardiac myocytes at a range of concentrations and incubation times (0.1 - 4mM) (Chen *et al.*, 2013; Chen *et al.*, 2012b; Liu *et al.*, 2019). It is a well-established tool and is therefore useful as a generic scavenger of ROS.

4.1.5 Aims

- Validate the novel methodology for measurement of oxidative stress in cardiac myocytes.
- Determine the effect of anthracyclines on levels of oxidative stress in cardiac myocytes.
- Establish to what extent anthracycline-induced oxidative stress are responsible for the alterations in Ca handling and contractility.

4.2 Methodology

4.2.2 Cell preparation

Primary left ventricular cardiac myocytes were isolated from young healthy sheep as described in section 2.1.2 and stored in a solution produced by mixing NT and taurine (see 2.1.1) solution in equal parts to produce an intermediate Ca concentration of 0.95 mM.

4.2.3 Oxidative stress measurements in cardiac myocytes

4.2.3.1 Preparation of cells to measure oxidative stress

As discussed in 1.4.4 existing methods used to measure oxidative stress may be unsuitable for use with cardiac myocytes. For example, the size of the cells and the inability to distinguish live from dead poses an issue. To overcome this a fluorophore compatible with live cells and a live imaging microscope was utilised (see section 2.9 for detail). Primary left ventricular myocytes were exposed to DOX or DAUN (0.001, 0.01, 0.1, 1 and 10 μ M) for 30 minutes before loading with CellROX red (final concentration 5 μ M) for 30 minutes. Cells were then washed and resuspended in NT solution (see table 2.3).

In some experiments, cells were pre-incubated with N-acetyl cysteine (NAC). NAC was prepared by dissolving NAC in sodium hydroxide and pH was readjusted to 7.34 before volumizing using NT. This was then further diluted to 1mM for use on the cells for 30 minutes. To investigate the time-dependent effects (fig 4.3) cells were exposed to DOX (with and without 1 mM NAC pre-treatment) for the period stated prior to loading with 5 μ M CellROX deep red.

4.2.3.2 Measurement of oxidative stress in cardiac myocytes

As discussed in 2.9.2, using the Biotek cytation imaging system brightfield and fluorescent images of cells were captured. The Texas red filter cube was utilised at x 10 magnification.

Analysis was conducted using imageJ software. Brightfield images were overlayed on to the corresponding fluorescence images to identify viable cells. As discussed in 2.5.2 fluorescence was quantified and the background subtracted.

4.2.4 Excitation-contraction coupling measurements.

Cells were loaded with fura-2 (see section 2.6.2) after pre-treatment with 1mM NAC. Field stimulation at 0.5 Hz was used to pace cells as described in section 2.2.4. Cells were exposed to NT and a steady state obtained prior to exposure to 1 nM DOX. Once a new steady state was reached, the reversibility of any effect was determined with a washout (NT). Original recordings were calibrated and analysed offline using custom software (see 2.8).

4.2.5 Sarcoplasmic reticulum Ca content

Comparison of Ca transients evoked by rapid application of 10 mM caffeine allowed changes in SR Ca content to be estimated. Elements of ECC were measured as described in 3.2.5. Statistical significance was determined by conducting a one-way ANOVA unless otherwise stated.

4.3 Results

4.3.1 Validation of method sensitivity

CellROX deep red reagent allows detection of OS in the cell cytoplasm (see 2.9.1). To be useful, the method must be sensitive enough to detect and discriminate pathophysiological levels of oxidative stress. Therefore, cells were exposed to sub-lethal concentrations of hydrogen peroxide. Hydrogen peroxide produced a significant, concentration-dependent increase in fluorescence (thus oxidative stress) reaching 83 % \pm 17 % at 200 μ M (n = 6, p < 0.001) (fig4.1B).



Figure 4.1: ROS detected from sheep ventricular cardiac myocytes treated with various concentrations of hydrogen peroxide (10, 20, 50, 100 and 200 μ M). (A) specimen fluorescent image of cardiac myocytes in control and 200 μ M hydrogen peroxide treatment. (B) mean 100

normalised fluorescence of cardiac myocytes treated with hydrogen peroxide. n = 12-32 cells from 4 animals, * denotes statistical significance.

4.3.2 The effect of DOX and DAUN on cytoplasmic oxidative stress

Initial experiments aimed to determine if anthracyclines increase oxidative stress in a concentration dependent manner. DAUN only produced a significant increase in fluorescence at the highest concentration of 10 μ M; 86 ± 6 % (n = 4, p = 0.005; fig 4.2C). Incubation with DOX for 30 minutes produced a concentration-dependent increase in oxidative stress reaching with a significant increase of 52% ± 11 % (n = 6, p = 0.008; fig 4.2D) and 63.2 % ± 16.1 % (n = 5, p = 0.002; fig 4.2D) at 1 and 10 μ M respectively.



treated with various concentrations of DOX (0, 0.001, 0.01, 0.1, 1 and 10 μ M). (C) OS detected in cardiac myocytes when treated with varying concentrations of DAUN for 30 minutes (n = 29-39 cells from 4 animals). (D) OS detected in cardiac myocytes when treated with varying concentrations of DOX for 30 minutes (n = 41-53 cells from 5 animals), * denotes statistical significance.

4.3.3 The attenuation of DOX-induced OS by N-acetyl cysteine pre-treatment The reduction of OS via 1 mM NAC and the time dependent effect of DOX was investigated as shown in figure 4.3. These experiments sought to (1) provide further validation that the increase in fluorescence due to DOX was indeed due to OS and (2) to ensure that NAC is suitable to reduce DOX-induced OS for future experiments (see 4.3.4 and 4.3.5). No change in ROS was detected when cardiac myocytes were pre-treated with 1 mM NAC at any time point (n = 3, p > 0.05; fig 4.3). 0.1 μ M DOX resulted in a significant increase in OS of 47.42 % ± 12.43 % at 120 minutes (n = 10, p = 0.012) and 49.97 % ± 12.54 % at 240 minutes

(n = 10, p = 0.007). Furthermore, NAC pre-treatment resulted in significantly lower OS by

36.57 % at 240 minutes (n = 3, p = 0.018).



Figure 4.3: The time dependent effect of 0.1 μ M DOX on ROS production in sheep ventricular myocytes with and without 1 mM NAC pre-treatment. Significant results were determined using a two-way ANOVA. n = 15-46 cells from 4-8 animals, * denotes statistical significance.

4.3.4 The effects of DOX on global Ca and contractility following pre-treatment with NAC

The results presented in chapter 3 show DOX alters aspects of ECC in cardiac myocytes. To determine the extent to which those changes are dependent on oxidative stress, the experiments were repeated in the presence of NAC.

The specimen traces in figure 4.4 show the effect of 1 nM DOX on intracellular Ca and contractility in cells pre-treated with 1 mM NAC. On average, no significant effect on diastolic Ca (n = 28, p = 0.181; fig 4.5A) or diastolic sarcomere length (n = 28, p = 0.256; fig 4.5B) was

observed. However, Ca transient amplitude reduced by 18.95 % \pm 2.77 % (n = 29, p < 0.001) and did not recover in wash (n = 27, p < 0.001; fig 4.5C). This effect was accompanied with a decrease in systolic sarcomere shortening by 17.54 % \pm 5.63 % (n = 29, p = 0.007; fig 4.5D) which reversed upon washout (n = 28, p = 0.267). On average, peak Ca decreased by 8.94 \pm 1.63 % (n =29, p < 0.001) which did not recover upon washout (n = 28, p < 0.001; fig 4.5E). On average, no significant effect on sarcomere relaxation was observed (n = 27, p = 0.287).



Figure 4.4: Specimen traces of the effect of 1 mM NAC pre-treatment along with 1 nM DOX treatment on cytoplasmic Ca and contractility in sheep ventricular myocytes. (A) specimen trace of Ca transients. (B) specimen trace of sarcomere length.



Figure 4.5: The effect of 1 mM NAC pre-treatment on Ca handling and contractility in sheep ventricular myocytes when treated with 1 nM DOX. (A) mean normalised diastolic Ca. (B) mean normalised diastolic sarcomere length. (C) mean normalised Ca transient amplitude. 105

(D) mean normalised sarcomere shortening. (E) mean normalised peak Ca. (F) mean normalised 90-10 % decay time of sarcomere length. n = 27 (from 5 animals), * denotes statistical significance.

4.3.5 The dependence of DOX-mediated changes to global Ca and contractility on oxidative stress

Fig 4.6 shows that pre-treatment with 1 mM NAC significantly attenuated the effects of DOX on both Ca transient amplitude by 52 % (n = 29, p < 0.001; fig 4.6A) and sarcomere length shortening by 58 % (n = 29, p < 0.001; fig 4.6B). NAC pre-treatment also significantly reduced the effect on peak Ca by 68 % (n = 29, p < 0.001; fig 4.6C) and rate of sarcomere relaxation by 120 % (n = 28, p < 0.001; fig 4.6D).



Figure 4.6: The percentage change compared to control in cardiac myocytes treated with 1 nM DOX only (n = 36 from 9 animals) and those treated with 1 mM NAC prior to DOX treatment (n = 27 from 5 animals). (A) mean percentage change compared to control of Ca transient amplitude. (B) mean percentage change compared to control of sarcomere length shortening. (C) mean percentage change compared to control of peak Ca. (D) mean percentage change compared to control of peak Ca. (D) mean percentage change compared using a T-test, * denotes statistical significance.

4.3.6 The effects of DOX on SR Ca content and cytoplasmic Ca removal following pre-treatment with NAC

DOX was found to reduce SR Ca content as shown in section 3.3.3. As described in section 3.2 caffeine allows the investigation of changes to SR Ca content. On average, following pretreatment with NAC the amplitude of the caffeine evoked Ca transient was reduced by DOX by 18.34 % ± 3.56 (n = 15, p < 0.001) and did not recover upon washout (n = 15, p < 0.001; fig 4.7A). On average, the Ca transient rate of constant decay (k_{sys}) was unaltered (n = 14, p = 0.119; fig 4.7B). The caffeine-evoked Ca transient rate of decay (k_{caff}) increased by 13.52 % ± 7.87 % (n = 16, p = 0.165) in DOX and significantly increased by 18.86 % ± 7.15 % (n = 15, p = 0.025) in wash (fig 4.7C). The activity of SERCA (k_{SERCA}) was calculated by subtracting k_{caff} from k_{sys} and was not significantly altered (n = 14, p = 0.362; fig 4.7D).



Figure 4.7: The effect of 1 nM DOX on Ca removal in sheep ventricular myocytes were pretreated with 1 mM NAC. (A) mean normalised amplitude of caffeine evoked Ca transient. (B) mean normalised k_{sys} . (C) mean normalised k_{caff} . (D) mean normalised k_{SERCA} . n= 14 - 16 (from 3 animals, * denotes statistical significance.

4.3.7 The dependence of DOX-mediated changes to SR Ca and cytoplasmic Ca removal on oxidative stress

As can be seen in figure 4.8, NAC significantly reduced the effect of DOX on the caffeineevoked Ca transient amplitude by 45 % (n = 15, p = 0.046; fig 4.8A). The effect on k_{sys} and k_{SERCA} was attenuated by 102 % (n = 28, p < 0.001; fig 4.8B) and 96 % (n = 14, p < 0.001; fig

4.8D) respectively. NAC pre-treatment caused a 79 % attenuation on k_{caff} however this was not significant (n = 16, p = 0.117; fig 4.8C).



Figure 4.8: The percentage change compared to control in cardiac myocytes treated with 1 nM DOX (n = 19 from 5 animals) and those pre-treated with 1 mM NAC prior to DOX treatment (n = 14 from 3 animals). (A) mean percentage change of caffeine-evoked Ca transient amplitude. (B) mean percentage change of k_{sys} . (C) mean percentage change of k_{caff} . (D) mean percentage change of k_{SERCA} . Significance was determined using a T-test, * denotes statistical significance.

4.4 Discussion

4.4.1 Is this method sensitive enough to detect changes in oxidative stress? Cells were exposed to hydrogen peroxide (10, 20, 50, 100 and 200 μ M) as a positive control to check the validity of this method. A concentration-dependent increase in fluorescence was detected indicating an increase in OS (fig 4.1). This confirms that this method can detect an increase in OS and is sensitive enough to detect a concentration-dependent effect.

The validity of this method was further confirmed using NAC (see fig 4.3). NAC pre-treatment resulted in a decrease in fluorescence compared to DOX treatment alone confirming that the changes in fluorescence are due to OS.

4.4.2 Does DOX increase levels of oxidative stress in sheep ventricular myocytes? DOX produced a concentration-dependent increase in cytoplasmic oxidative stress after 30 minutes of treatment, with a significant increase evident at 1 and 10 μ M.

Other studies have shown that DOX increases ROS using the fluorophore DCF-DA however have not investigated a concentration dependent-effect (Kim *et al.*, 2006a; Ma *et al.*, 2013; Zhou *et al.*, 2001). While DCF-DA is a commonly used fluorophore it has lower photostability to CellROX reagents which may impact reproducibility (Souza *et al.*, 2020). Furthermore, CellROX reagents are selective to OS as evidenced by the inhibitor experiment (fig 4.3) and shown by the manufacturer (Invitrogen- Molecular probes). Invitrogen specify that CellROX reagents are used for generalised OS measurements, whereas DCF-DA detects hydrogen peroxide, with some versions of the fluorophore suitable for hydroxyl and peroxyl radical detection, however it does not detect singlet oxygen or superoxide anions. While other studies show an increase in ROS in cardiac myocytes due to DOX the use of DCF-DA means not all ROS production is considered. Our experiments support the findings that DOX increases ROS and shows DOX increases OS in a concentration-dependent manner (fig 4.2).

Furthermore, our studies showed a time dependent effect in OS with an increase of 24 % being evident after 10 minutes (fig 4.3). This indicates a rapid response in DOX-induced OS which is supported by a study that found an increase in ROS production due to DOX close to the mitochondria after only 20 minutes of exposure (Sarvazyan, 1996). This suggests mitochondrial ROS are a large contributor and may then diffuse into the cytoplasm (Munro & Pamenter, 2019). As discussed in 4.1.4 cardiac myocytes are densely packed with mitochondria (Barth *et al.*, 1992) and under normal physiological conditions the cells are able to cope with ROS production. However, our findings show that DOX increases the levels of ROS beyond the coping mechanisms of the cell resulting in OS.

4.4.3 Does DAUN increase levels of oxidative stress in sheep ventricular myocytes?

In this study DAUN increased OS by 86.36 \pm 6.3 % after 30 minutes at 10 μ M (n = 4, p = 0.005), however did not result in a concentration-dependent increase. The lower concentrations of DAUN (0.001, 0.01, 0.1 and 1 μ M) did not increase OS. This is supported by a study that found 1 μ M DAUN did not increase ROS in cardiac myocytes after 3 hours of treatment (Sawyer *et al.*, 1999). Furthermore, in cancer cell lines 10 μ M DAUN increases ROS after an incubation of 4 hours and remained high post-recovery (Al-Aamri *et al.*, 2019). These findings along with our own suggest that increased ROS production and ultimately OS requires high concentrations of DAUN (10 μ M). The effects of DAUN on ROS production in cardiac cells is less studied than DOX due to the assumption that the compound will produce similar effects, however our findings contradict that. As shown in figure 1.10 DAUN has a quinone structure similar to DOX (Alves *et al.*, 2017) meaning it has the capability of being reduced to a semiquinone as discussed in 4.1.2 and producing ROS as a biproduct. The biochemical

properties are beyond the scope of this study but given the difference in OS production between DAUN and DOX may be an area of interest for further investigation.

4.4.4 To what extent are the effects of DOX on Ca handling and contractility dependent on oxidative stress?

DOX increases oxidative stress as shown in figures 4.2D and 4.3 however to what extent this causes alterations in Ca handling and contractility in cardiac myocytes remains unknown. The data discussed in 4.4.1 confirms NAC reduced OS in cardiac myocytes and is supported by others who show a reduction in ROS (Kuznetsov *et al.*, 2011), therefore confirming that NAC is a suitable tool to ascertain the extent to which DOX-induced OS alters Ca handling and contractility.

When pre-treated with NAC, DOX did not alter diastolic Ca but lead to a significant reduction in Ca transient amplitude, systolic shortening, and peak Ca (fig 4.5C). Despite a reduction in these Ca handling and contractility parameters, the NAC pre-treatment still attenuated the effect of DOX on Ca amplitude, systolic shortening, peak Ca and rate of sarcomere length shortening by 52, 58, 68 and 120 % respectively (fig 4.6).

These results suggest that OS is responsible for more than half of the decrease in Ca transient amplitude and systolic shortening due to DOX treatment. The Ca handling and contractility experiments are conducted over 3 -10 minutes meaning that any impact due to an increase in ROS production must be rapid. An increase of $24 \pm 9\%$ in cytoplasmic OS was detected after 10 minutes of incubation with 0.1 μ M DOX, which indicates a quick response in ROS production with OS increasing over the 4-hour period (fig 4.3). This is supported by a study that found under stress ROS production increases within 5 minutes (Kuznetsov *et al.*, 2011) and supports that alterations to Ca handling and contractility as a result of DOX treatment may be due to increased ROS production leading to OS. NAC attenuated the effect of DOX on Ca amplitude and systolic shortening by 52 % and 58 % (fig 4.6) which indicates OS plays at least half a role in this alteration. As discussed in 4.1.3 studies have shown ROS reduces Ca amplitude and contractility (Goldhaber & Liu, 1994; Greensmith *et al.*, 2010; Kuster *et al.*, 2010). Our findings indicate that DOX-induced OS can account for at least half the reduction in Ca amplitude and contractility identified in chapter 3. Interestingly sarcomere length relaxation was fully attenuated by NAC pre-treatment indicating that DOX-induced OS is a major contributor to this alteration and is potentially related to altering the rate of Ca removal.

4.4.5 To what extent are the effects of DOX on SR Ca dependent on oxidative stress?

The caffeine-evoked Ca transient amplitude was reduced by DOX after NAC pre-treatment indicating a decrease in SR Ca content. The Ca transient amplitude is proportional to the third power of SR Ca (Dibb *et al.*, 2007). NAC-pre-treatment resulted in DOX reducing SR Ca content to 82 % of control. This would be expected to decrease the Ca transient amplitude to $(0.82)^3$ = 55 %. The Ca transient amplitude only decreased to 77 % indicating that the decreased SR Ca can account for the change in Ca transient amplitude.

When compared to DOX treatment alone NAC attenuated the effects to the SR Ca content by 45 % (fig 4.8A). ROS has been shown to reduce SR Ca content (Boraso & Williams, 1994; Greensmith *et al.*, 2010; Kuster *et al.*, 2010) and this along with our findings suggests DOX-induced OS is responsible for more than a third of the reduction in SR Ca content.

4.4.6 To what extent are the effects of DOX on Ca removal mechanisms dependent on oxidative stress?

NAC pre-treatment fully attenuated the effect of DOX on K_{sys} and K_{SERCA} . The effects on NCX were partially attenuated. This suggests that OS is the direct cause of a decrease in SERCA activity due to DOX which is supported by various studies that found ROS decreased SERCA

activity (Greensmith *et al.*, 2010; Kuster *et al.*, 2010; Morris & Sulakhe, 1997; Reeves *et al.*, 1986; Scherer & Deamer, 1986). ROS has been shown to increase NCX activity (Eigel *et al.*, 2004; Goldhaber, 1996; Kuster *et al.*, 2010; Reeves *et al.*, 1986) and the partial attenuation of 79 % shown in this study (fig 4.8C) indicates that OS has a substantial role in the alteration to NCX as a result of DOX treatment.

While the increase in NCX activity shown in figure 4.7 is not significant it may still lead to a reduction in SR Ca content. However, as discussed in 3.1.1 DOX may alter L-type Ca channel activity as well as the open probability of RyR which can impact SR Ca content and these alterations may not be dependent on DOX-induced OS. This may explain the decrease in SR Ca content identified in this study.

4.5 Concluding remarks

In conclusion, this study has highlighted the role of DOX-induced OS on Ca and contractility in ventricular myocytes. DOX resulted in an increase in intracellular OS in cardiac myocytes which may originate both from the cytoplasm and mitochondria. The oxidative stress response in cardiac myocytes due to DAUN and DOX differed. DAUN increased OS at high concentrations, whereas DOX resulted in a rapid response and showed a concentration-dependent increase. DAUN has been associated with less cardiotoxicity than DOX (Feijen *et al.,* 2015a) which based on this study may be due to higher concentrations being required to increase OS.

NAC was utilised to lower OS and resulted in attenuation on various Ca handling and contractility parameters that DOX impacted. Approximately half of the effect on Ca amplitude, systolic shortening and SR Ca content can be attributed to DOX-induced OS. Interestingly, the effect of DOX on the rate Ca removal was fully attenuated by the reduction of OS, suggesting that ROS directly impacts SERCA and NCX. The pathway by which DOX increases OS is still unknown.

Chapter 5

The role of xanthine oxidase and NADPH oxidase in anthracycline-induced oxidative stress

5.1 Introduction

5.1.1 The mechanisms of increased oxidative stress by anthracyclines

Anthracyclines have been shown in this study to increase ROS in cardiac myocytes. In chapter 4, removal of OS with NAC - a generic ROS scavenger - demonstrated that OS plays a role in the alterations in Ca handling and contractility caused by DOX. However, the sources -namely ROS production - of this OS remain unknown. As discussed in 4.1.2 anthracyclines may undergo redox reactions by various pathways in cardiac myocytes forming a semi-quinone leading to ROS production (Menna *et al.*, 2010).

Several studies on both primary and culture cardiac myocytes show DOX can increase intracellular ROS production (Kim *et al.*, 2006b; Ma *et al.*, 2013; Zhou *et al.*, 2001). Enzymatic pathways for ROS production such as xanthine oxidase and NADPH oxidase have been thought to play a role in anthracycline-induced ROS (Bates & Winterbourn, 1982; Huang *et al.*, 2022).

5.1.2 ROS inhibition

As discussed in 1.4 ROS may be produced through various pathways including via NADPH oxidases and xanthine oxidases with physiological removal pathways of ROS including catalase and glutathione peroxidase.

Experimentally and clinically, various inhibitors exist to reduce the production of ROS. As previously mentioned in 1.4.1.1, both NADPH oxidase and xanthine oxidase may be sources of ROS in cardiac myocytes (Akki *et al.*, 2009). Inhibitors of these enzymes can be used to establish the impact of different pathways on ROS production and whether this ROS production is leading to other cellular alterations. Inhibition of ROS production provides a preventative strategy rather than a reactionary approach to reducing ROS. While this inhibition may not be physiological it is (1) pharmacologically useful and (2) a useful tool to determine the sources of DOX-induced ROS shown in chapter 4.

5.1.3 Xanthine oxidase inhibition

Xanthine oxidoreductase is an enzyme that produces ROS as a biproduct of uric acid production from hypoxanthine metabolism (Kang & Ha, 2014) as shown in fig 5.1. It exists in two interconvertible forms in mammals; xanthine dehydrogenase and xanthine oxidase (XO), with the latter having an important role when cells are under stress (Chen & Meng, 2022) This enzyme is thought to have a role in the physiology of congestive heart failure (Berry & Hare, 2004b). Inhibition of xanthine oxidase in heart failure models increases contractility and improves left ventricular function (Ekelund *et al.*, 1999; Ukai *et al.*, 2001), highlighting the key role of this enzyme in cardiac dysfunction.





Xanthine oxidase inhibitors are used therapeutically for the reduction of serum urate levels to treat gout, with febuxostat (FEB) approved for use medically in 2009 (White, 2018). The cardiovascular safety of febuxostat has been investigated and while one trial found patients treated with FEB had higher cardiovascular mortality (White *et al.*, 2018), a more recent trial found that there was not an increased risk of serious adverse events or cardiovascular death (Mackenzie *et al.*, 2020).

5.1.4 NADPH oxidase inhibition

NADPH oxidase was first identified as having a role in immune defence by producing ROS (Kuroda & Sadoshima, 2010). Within the NADPH oxidase group, two isoforms (NOX2 and NOX4) are present in cardiac myocytes and are associated with cardiac dysfunction (Szekeres *et al.*, 2021a). This cardiac dysfunction may be due to pathological activation especially considering the relatively low antioxidant capacity of the heart (Costa *et al.*, 2013).

VAS3947 (VAS) is a specific inhibitor of NOX2 and NOX4 (Reis *et al.*, 2020) making it an ideal compound to investigate the production of ROS via NADPH oxidases in cardiac myocytes.

5.1.5 Summary

Chapter 3 showed that DOX and DAUN perturb Ca handling and contractility and chapter 4 highlighted the contributing role of oxidative stress. As reviewed above, inhibition of XO and NOX - the main enzymatic sources of ROS in the heart – is pharmacologically useful. Here, we use those inhibitors to determine the extent to which XO and NOX contribute to DOX-mediated oxidative stress.

5.1.6 Aims

- Determine if xanthine oxidase and NOX2 and NOX4 contribute to DOX-induced oxidative stress.
- Investigate if the effects of DOX on Ca handling and contractility can be attenuated by XO and/or NOX inhibition.
- Investigate whether the effects of DOX on myofilament sensitivity can be attenuated by XO and/or NOX inhibition.

5.2 Methods

5.2.1 Cell preparation and excitation-contraction coupling measurements

Primary left ventricular cardiac myocytes were isolated from 18-month-old sheep as described in section 2.2.2. After pre-treatment with inhibitors described in 5.2.4, they were loaded with fura-2 (see section 2.6.2) and stored in 0.95 mM Ca.

Cells were paced at 0.5 Hz using field stimulation as described in section 2.5. NT solution (see table 2.3) was perfused on to cells and allowed to reach steady state before being switched to 1 nM DOX. Following DOX treatment, a wash period was allowed to establish if the effects of the drug were reversible. The DOX treatment data shown in figure 3.4 are used as a comparison in figure 5.2 to highlight the impact of the antioxidants used.

5.2.2 Reactive oxygen species inhibitor preparation and treatment

Cells were pre-treated with one of two pathway specific antioxidants.

Xanthine oxidase inhibition was evident at 25 nM of FEB in vascular endothelial cells (Malik *et al.*, 2011), while in macrophages a concentration of 30 μ M was shown to cause inhibition (Nomura *et al.*, 2013). A study in rat cardiomyocytes found 10 μ M FEB for 1 hour treatment to be effective (Wang *et al.*, 2015) and therefore this concentration and treatment time were used in this study.

Studies have demonstrated the reduction in ROS due to NOX inhibition due to VAS with IC50 values ranging from 2 μ M – 12 μ M in cancer cell lines (El Dor *et al.*, 2020; Wind *et al.*, 2010b). A reduction in ROS in spontaneously hypertensive rat aortas was detected when treated with 10 μ M VAS for 30 minutes (Wind *et al.*, 2010b). Furthermore, using purified human NOX4 Reis *et al* (2020) found an IC50 of 31.9 μ M for VAS and almost 50 % inhibition after 10 minutes

with 100 % inhibition evident after 1 hour of treatment, while NOX2 inhibition was 50 % after 1 hour. Given this data and initial assessment of suitability a concentration of 30 μ M for 1 hour was utilised.

Both FEB and VAS3947 were prepared to a stock solution of 50 mM in DMSO. FEB dilutions at 10 μ M and VAS dilutions at 30 μ M were prepared in NT solution. Cells were incubated with the NADPH oxidase inhibitor VAS3947 at 30 μ M for 30 minutes prior to loading with fura-2. A separate group of cells were treated with febuxostat; a xanthine oxidase inhibitor at 10 μ M for 1 hour before fura-2 loading.

These concentrations and incubation durations have been shown to reduce ROS as shown in figure 5.1 and have been used in other studies (Altenhöfer *et al.*, 2014; Nomura *et al.*, 2013; Wang *et al.*, 2015; Wind *et al.*, 2010a).

5.2.3 Measurement of sarcoplasmic reticulum Ca content and the activity of Ca removal mechanisms

Changes in SR Ca content were investigated by rapid caffeine application (10mM) as described in section 3.2.5. The rate constant of decay of systolic Ca transient (k_{sys}) represents the activity of NCX and SERCA. In the presence of caffeine SERCA activity is rendered futile due to the increase in open probability of RyR (Eisner *et al.*, 2017). Therefore, the RC of decay of the caffeine-evoked Ca transient (k_{caff}) represents the activity of NCX alone. By subtraction of k_{caff} from k_{sys} the relative activity of SERCA can be determined.

5.2.4 Measurement of myofilament sensitivity

Myofilament sensitivity was investigated by plotting phase-plane loops of cytoplasmic Ca versus sarcomere length of each cardiomyocyte and the EC_{50} (cytoplasmic Ca at 50 % relaxation) was used to quantify any change in myofilament sensitivity as described in detail in 3.2.4. Statistical significance was identified as p < 0.05 and calculated using a paired T-test.

5.3 Results

5.3.1 The effect of inhibition of XO and NOX2 and NOX4 on DOX-induced oxidative stress

Pre-treatment with 10 μ M FEB reduced levels of 1nM DOX-mediated OS on average by 22 % and 29 % after treatment with DOX for 10 and 240 minutes. While this change in OS was not statistically significant (n = 12-20 cells from 3 animals, p > 0.05), it indicates that XO has a role (at least partially) in DOX-induced OS. No increase in OS was detected in cells pre-treated with FEB following DOX incubation (n =12-20 cells from 3 animals, p > 0.05; fig 5.2). Pre-treatment with 30 μ M VAS3947 (NOX2 and NOX4 inhibitor) for 30 minutes reduced levels of oxidative stress by 65 %, 54 %, 70 % and 50 % when compared to DOX treatment alone at 10, 60, 120 and 240 minutes (n = 10-16 cells from 3 animals, p < 0.05; fig 5.2) and no change in OS over the 240 minutes was detected (n = 10-16 cells from 3 animals, p > 0.05; fig 5.2).



Figure 5.2: The time dependent effect of 0.1 μ M DOX on OS production in sheep ventricular myocytes with and without XO and NOX inhibition. Statistically significant results were determined using a two-way ANOVA. n = 10-46 cells from 3–8 animals, * denotes statistical significance.

5.3.2 The effect of xanthine oxidase inhibition on the alterations to Ca handling and contractility produced by DOX

The specimen traces in figure 5.3 show the effect of 1 nM DOX on intracellular Ca and contractility in cardiac myocytes after pre-treatment with 10 μ M FEB. On average, no significant effect on diastolic Ca (n = 19, p =0.082; fig 5.4A) or diastolic sarcomere length (n = 15, p = 0.605; fig 5.4B) was observed by DOX after FEB treatment. DOX did not significantly reduce Ca transient amplitude (n = 18, p = 0.527) after FEB pre-treatment however this was significantly reduced in wash by 18.8 ± 4.92 % (n = 19, p = 0.015; fig 5.4C). On average, FEB pre-treatment resulted in DOX reducing sarcomere length shortening by 16.47 ± 6.05 % however this was not significant (n = 14, p = 0.252; fig 5.4D). The rate constant of decay (k_{sys}) was not altered by DOX in the presence of FEB (n = 16, p = 0.414; fig 5.4E). On average, DOX did not alter sarcomere relaxation after FEB treatment (n = 14, p = 0.261) however a significant decrease of 14.53 ± 9.16 % occurred in wash (n =14, p = 0.01; fig 5.4E).



Figure 5.3: The effect of xanthine oxidase inhibition using 10 μ M FEB and 1 nM DOX on cytoplasmic Ca and contractility in sheep ventricular myocytes. (A) specimen trace of Ca transients. (B) specimen trace of sarcomere length.


Figure 5.4: The effect of xanthine oxidase inhibition using 10 μ M FEB and 1 nM DOX on intracellular Ca and contractility in sheep ventricular myocytes. (A) mean normalised diastolic Ca. (B) mean normalised diastolic sarcomere length. (C) mean normalised Ca

transient amplitude. (D) mean normalised sarcomere length shortening. (E) mean normalised Ca transient rate constant of decay (k_{sys}). (F) mean normalised 90 – 10 % decay time of sarcomere length. n = 14 – 19 (from 6 animals), * denotes statistical significance.

5.3.3 The effect of NOX inhibition on the alterations to Ca handling and contractility produced by DOX

The specimen traces in figure 5.5 show the effect of 1 nM DOX on intracellular Ca and contractility in cardiac myocytes after pre-treatment with 30 μ M VAS. On average, no significant effect on diastolic Ca (n = 18, p = 0.216; fig 5.6A) or sarcomere length (n = 18, p = 0.916) (fig 5.6B) was observed. DOX decreased Ca transient amplitude by 21.58 % ± 4.24 % (n = 17, p < 0.001) and did not recover upon wash out after VAS pre-treatment (n = 17, p < 0.001; fig 5.6C). This was accompanied by a 15.37 ± 6.79 % decrease in sarcomere length shortening (fig 5.6D), however this decrease was not significant (n = 17, p = 0.063). On average, DOX did not alter the rate constant of decay (k_{sys}) following VAS pre-treatment (fig 5.6E; n = 18, p = 0.916). DOX did not alter sarcomere relaxation (n = 16, p = 0.636; 5.6F) after VAS treatment.



Figure 5.5: Specimen traces of the effect of NOX2 and NOX4 inhibition using 30 μM VAS and DOX on cytoplasmic Ca and contractility in sheep ventricular myocyte. (A) specimen trace of Ca transients. (B) specimen trace of sarcomere length.



Figure 5.6: The effect of NOX2 and NOX4 inhibition using 30 µM VAS and 1 nM DOX on intracellular Ca and contractility in sheep ventricular myocytes. (A) mean normalised diastolic Ca. (B) mean normalised diastolic sarcomere length. (C) mean normalised Ca

transient amplitude. (D) mean normalised sarcomere length shortening. (E) mean normalised Ca transient rate constant of decay (k_{sys}). (F) mean normalised 90 – 10 % decay time of sarcomere length. n = 15 - 18 (from 5 animals), * denotes statistical significance.

5.3.4 The impact of NOX an XO inhibition on DOX-induced alterations to Ca handling and contractility

Though DOX still resulted in a decrease in Ca transient amplitude when cells were pre-treated with xanthine oxidase (FEB) and NOX (VAS) inhibitors (fig 5.4C and fig 5.6C) these effects were attenuated by 79 % and 45 % respectively (n = 18, p < 0.001 and n = 17, p = 0.015; fig 5.7A). The reduction in sarcomere length shortening produced by DOX was attenuated by 61 % and 64 % by FEB and VAS pre-treatment respectively (fig 5.7B; n = 14, p = 0.004 and n = 17, p = 0.001). The removal of Ca by all removal mechanisms was shown to be reduced by 1 nM DOX (fig 3.7A), this effect was attenuated by VAS 103 % (n = 18, p = 0.003). While FEB attenuated the impact of DOX on k_{sys} by 51 % this change was not significant (n = 16, p = 0.244). 1 nM DOX increased the rate of sarcomere relaxation (fig 3.5F), however pre-treatment with FEB and VAS fully attenuated this effect by 120 % and 115 % respectively (n = 14, p = 0.008 and n = 16, p = 0.007).



Figure 5.7: The percentage change compared to control in cardiac myocytes treated with 1 nM DOX only (n = 36 from 9 animals) and those treated with FEB (n = 14 from 6 animals) or VAS3947 (n = 15 from 5 animals) prior to DOX treatment. (A) mean percentage change compared to control of Ca transient amplitude. (B) mean percentage change compared to control of sarcomere length shortening. (C) mean percentage change compared to control of rate of sarcomere relaxation, * denotes statistical significance.

5.3.5 The effect of NOX and XOR inhibition on DOX-mediated alterations to myofilament sensitivity

Phase-plane loops proceed counter-clockwise with sarcomere relaxation depicted by the lefthand side of the loop. The EC₅₀ represent the cytoplasmic Ca at 50 % sarcomere relaxation. A shift of the loop highlights a change in myofilament sensitivity, for example a shift to the left indicates an increase in myofilament sensitivity. Chapter 3 shows DOX shifts phase plane loop. When pre-treated with FEB or VAS, this shift was prevented (fig 5.8A & C). To quantify this the EC₅₀ was calculated. There was no significant difference between the EC₅₀ of cells in control compared to 1 nM DOX treatment when pre-treated with either FEB or VAS (n = 15, p = 0.105 and n = 15, p = 0.43, respectively) indicating no alteration in myofilament sensitivity.

For comparison, data was normalised to each cell's control EC_{50} and the average EC_{50} is displayed in figure 5.9. The EC_{50} of each individual cell treated with DOX only is illustrated in figure 3.10, presented in figure 5.9 is the normalised average of this data. Compared to control the EC_{50} of DOX only was statistically significantly lower indicating an increase in myofilament sensitivity as explained in section 3.3.8. FEB and VAS pre-treatment resulted in no change in myofilament sensitivity during DOX treatment.



Figure 5.8: The effect of 1 nM DOX on myofilament sensitivity in sheep ventricular cardiac myocytes pre-treated with antioxidants. (A) phase-plane loop of sarcomere length versus cytoplasmic Ca in a cell pre-treated with 10 μ M FEB (xanthine oxidase inhibitor) for 1 hour before and after DOX exposure. (B) EC₅₀ (cytoplasmic Ca at 50 % sarcomere relaxation) of cells pre-treated with FEB in NT compared to 1 nM DOX. (C) phase-plane loop of sarcomere length versus cytoplasmic Ca in a cell pre-treated with 30 μ M VAS (NOX inhibitor) for 30 minutes before and after DOX exposure. (D) EC₅₀ (cytoplasmic Ca at 50 % sarcomere relaxation) of cells pre-treated with VAS in NT compared to 1 nM DOX treatment. Significant results (p < 0.05)

were determined by a paired T-test. n = 15 (from 5-6 animals), * denotes statistical significance.



Figure 5.9: Mean normalised cytoplasmic Ca at 50 % sarcomere length relaxation (EC₅₀). n = 15-27 (from 4-6 animals), * denotes statistical significance.

5.4 Discussion

5.4.1 Does inhibition of XO and NOX2 and NOX4 reduce anthracycline-induced ROS?

Inhibition of XO and NOX2 and 4 reduces the levels of oxidative stress produced by 1 nM DOX (fig 5.2). Inhibition of XO resulted in a 22 % – 31 % reduction in DOX-induced OS on average over the stated incubation times (fig 5.2). While this is not statistically significant, given the magnitude of change we cannot exclude the possibility that XO contributes to the increase in OS caused by DOX. In mice treated with DOX for 7 days cardiac XO activity was found to be increased (Tanaka *et al.*, 2021). However, the cardiac XO activity in mice has been calculated to be greater than that of large mammals such as pigs and humans (Jong *et al.*, 1990). This

highlights the need for large animal model studies as it suggests that while DOX may increase XO activity, this increase would be to a lesser degree in humans than small rodents. It is likely that XO has a role in DOX-induced OS, however the extent of that role appears to be less than that of NOX from our findings.

Inhibition of NOX significantly reduced levels of oxidative stress by 50 – 70 % on average across the time points measured (fig 5.2). This suggests that at least half of the OS induced by DOX is a result of NOX activation. This is likely the combined activity of both NOX2 and NOX4. NOX4 is constitutively active whereas NOX2 requires regulatory subunits for activation (Nisimoto et al., 2010). NOX2 activation involves phosphorylation of the adaptor protein p47phox which then allows interaction between NOX2 and p67phox which leads to ROS production (Brandes et al., 2014). Given the significant increase in ROS production by DOX (~ 50 %) it may be that DOX promotes the phosphorylation or interaction leading to NOX2 activation. This is supported by a study conducted on endothelial progenitor cells that found DOX to activate NOX2 (De Falco et al., 2016). NOX4 is localised to the inner mitochondrial membrane and is a major source of mitochondrial oxidative stress (Kuroda et al., 2010; Peoples *et al.*, 2019). Due to the abundancy of mitochondria in cardiac myocytes it is logical that NOX4 contributes to a large proportion of cardiac myocyte ROS. This is evidenced by the large decrease in OS induced by DOX when NOX is inhibited (fig 5.2). While it is possible DOX may increase the activity of NOX2 and NOX4, it is also plausible that NOX2 and NOX4 contribute to reducing DOX to a semiquinone and releasing ROS in the process (Yousefian et al., 2021). Due to NOX4 being continuously active the latter explanation is likely the most plausible to explain the role of this enzyme in DOX-induced OS.

These data suggest that NOX-mediated ROS production is the dominant contributor to DOXinduced oxidative stress. XO may also play an important role but requires further investigation.

5.4.2 Does xanthine oxidase inhibition alter the effect of DOX on Ca handling and contractility?

As shown in figure 5.4 when pre-treated with FEB, DOX did not significantly alter any parameters of Ca handling, however in wash Ca amplitude and rate of sarcomere relaxation were significantly decreased. This suggests that at the least, the immediate effects of DOX are heavily dependent on XO. As mentioned, the relationship between ROS and Ca is bidirectional. ROS have been shown to alter the activity of Ca channels and exchangers, for example activation of voltage dependent channels such as L-type Ca channels may be facilitated by ROS (Görlach *et al.*, 2015). Therefore, the gradual reduction in Ca amplitude and sarcomere relaxation may not be a direct implication of DOX treatment but rather an effect of lack of ROS production. The lack of ROS production may result in a decrease in L type Ca channel activation resulting in a decrease in Ca amplitude (Tabet *et al.*, 2004). As shown in figure 5.2 OS is reduced by FEB by ~ 20 %, though not significant this still highlights a decrease in ROS which would support this theory.

When compared to DOX treatment alone the effect on Ca transient amplitude was attenuated by 79 % and the effect on sarcomere length shortening was attenuated by 61 % as shown in figure 5.7. This indicates that ROS production via xanthine oxidase contributes to more than half of the effect seen to Ca amplitude and therefore sarcomere length shortening. This is surprising given that the reduction in DOX-induced OS after XO inhibition was 22 % following a 10-minute treatment. The attenuation effects are large given the relatively modest reduction in OS. However, when pre-treated with FEB, the removal of Ca by NCX and SERCA was attenuated by 51 % and the effect of DOX on sarcomere relaxation was fully prevented. This may indicate SR Ca is not altered. Given the steep relationship between SR Ca and Ca amplitude (Dibb *et al.*, 2004) this may explain the attenuation effect on Ca amplitude. The reduction in OS though not statistically significant may allow activity of Ca channels and exchangers to be restored while not being reduced to the extent that they are negatively affected by the lack of ROS (Görlach *et al.*, 2015). These studies in combination highlight that XO may contribute (at least partially) to DOXinduced OS, and that even a \sim 20 % reduction in OS may restore the alterations in Ca handling and contractility induced by DOX.

5.4.3 Does NOX2 and NOX4 inhibition alter the effect of DOX on Ca handling and contractility?

Cells pre-treated with VAS did not alter Ca handling or contractility in ventricular myocytes except for an irreversible decrease in Ca transient amplitude as shown in figure 5.6. The effect of DOX on Ca transient amplitude was attenuated by 45 % in the presence of VAS with sarcomere length shortening attenuated by 64 % (fig 5.7). Inhibition of NOX (via VAS) resulted in a 65 % reduction in DOX-induced OS at 10 minutes (fig 5.2). This suggests DOX-induced OS through NOX accounts for at least half of the alterations identified in Ca amplitude, this is supported by a study that found NOX inhibition restored contractility and improved Ca handling in an aged-rat model (Valdés et al., 2018). ROS production via NOX has been shown to alter Ca handling and contractility (Szekeres et al., 2021b) and our study supports that the alterations in Ca handling and contractility due to DOX are largely attributable to ROS production facilitated by NOX. Given that k_{sys} (removal of Ca by all mechanisms) was fully attenuated as was the rate of sarcomere relaxation in the presence of the NOX inhibitor this suggests the role of NOX in DOX-induced OS is one of the main mechanisms by which Ca removal is altered. As discussed in chapter 4 ROS may alter SERCA and NCX, NOX may contribute to ROS production either through an increase in activity due to DOX and/or by facilitating the reduction reaction of DOX leading to ROS formation.

Our data show that DOX-induced ROS is largely attributed to NOX and has a vital role in altering Ca handling and contractility.

136

5.4.4 Does XO and NOX inhibition alter the effect of DOX on myofilament sensitivity?

DOX was shown to increase myofilament sensitivity using phase-plane loops (fig 3.12) when cells were exposed to FEB and VAS this increase in sensitivity was not apparent (fig 5.8). ROS has been found to increase myofilament sensitivity (Haizlip *et al.*, 2012; Hiranandani & Janssen, 2009; Miura *et al.*, 2015). Whereas NOX inhibition has been shown to decrease myofilament sensitivity, likely due to the reduction in ROS (Valdés *et al.*, 2018). Xanthine oxidase inhibition with allopurinol was shown to increase myofilament sensitivity (Pérez *et al.*, 1998), given that ROS would be reduced this is surprising.

The reduction in ROS by both FEB and VAS as shown in figure 5.2 along with the lack of alteration to myofilament sensitivity (fig 5.8) indicate that DOX-induced OS through both xanthine oxidase and NOX causes an increase myofilament sensitivity. It is also possible that NOX and xanthine oxidase inhibition desensitise the myofilaments to Ca (Valdés *et al.*, 2018) therefore combating the increase in sensitivity caused by DOX.

5.5 Concluding remarks

In conclusion, xanthine oxidase inhibition resulted in a 22 % - 31 % reduction in DOX-induced OS on average. Despite this reduction not being *statistically* significant XO inhibition attenuated various DOX induced effects on Ca handling and contractility suggesting it does indeed play an important role. On average, NOX inhibition resulted in a 50 - 70 % reduction in DOX-induced OS highlighting the substantial role of this enzyme; NOX-mediated production of superoxide is clearly a major mechanism underlying DOX-induced OS. Inhibition of NOX attenuated effects of DOX on Ca handling and contractility. Ca amplitude was still reduced however this reduction was significantly less than DOX treatment alone. Myofilament sensitivity was not altered by DOX following ROS inhibition. These data highlight the potential role of XO in DOX cardiotoxicity however NOX has been shown to be a major contributor in DOX-induced OS.

Chapter 6

The effects of doxorubicin and daunorubicin on Ca handling and contractility in sheep ventricular myocytes from sheep with heart failure

6.1 Introduction

Heart failure (HF) is a leading cause of morbidity and mortality and can manifest as systolic or diastolic dysfunction (Lloyd-Jones *et al.*, 2002). This results in a variety of symptoms such as laboured breathing, fatigue and peripheral oedema (swelling due to fluid accumulation) (King *et al.*, 2012). Those with heart failure will often have an ejection fraction of < 40 % however new terminology for patients with HF with a mid-range ejection fraction (40 – 49 %) has emerged to aid research and treatment of the subpopulation (Ponikowski *et al.*, 2016). Ejection fraction may be used as a predictor of mortality. Diagnosis for HF requires a combination of examinations and tests such as a review of medical history, chest radiography and electrocardiography (King *et al.*, 2012). The cellular basis of heart failure is attributed to the loss of cardiac myocytes and altered Ca handling (Briston *et al.*, 2011a; Mittmann *et al.*, 1998).

6.1.1 Altered Ca handling and contractility in heart failure

This study looks at the response of cells isolated from a HF large animal model to DOX and DAUN. Chapter 3 shows and DOX and DAUN alter Ca handling and contractility in cardiac myocytes, we also know from the research of others that Ca handling is altered (Baartscheer *et al.*, 2003a; Clarke *et al.*, 2015; Sipido *et al.*, 1998). Therefore, those with pre-existing heart conditions who also develop cancer may be at a higher risk of anthracycline-mediated complications due to certain adverse cell changes already present. Treatment with anthracyclines may amplify already existing cell changes associated with HF or effects may already be saturated to a level that anthracyclines do not increase alterations. Furthermore, both cancer and HF are associated with aging, while this is not directly relevant to childhood cancer it is still possible for children to have pre-existing heart conditions and have existing cell changes. This study seeks to understand the effect of anthracyclines on cells that already have a HF phenotype, as currently patients with underlying heart conditions are considered higher risk for anthracycline-induced cardiotoxicity (Armenian *et al.*, 2017).

As previously discussed, excitation-contraction coupling underpins heart function and alterations in this can result in a variety of clinical manifestations. While an increase in diastolic Ca has been found in cardiac myocytes isolated from human patients with HF (Sipido *et al.*, 1998), the impact on diastolic Ca in animal models of HF are inconsistent. For example, diastolic Ca was increased in a rabbit animal model of HF (Baartscheer *et al.*, 2003a) but decreases in a sheep model of HF (Briston *et al.*, 2011a). In a canine model of heart failure no change in diastolic Ca was identified (Hobai *et al.*, 2004).

In models of heart failure Ca transient amplitude has been found to be reduced (Clarke *et al.*, 2015; Hobai *et al.*, 2004; Piacentino *et al.*, 2003; Sipido *et al.*, 1998). While some found this reduction in amplitude could be explained by a decrease in SR Ca content (Hobai *et al.*, 2004; Piacentino *et al.*, 2003), in atrial myocytes from a model of HF, SR Ca content was increased (Baartscheer *et al.*, 2003b; Briston *et al.*, 2011a; Clarke *et al.*, 2015). Alterations in sheep ventricular cardiac myocytes Ca amplitude was likely due to decreased L type Ca current rather than a reduction in SR Ca (Briston *et al.*, 2011a).

In myocytes isolated from a HF animal model, intracellular Na has been found to be increased (Despa *et al.*, 2002) which favours reverse mode NCX activity facilitating the influx of Ca (Schillinger *et al.*, 2003). This Ca influx in HF may offset the effects in SR Ca load (Bers *et al.*, 2006). The initial increase in Na may be due to a decrease in Na removal via Na/K pump or increased influx of Na (Bers *et al.*, 2006).

Evidence also exists that suggests in end stage chronic heart failure the expression of SERCA is reduced (Kubo *et al.*, 2001). Interestingly, in the large animal model for HF the expression of SERCA and phospholamban were not altered (Briston *et al.*, 2011a). An integrated approach must be taken when observing the alterations in individual channels involved in Ca handling in heart failure models to understand the pathophysiological changes.

6.1.2 Heart failure and anthracyclines

OS has been found to contribute to the progression of various cardiovascular diseases and development of heart failure (Moris *et al.*, 2017a). As discussed in 4.1.3 ROS have been shown to alter Ca handling and contractility. A decrease in Ca transient amplitude, reduced SR Ca content and decreased SERCA activity are key observations in cardiac myocytes as a result of ROS (Greensmith *et al.*, 2010; Kim *et al.*, 2006b; Kuster *et al.*, 2010; Sag *et al.*, 2013).

Anthracyclines increase oxidative stress via stimulation of ROS production as discussed in 1.5.2.2 and evidenced in 4.3.3. DOX-induced OS has been shown to play a vital role in alteration to Ca handling and contractility in this study (chapter 4 and 5). The already elevated level of ROS in the heart failure model may lead to no further alterations in Ca handling and contractility as effects may be saturated as discussed in 6.1.1.

The potential of cardiac dysfunction is discussed with patients prior to treatment and those considered at higher risk may undergo prevention and screening strategies to allow intervention if warranted (Armenian *et al.*, 2017). Patients considered high risk are those on a high dose of anthracycline, compromised cardiac function and more than two cardiovascular risk factors such as obesity and diabetes (Armenian *et al.*, 2017). While some factors such as smoking would not be relevant for children some may suffer from other cardiac conditions and the effect anthracyclines in these patients must be considered. It is generally agreed that pre-existing heart conditions would be worsened by anthracycline treatment. The use of a heart failure model allows the identification of the effect of anthracyclines on cardiac myocytes isolated from a diseased state.

6.1.3 Aims

- Use integrative analysis to measure the effects of DOX and DAUN on ECC in the dynamic setting of intact ventricular myocytes from a sheep model of heart failure.

- Investigate if DOX and DAUN result in altered myofilament sensitivity in cells from a sheep model of heart failure.
- Establish if the effects of DOX and DAUN are enhanced in cardiac myocytes isolated from a heart failure model compared to a control model.

6.2 Methods

6.2.1 Heart failure model

Sheep used to produce a heart failure model were ~ 18 months old (matched to healthy model in previous results). Transvenous right ventricular tachypacing was utilised to induce heart failure. A pacing lead was positioned at the apex of the right ventricle and was attached to a Medtronic Thera cardiac pacemaker (located in a cervical subcutaneous pocket). After surgical recovery (10 - 14 days) the pacemaker was activated and tachypacing commenced. Hearts were paced at 210 bpm for ~ 50 days. Sheep were monitored for symptoms of heart failure such as lethargy and shortness of breath. To determine end-stage heart failure alterations in fractional shortening were determined using echocardiography on conscious, non-sedated animals. Animals were sacrificed once end-stage heart failure was determined. The same model methods are described in other articles (Clarke *et al.*, 2015; Dibb *et al.*, 2009).

All procedures to generate heart failure models were carried out at the University of Manchester by the Trafford/Dibb research group and were in accordance with the Animals (Scientific Procedures) Act, UK, 1986 and Directive 2010/63/EU of the European Parliament. All procedures carried out at the University of Manchester have ethical approval from the University of Manchester Welfare and Ethical Review board and the experiments have ethical approval from the University of Salford ethical review board.

6.2.2 Cell preparation

Primary left ventricular cardiomyocytes were isolated from ~18-month-old sheep with induced heart failure (see section 2.2.2). Cells were loaded with fura-2 (see 2.6.2) and stored in 0.95 mM Ca by combining NT and taurine solution (see 2.6.2) in equal parts.

6.2.3 Excitation-contraction coupling measurements

As described in section 2.5 cells were paced at 0.5 Hz using field stimulation in NT. Ca and sarcomere length transients were obtained in NT once steady state was reached before switching to 1 nM DOX or DAUN. After a new steady stage was established, the reversibility of any effect was determined by washing out the drug using NT. Recordings were calibrated and analysed offline using custom software as described in section 2.8.

6.2.4 Measurement of sarcoplasmic reticulum Ca content and Ca removal mechanisms

10 mM caffeine was utilised to estimate changes in SR Ca content. Caffeine increases the open probability of RyR and results in mass Ca release from the SR (Eisner *et al.*, 2017).

The rate constant of decay of systolic Ca transients (k_{sys}) represents the removal of Ca via main mechanisms. In mammalian cardiac myocytes these are NCX and SERCA as discussed in 1.2.9.1. The RC of decay of the caffeine-evoked Ca transient (k_{caff}) represents the activity of NCX due to caffeine rendering the activity of SERCA ineffective. Subtraction of k_{caff} from k_{sys} represents the activity of SERCA (k_{SERCA}).

6.2.5 Myofilament sensitivity measurements

Phase-plane loops were plotted as described in 3.2.4. The loops proceed counter-clockwise with cell shortening depicted by the ascending limb (right hand side of loop). The EC_{50} for each cells phase-plane loop was determined to quantify any change in myofilament sensitivity. Statistical significance was taken as p < 0.05 and calculated using a paired T-test.

6.3 Results

6.3.1 A detailed investigation of the effects of DOX on Ca and contractility in a heart failure model

The specimen traces in figure 6.1 show the effect of 1 nM DOX on intracellular Ca and contractility from a heart failure model and are representative of the means shown in figure 6.2. On average, 1 nM DOX had no significant effect on diastolic Ca (n = 18, p = 0.144; fig 6.2A) or diastolic sarcomere length (n = 18, p + 0.718; fig 6.2B). However, diastolic Ca significantly decreased by 6.32 % \pm 2.78 % in washout (n = 18, p = 0.03; fig 6.2A) accompanied by a significant decrease in diastolic sarcomere length by 0.58 % \pm 0.26 % (n = 18, p = 0.021; fig 6.2B). On average, DOX reduced Ca transient amplitude by 22.73 % \pm 2.95 % (n = 19, p < 0.001; fig 6.2C) and did not recover in washout with a decrease of 30.3 % \pm 4.76 % (n = 19, p < 0.001; fig 6.2C). This effect on Ca was accompanied by a 27.51 % \pm 4.98 % (n = 16, p = 0.025) decrease in sarcomere shortening which recovered upon washout (n = 16, p = 0.741; fig 6.2D). On average, peak Ca decreased by 14.28 % \pm 1.94 % (n = 19 < 0.001) and did not recover upon washout (n = 19, p < 0.001; fig 6.2E). On average, the rate of sarcomere relaxation decreased by 19.52 % \pm 4.61 (n = 16, p = 0.02) and recovered upon washout (n = 16, p = 0.273; fig 6.2F).



Figure 6.1: The effect of 1 nM DOX on cytoplasmic Ca and sarcomere shortening in cardiac myocytes from a heart failure model. (A) long time base record that shows specimen Ca transients. (B) specimen contractility transients. (C) Ca transient overlays. (D) contractility transient overlays. Long time base records represent steady state from which five Ca transients were averaged to produce the overlays. Data from this single cell is representative of the mean data shown in fig 6.2.



Figure 6.2: The effect of 1 nM DOX on intracellular Ca and contractility in sheep ventricular **myocytes from a heart failure model.** (A) mean normalised diastolic Ca. (B) mean normalised diastolic sarcomere length. (C) mean normalised Ca transient amplitude. (D) mean normalised

sarcomere shortening. (E) mean normalised peak Ca. (F) mean normalised 90-10 % decay time of sarcomere length. n = 16 - 19 (from 3 animals), * denotes statistical significance.

6.3.2 The effect of doxorubicin on SR Ca content and rate of Ca removal in a heart failure model

As described in section 6.2 caffeine enables the investigation of changes in SR Ca content. The specimen shown in figure 6.3A shows the effect of 1 nM DOX on the amplitude of the caffeine evoked Ca transient. On average, this reduced by 28.47 % ± 2.19 (n = 9, p < 0.001) progressing to a 48.28 % ± 4 % reduction in washout (n = 9, p < 0.001; fig 6.4A). Specimen figure 6.3B illustrates the effect of DOX on the Ca transient rate of constant decay (k_{sys}). k_{sys} was unaltered by DOX (n = 19, p = 0.673; fig 6.4B). The caffeine-evoked Ca transient rate constant of decay (k_{coff}) was not significantly altered (n = 9, p = 0.238; fig 6.4C). On average, k_{SERCA} was not significantly altered (n = 9, p = 0.572; fig 6.4D).



Figure 6.3: The effect of 1 nM DOX on SR Ca content in a heart failure model. (A) specimen Ca transients in the presence of caffeine. (B) specimen normalised Ca transients to permit direct comparison of the rate of Ca decay. (C) specimen normalised Ca transients to permit direct comparison of the rate of Ca decay in the presence of caffeine.



Figure 6.4: The effect of 1 nM DOX on Ca removal in sheep ventricular myocytes from a heart failure model. (A) mean normalised amplitude of caffeine evoked Ca transient. (B) mean normalised k_{sys} . (C) normalised mean k_{caff} . (D) mean normalised k_{SERCA} . n = 9 (from 3 animals), * denotes statistical significance.

6.3.3 A detailed investigation of the effects of daunorubicin on Ca and contractility in a heart failure model

The specimen traces in figure 6.5 show the effect of 1 nM DAUN on intracellular Ca and contractility. On average, diastolic Ca was not altered by 1 nM DAUN (n = 12, p = 0.249), however a significant decrease of 4.03 % \pm 1.79 % was observed upon wash out (n = 12, p = 0.03; fig 6.6A). On average, diastolic sarcomere length was not significantly altered in DAUN (n = 11, p = 0.338; fig 6.6B). A significant decrease in Ca transient amplitude of 24.28 % \pm 3.56 % (n = 12, p < 0.001) was produced by DAUN which did not recover upon wash (n = 12, p < 0.001; fig 6.6C). This was accompanied by a 33.34 % \pm 9.74 % decrease in sarcomere shortening (n = 10, p < 0.01) that recovered upon washout (n = 11, p = 0.999; fig 6.6D). On average, on the Ca transient rate of constant decay (k_{sys}) was unaltered (n = 12, p = 0.353; fig 6.6E). On average, sarcomere relaxation was not significantly altered (n = 9, p = 0.156; fig 6.6F).



Figure 6.5: The effect of 1 nM DAUN on cytoplasmic Ca and sarcomere shortening in cardiac myocytes from a heart failure model. (A) long time base record that shows specimen Ca transients. (B) specimen contractility transients. (C) Ca transient overlays. (D) contractility transient overlays. Long time base records represent steady state from which five Ca transients were averaged to produce the overlays. Data from this single cell is representative of the mean data shown in figure 6.6.



Figure 6.6: The effect of 1 nM DAUN on intracellular Ca and contractility in sheep ventricular myocytes from a heart failure model. (A) mean normalised diastolic Ca. (B) mean normalised diastolic sarcomere length. (C) mean normalised Ca transient amplitude. (D) mean

normalised sarcomere shortening. (E) mean normalised peak Ca. (F) mean normalised 90 – 10 % decay time of sarcomere length. n = 9 - 12 (from 3 animals), * denotes statistical significance.

6.3.4 The effects of doxorubicin and daunorubicin on myofilament sensitivity in a heart failure model

Figure 6.7A and C show representative phase-plane loops for cells treated with 1 nM DOX and DAUN from a heart failure model. Phase-plane loops proceed counter-clockwise and plot the change of cytoplasmic Ca versus the change of sarcomere length in the absence and presence of DOX and DAUN. Neither DOX or DAUN resulted in a shift leftward or rightward in the loops (fig 6.7A & C). To quantify if a shift occurred the intracellular Ca concentration required for half relaxation (EC₅₀) was measured. On average, DOX and DAUN did not alter the EC₅₀ (n = 18, p = 0.204; fig 6.7B) and (n = 12, p = 0.401; fig 6.7D) respectively.



Figure 6.7: The effect of DOX and DAUN on myofilament sensitivity in sheep left ventricular myocytes from a heart failure model. (A) phase-plane loop of sarcomere length versus cytoplasmic Ca when treated with 1nM DOX. (B) EC_{50} (cytoplasmic Ca at 50 % relaxation) of cells in NT compared to 1nM DOX treatment. (C) phase-plane loop of sarcomere length versus cytoplasmic Ca when treated with 1nM DAUN. (D) EC_{50} of cells in NT compared to 1nM DAUN treatment. n = 12 - 18 (from 3 animals), * denotes statistical significance.

6.3.5 A comparison of the effects of DOX on global Ca and contractility in cells from healthy and HF sheep

Data from healthy sheep presented in chapter 3 was used for comparison purposes.

In cells from healthy sheep, the effect of DOX on the Ca transient amplitude was significantly greater than that in cells from HF sheep (healthy; - 39 %, HF; - 23 %, n = 19, p = 0.0023; fig 6.8A). On average, the effect of DOX on peak Ca was not significantly different between cells from healthy and HF sheep (n = 19, p = 0.355; fig 6.8C). In cells from healthy sheep, the effect of DOX on sarcomere length shortening was greater than that in cells from HF sheep (healthy; - 42 %, HF; - 28 %, n = 16, p = 0.036; fig 6.8B). On average, the effect of DOX on the rate of sarcomere relaxation was greater in cells from healthy sheep than that of cells from HF sheep (healthy; + 39 %, HF; - 20 %, n = 16, p < 0.001; fig 6.8D).





6.3.6 A comparison of the effects of DOX on SR Ca and Ca removal in cells from healthy and HF sheep

No significant difference in the effect of 1 nM DOX on the amplitude of the caffeine evoked Ca transient between cells from healthy and those from HF sheep was identified (n = 9, p = 0.598; fig 6.9A). On average, the in cells from healthy sheep the effect of DOX on the Ca transient of constant decay (k_{sys}) was greater than that of cells from HF sheep (healthy; - 17 %, HF; - 0.4 %, n = 19, p = 0.0018; fig 6.9B).The effect of DOX on k_{caff} in cells from healthy sheep was greater than that of cells from HF sheep (healthy; sheep was greater than that of cells from HF sheep however this was not statistically significant (healthy; + 64 %, HF; + 7 %, n = 9, p = 0.19; fig 6.9C). The effect of DOX on k_{SERCA} was greater in cells from healthy sheep compared to HF however this was not statistically significant (healthy; - 34 %, HF; + 7 %, n = 9, p = 0.058; fig 6.9D).



Figure 6.9: The effect of 1 nM DOX on Ca removal in sheep ventricular myocytes isolated from healthy versus heart failure animal model. (A) mean normalised percentage change in the amplitude of caffeine evoked Ca transient (B) mean normalised percentage change k_{sys} . (C) mean normalised percentage change k_{caff} . (D) mean normalised percentage change in k_{SERCA} . n = 9 – 19 (from 3 - 5 animals), * denotes statistical significance.

6.3.7 Alterations in response to daunorubicin in Ca handling and contractility between a healthy and heart failure model

In cells from healthy sheep, the effect of DAUN on the Ca transient amplitude was significantly greater than that in cells form HF sheep (healthy; - 45 %, HF; - 25 %, n = 12, p < 0.001; fig 6.10A). On average, the effect of DAUN on Ca transient of constant decay (k_{sys}) was greater in cells from healthy compared to cells from HF sheep (healthy; - 23 %, HF; - 3%, n = 12, p = 0.0035; fig 6.10C). In cells from healthy sheep, the effect of DAUN on sarcomere length shortening was greater than that in cells from HF sheep (healthy; - 64 %, HF; - 33 %, n = 10, p = 0.0026; fig 6.10B). The rate of sarcomere relaxation was affected to greater extent by DAUN in cells form healthy sheep compared to cells from HF sheep (healthy; + 88 %, HF; + 7 %, n = 9, p = 0.042; fig 6.10D).



Figure 6.10: The effect of 1 nM DAUN on Ca handling and contractility in healthy versus heart failure animal model. (A) mean normalised percentage change in Ca transient amplitude. (B) mean normalised percentage change in sarcomere length shortening. (C) mean normalised percentage change in k_{sys} . (D) mean normalised percentage change in rate of sarcomere relaxation. n = 9 – 20 (from 3 – 6 animals), * denotes statistical significance.
6.4 Discussion

6.4.1 Does 1 nM doxorubicin alter intracellular Ca and contractility in sheep cardiomyocytes isolated from a heart failure model?

1 nM DOX did not alter diastolic Ca or sarcomere length in sheep ventricular myocytes isolated from a heart failure model (fig 6.2), however the decrease upon washout was significant. Briston *et al* (2011) found a decrease in diastolic Ca in the sheep model of HF. If a decrease in diastolic Ca occurs our findings suggest that DOX has an irreversible effect on diastolic Ca leading to a further reduction upon washout.

In HF cells DOX resulted in an irreversible decrease in Ca transient amplitude (fig 6.2C) which was accompanied by a decrease in sarcomere shortening that did recover upon washout (fig 6.2D). As discussed in 6.1.1 Ca transient amplitude is known to reduced in HF (Clarke *et al.*, 2015; Hobai *et al.*, 2004; Piacentino *et al.*, 2003; Sipido *et al.*, 1998). ROS are also known to be increased in HF (Moris *et al.*, 2017) and can result in a decrease in Ca amplitude (Goldhaber & Liu, 1994; Greensmith *et al.*, 2010; Kuster *et al.*, 2010) (see 4.1.2). The effect of DOX on Ca amplitude was not saturated by the effects of HF alone and so the rise in ROS by DOX may not be the only influence on Ca amplitude and sarcomere shortening. This is supported by a reduction in Ca amplitude and sarcomere shortening still being evident due to DOX after ROS inhibition (see 4.3.5).

The change in Ca transient amplitude is supported by a sustained decrease in peak Ca (fig 6.2E). Peak Ca represents absolute systolic Ca levels whereas Ca transient amplitude represents the difference between peak Ca and diastolic Ca (Greensmith, 2014b). Therefore, a change in diastolic Ca may offset any change to peak Ca resulting in amplitude being unaltered. The alteration in both amplitude and peak Ca supports no alteration in diastolic Ca.

The rate of sarcomere relaxation was decreased by 1 nM DOX in cells from HF (fig 6.2F) which suggests an increase in Ca removal however this was not detected as discussed in 6.4.2.

6.4.2 Is SR Ca and the activity of the Ca removal mechanisms altered by 1 nM doxorubicin in a heart failure model?

DOX decreased SR Ca content in cardiac myocytes isolated from HF model. The amplitude of the Ca transient is proportional to the third power of SR Ca (Dibb *et al.*, 2007). DOX reduced SR Ca content to 71.5 % of control (fig 6.4A). This would be expected to decrease the Ca transient amplitude by $(0.715)^3 = 36.6$ %. The Ca transient amplitude decreased to 77 % indicating that SR Ca content may not be the only factor responsible for the reduction in Ca transient amplitude. This is supported by (Clarke *et al.*, 2015). Clarke *et al* (2015) found HF SR Ca content was increased but Ca transient amplitude was reduced, they determined that a decrease in the L-type Ca current underpinned the alteration to Ca transient amplitude. Therefore, it may be in this model that the L-type Ca channel plays a greater role in reducing the Ca transient amplitude than SR Ca content.

 K_{sys} , k_{caff} and k_{SERCA} were not significantly altered by DOX in HF cells (fig 6.4). This suggests that Ca removal is not responsible for the decrease in SR Ca content. It is possible that an increase in RyR leak results in a reduction in SR CA content (Bers *et al.*, 2003).

6.4.3 Does 1 nM daunorubicin alter intracellular Ca and contractility in sheep cardiomyocytes isolated from a heart failure model?

Diastolic Ca and sarcomere length were unaltered in 1 nM DAUN (fig 6.6A & B). This response is similar to that seen in cells isolated from control animals (see fig 3.9). This suggests that DAUN does not alter diastolic Ca or sarcomere length regardless of model. An irreversible decrease in Ca transient amplitude was detected (fig 6.6C) and this was accompanied by a reversible decrease in sarcomere shortening (fig 6.6D). As discussed, Ca amplitude is reduced in HF models (see 6.1.1). Treatment with DAUN prevents recovery of Ca transient amplitude in HF cells suggesting that the effect is too severe to be reversed. Sarcomere shortening recovers upon wash which may be due to an increase in myofilament sensitivity (discussed in 6.4.4). k_{sys} and rate of sarcomere relaxation were not altered by DAUN (fig 6.6E & 6.6F). While the effect on Ca removal could not be tested due to the previously mentioned interaction between DAUN and caffeine (see 3.3.6), the lack of change in k_{sys} suggests no change in Ca removal. This was the case for DOX (see 6.4.2).

6.4.4 Do doxorubicin and daunorubicin alter myofilament sensitivity in a heart failure model?

DOX and DAUN did not alter myofilament sensitivity in cells isolated from HF (fig 6.7) however in cells from healthy cells an increase in myofilament sensitivity was detected (see figure 3.12).

Myofilament sensitivity has been found to be increased in cardiac myocytes isolated from mammalian models of heart failure (van der Velden *et al.*, 2003) but the increase in sensitivity to Ca does not result in an increase in force due to changes in pH (van Der Velden *et al.*, 2001). This increase in sensitivity is thought to be due to reduction of β -adrenergic phosphorylation of troponin 1 (Wolff *et al.*, 1996). However, no change in myofilament sensitivity regardless of HF severity has also been found (Denvir *et al.*, 1995). While these studies provide insight into the possible mechanisms of increased myofilament sensitivity in HF, the issue with human samples is the lack of a control due to treatment already carried out in patients (Marston & de Tombe, 2008).

If in the case of the studies discussed myofilament sensitivity is increased in the sheep HF model, then our results suggest that the effect is saturated to the extent that neither DOX nor DAUN increase sensitivity further.

6.4.5 Do the effects of doxorubicin on Ca handling and contractility differ between a healthy and heart failure model?

In cells from the HF model Ca amplitude was reduced to a lesser extent by DOX than in control (fig 6.8A). This may be due to the fact the baseline Ca transient amplitude is already reduced in HF model as discussed in 6.1.1. ROS has been found to be elevated in HF (Hill & Singal, 163

1996; Ide *et al.*, 1999; Moris *et al.*, 2017a). ROS reduced Ca transient amplitude in cardiac myocytes (Greensmith *et al.*, 2010; Kuster *et al.*, 2010) therefore is it logical that Ca transient amplitude will be at a reduced state in cells from HF compared to control. This may mean that the effects of DOX on Ca transient amplitude are saturated in the HF model.

DOX altered sarcomere shortening and sarcomere relaxation to a lesser extent in cells from the HF model (fig 6.8B & D). As discussed in 6.4.4 myofilament sensitivity may be increased in the HF model. DOX has also been shown to increase myofilament sensitivity (fig 3.12). The effects on myofilament sensitivity may be saturated in the HF model meaning that the effect of DOX on sarcomere shortening is to a lesser extent than would be expected.

There was no statistically significant difference in the reduction of SR Ca content between cells from healthy animals and cells from the HF model (fig 6.9A). However, k_{sys} was not altered in HF. This suggests that the effect of DOX on Ca removal channels such as SERCA and NCX differs in the HF model. This is not surprising given the notable difference in activity and expression of both channels in HF models. In HF SERCA expression has been found to be reduced (Zhihao *et al.*, 2020) resulting in a decrease in SERCA activity while NCX activity is increased but in reverse mode (Schillinger *et al.*, 2003). This suggests that less Ca is actively transported into the SR however more Ca is brought in through the reverse mode of NCX. Overall, these findings may explain why k_{sys} is not altered by DOX in the HF model as NCX and SERCA activity is not altered possibly due to the effect on the channels being saturated and/or altered expression.

In healthy cells DOX resulted in a decrease in SERCA activity (fig 3.7). The effects in HF suggest that the decrease in SERCA activity is saturated due to the alterations already present and aren't worsened by DOX as the activity of SERCA was not altered (fig 6.4D). NCX activity was not altered in HF cells by DOX (fig 6.4C) whereas in healthy cells the activity was increased (fig 3.7).

The decrease in SR Ca content in HF due to DOX cannot be accounted for by removal of Ca as is the case for cells from healthy sheep (see chapter 3). Therefore, as discussed in 3.1.1 DOX may alter L-type Ca channel activity as well as the open probability of RyR which can impact SR Ca content.

6.4.6 Do the effects of daunorubicin on Ca handling and contractility differ between a control and heart failure model?

The change in Ca transient amplitude was altered to a lesser extent in the heart failure model (fig 6.10A). The explanation for this may be similar to that discussed in 6.4.5, however results from chapter 4 found that DAUN only increased OS at a higher concentration (10 μ M). As previously discussed, SR Ca content is a major determinant of Ca transient amplitude. Given the apparent interaction between caffeine and DAUN we were unable to investigate this. However, k_{sys} was unaltered in HF cells by DAUN indicating the combined activity of SERCA and NCX is unaltered. DAUN may alter the open probability of RyR and/or activation of L-type Ca channels as discussed in 3.1.1 which in turn may alter SR Ca content.

6.5 Concluding remarks

In summary, DOX and DAUN decrease Ca transient amplitude which may be partially explained by a decrease in SR Ca content. In HF this decrease does not recover upon washout as was the case for healthy cells suggesting cells from HF are less able to recover from the effects of the DOX and DAUN. In HF models DOX and DAUN may lead to an increase in the open probability of RyR or alteration in L-type Ca channel activation which could account for a reduction in SR Ca content that cannot be accounted for NCX and SERCA. An increase in myofilament sensitivity was detected in healthy cells treated with DOX and DAUN however this was not altered in the HF model. The potential shift in myofilament sensitivity already induced in the HF model may mean the effects are saturated to a point were DOX and DAUN cause no further alteration. The effects of DOX and DAUN on cells from a HF sheep are not amplified in comparison to the cells form the healthy model suggesting that rather than having an additive effect the alterations in Ca handling and contractility are saturated.

Chapter 7 General Discussion

7.1 The effects of doxorubicin and daunorubicin on excitationcontraction coupling

7.1.1 The effect of doxorubicin and daunorubicin on intracellular Ca and contractility

This study examined the effects of both DOX and DAUN on Ca handling and contractility in intact sheep cardiac myocytes. This is because, while other studies have previously investigated the effects of DOX on aspects such as Ca transient amplitude and contractility very few have studied the effects of DAUN with the assumption being the mechanism of cardiotoxicity is the same. Furthermore, most of these studies were conducted using artificial planar membranes or small animal models and look at components individually. Our experiments overcome this by using cardiomyocytes from a large animal model along with an integrative approach to study aspects of ECC in a dynamic setting.

In sheep cardiomyocytes DOX resulted in a decrease in Ca transient amplitude and sarcomere shortening (fig 3.5). This is supported by studies using other models (Maeda *et al.*, 1997; Sag *et al.*, 2011; Temma *et al.*, 1996; Zheng *et al.*, 2011). While these studies provide insight there are limitations. As mentioned most studies assume that DAUN will behave in the same way as DOX. While DAUN also decreased Ca transient amplitude and contractility it was to a greater extent than that observed with DOX (fig 3.9). This is interesting given that DAUN has been shown to result in less cardiotoxicity in childhood cancer survivors (Feijen *et al.*, 2015b). The effects of both DOX and DAUN on Ca amplitude and sarcomere shortening were reversed upon washout. This suggests that any interactions with transporters are reversible upon washout and do not cause permanent changes to channels and are able to recover.

Previous studies show DOX increases diastolic Ca (Mijares & López, 2001; Sag *et al.*, 2011). However, we are aware of no studies where the associated effects on cell / sarcomere length have been studied in parallel. In our study, DOX did not alter diastolic Ca or sarcomere length. Both studies that identified an increase in diastolic Ca were conducted in rat cardiac 167 myocytes. Although the fundamental processes of ECC are highly conserved it is recognised that small animal models may respond differently. This is due to evolutionary differences that result in changes in heart rate, cardiac output and contractile kinetics that are underpinned by variation in excitation and Ca handling (Milani-Nejad & Janssen, 2014) which may explain the difference in diastolic Ca.

7.1.2 The effect of doxorubicin and daunorubicin on SR Ca content

Other studies provide *indirect* evidence that suggests DOX reduces SR Ca content (Ondrias *et al.*, 1990; Sag *et al.*, 2011), though an effect of DAUN on SR Ca has yet to be reported. To our knowledge, we are the first to use a direct method to measure the effect of DOX *and* DAUN on SR Ca and show both produce a decrease.

As discussed in 1.2.9.2 this decrease in SR Ca content can account for the reduced Ca transient amplitude as shown in 3.4.3 and 3.4.5. We acknowledge that we do not know if DOX or DAUN alter intracellular Ca buffering; thus, the relationship between total SR Ca release and free Ca (to which fura–2 fluorescence is proportional), therefore it is not possible to quantify the absolute change of SR Ca content (Trafford *et al.*, 1999a; Varro *et al.*, 1993). Nonetheless, as the relative change of caffeine-evoked [Ca]₁ transient amplitude is greater than could be accounted for by a change to buffering, we are confident that a reduction of SR Ca is a key contributor to the reduced [Ca]₁ transient.

Our study also highlights a potential interaction between caffeine and DAUN. Others note that DAUN can bind to the RyR and this results in reduced SR Ca release in response to caffeine application (Abramson *et al.*, 1988). In effect, DAUN acts as a non-competitive inhibitor. If this were the case in our experiments, it would suggest that the reduction in SR Ca content we identified in figure 3.10 is underestimated. This potential interaction was brought to our attention by a lack of change in k_{sys} in the presence of caffeine (that was reduced without caffeine) (see figure 3.11). This suggests that in the presence of caffeine DAUN no longer alters the overall removal of Ca.

7.1.3 The effect of doxorubicin and daunorubicin on the mechanisms of Ca removal

DOX was found to decrease SERCA activity and Hanna et al (2014) have identified DOX can bind to SERCA, therefore it may alter SERCA activity through this interaction. The effect of DOX on NCX has been investigated to a lesser extent with some studies identifying no change in protein expression (Llach *et al.*, 2019) and others finding a decrease in expression (Olson *et al.*, 2005). However, alterations to the activity of the transporter remain unknown. As discussed in 1.2.9.4 SERCA and NCX are powerful regulators of SR Ca content (Eisner, 2014). Our study identified an increase in NCX activity that, along with the decrease in SERCA activity can explain the decrease in SR Ca content. The effect of DAUN on Ca removal could not be investigated due to the apparent interaction with caffeine as discussed in 7.1.2.

7.1.4 The effect of doxorubicin and daunorubicin on myofilament sensitivity to Ca

There are limited studies investigating the effects of DOX or DAUN on myofilament sensitivity as discussed in 3.4.6. Alteration in myofilament sensitivity alters the Ca-force relationship. For example, increased myofilament sensitivity leads to a leftward shift in the force versus Ca curve meaning that for any given Ca concentration the force of contraction is increased. Both DOX and DAUN showed evidence of increased myofilament sensitivity (see fig 3.12). This is supported by a study in female rats that found long term treatment with DOX (10 weeks) increased myofilament sensitivity (Rattanasopa *et al.*, 2019), however the Ca – force relationship was not altered by DOX in rabbit cardiac myocytes (Boucek *et al.*, 1993). There are very few studies that investigate the effect of DOX on myofilament sensitivity and to the best of our knowledge none that investigate the impact of DAUN on myofilament sensitivity. For all things being equal the reduction in Ca transient amplitude ought to produce a decrease in sarcomere shortening (see fig 1.8). In 3.4.2 I discuss that the decrease in Ca transient amplitude likely accounts for the reduction in sarcomere shortening however the effect of 169 myofilament sensitivity may make this relationship more complex. It appears that DOX and DAUN are acting as Ca sensitisers which would shift the relationship between Ca amplitude and force leftward. Clearly the net effect of DOX and DAUN is negative inotropy however given the effect on myofilament sensitivity that negative inotropic effect is likely offset.

7.1.5 An integrated model

Our study shows that in a large animal model DOX and DAUN decrease cardiac myocyte contractility which is underpinned by a reduction in systolic Ca. This is largely attributable to a decrease in SR Ca content. In DOX we identified a decrease in SERCA activity and an increase in NCX activity which can account for the overall reduction in SR Ca content. The same could not be established for DAUN due to an apparent caffeine interaction meaning further investigation is required. Both DOX and DAUN showed evidence of acting as Ca sensitisers as myofilament sensitivity was shown to be increased.



Figure 7.1: The effects of DOX and DAUN on Ca handling and contractility in sheep ventricular myocytes.

7.2 The role of oxidative stress in the anthracycline-mediated changes to excitation-contraction coupling

7.2.1 Do doxorubicin and daunorubicin increase oxidative stress?

DOX increased OS in a concentration-dependent manner (fig 4.2D) which is supported by others that have identified DOX results in an increase in ROS (Kim *et al.*, 2006a; Ma *et al.*, 2013; Zhou *et al.*, 2001). These studies are the first to show DOX not only increases ROS in cardiac myocytes but that this leads to an increase in OS and is evidenced by both the concentration and incubation time experiments as well as the addition of NAC which removed this OS response. An increase in OS was only present at the highest concentration of DAUN (10 μ M) (fig 4.2C). Studies conducted to identify ROS production in cardiac myocytes due to DAUN are extremely limited with one study confirming that 1 μ M DAUN did not increase ROS (Sawyer *et al.*, 1999).

The studies that show DOX increases ROS support our findings that DOX increases OS. This provides a clear basis for altered ECC as discussed in 7.2.3. Interestingly the OS response following DAUN treatment differs indicating that OS may not be responsible for the effects on ECC identified. It is also possible that the lack of OS via DAUN contributes to this anthracycline being associated with less cardiotoxicity in childhood cancer survivors (Feijen *et al.*, 2015b).

7.2.2 What is the source of DOX-induced oxidative stress?

As shown in chapter 4 DOX increased OS in a concentration-dependent manner however the mechanism by which this is achieved is unknown. As discussed in 4.1.1 ROS may be produced by various mechanisms in cardiac myocytes including the mitochondrial electron transport chain, XO and NOX. While the mitochondrial electron transport chain may be impacted by drugs this study focused on the enzymatic sources and investigated the role of XO and NOX on DOX-induced OS.

7.2.3 The role of xanthine oxidase in DOX-induced oxidative stress

XO is an enzyme involved in the production of uric acid and ROS as a biproduct of hypoxanthine metabolism (Kang & Ha, 2014). Clinically XO inhibitors such as allopurinol and febuxostat are used for gout treatment (Frampton, 2015). Inhibition of this enzyme has also been shown to improve contractility and left ventricular function in heart failure models (Ekelund *et al.*, 1999; Ukai *et al.*, 2001). Our study identified DOX resulted in a reduction in contractility due to a decrease in systolic Ca as shown in figure 7.1 as well as a rapid increase in OS. As discussed in 4.1.3 ROS may alter Ca handling and contractility and the attenuation of these effects by XO inhibition suggests this enzyme may have vital role. We sought to investigate if XO has a role in DOX-induced OS and if inhibition of this enzyme attenuated the effects on ECC.

Inhibition of XO via febuxostat reduced OS in the presence of DOX at all time points (though not significantly). Ca transient amplitude was reduced by 8 % in DOX with a significant decrease of 18.8 % being evident upon washout (fig 5.4). The rate of sarcomere relaxation was significantly reduced in wash. All other aspects of Ca handling and contractility measured were not altered by DOX following XO inhibition (fig 5.4).

These results indicate that XO contributes (at least partially) to DOX-induced OS production that then alters Ca handling and contractility in cardiac myocytes.

7.2.4 The role of NADPH oxidase in DOX-induced oxidative stress

NADPH oxidases produce ROS as part of the immune response with NOX2 and NOX4 being expressed in cardiac myocytes (Kuroda & Sadoshima, 2010; Szekeres *et al.*, 2021a). Our study showed inhibition of NOX2 and 4 reduced dox-induced OS for all incubation times (fig 5.2). These data highlight the vital role NOX has in mediating production of ROS in cardiac myocytes and thence OS due to DOX treatment. Furthermore, with the exception of Ca transient amplitude the effects of DOX on Ca handling and contractility were attenuated (fig 5.6).

This suggests that NOX ROS production is involved in altering the removal of Ca possibly through altering the activity of channels such as SERCA (Greensmith *et al.*, 2010; Kuster *et al.*, 2010; Morris & Sulakhe, 1997; Reeves *et al.*, 1986; Scherer & Deamer, 1986). The reduction in Ca transient amplitude is likely due a decrease in SR Ca content (as discussed in 1.2.9.2), however k_{sys} was unaltered suggesting the overall removal of Ca was not impacted. This suggests there may be an alteration to the L type Ca current and/or RyR which is not ROS dependent and this leads to a reduction in SR Ca content and therefore Ca amplitude.

7.2.5 Can DOX-induced oxidative stress account for the alterations to excitationcontraction coupling?

We have established that DOX-induces OS and perturbs ECC, we must now consider the role of DOX-induced OS on the ECC components.

In the presence of NAC sarcomere length shortening was reduced to a lesser extent by DOX as was Ca transient amplitude (fig 4.5 & 4.6). It is important to note that even though these parameters were still reduced in the presence of NAC it was to a lesser extent than DOX treatment alone; with approximately half of the effect of DOX on Ca amplitude and systolic shortening attributable to OS (see section 4.4.4). These findings are supported by others that show ROS reduces Ca amplitude and contractility (Goldhaber & Liu, 1994; Greensmith *et al.*, 2010; Kuster *et al.*, 2010).

The decrease in Ca transient amplitude that is present after OS reduction can be accounted for by a decrease in SR Ca content (fig 4.8). NAC pre-treatment attenuated the effects of DOX on SR Ca content by 45 % (fig 4.8A), however the reduction identified can account for the decrease in Ca amplitude. The attenuation effects highlight the impact of OS and others have shown ROS reduces SR Ca content (Boraso & Williams, 1994; Greensmith *et al.*, 2010; Kuster *et al.*, 2010). NAC pre-treatment fully attenuated the effect of DOX on k_{sys} and k_{SERCA} (fig 4.8), this suggests that DOX-induced OS is the cause of reducing SERCA activity. ROS reduces the activity of SERCA (Greensmith *et al.*, 2010; Kuster *et al.*, 2010; Morris & Sulakhe, 1997; Reeves *et al.*, 1986; Scherer & Deamer, 1986) which supports this observation. The effect of DOX on NCX activity was partially attenuated by NAC pre-treatment (fig 4.8C) which suggests that 79 % of the increase in NCX activity can be attributed to DOX-induced OS. This is logical given that ROS has been shown to increase NCX activity (Eigel *et al.*, 2004; Goldhaber, 1996; Kuster *et al.*, 2010; Reeves *et al.*, 1986).

NAC pre-treatment still resulted in a reduction in SR Ca content following DOX exposure (though attenuated by 50 %) and may be explained by the 14 % increase in k_{caff} . Although this increase in NCX activity was not significant the effect on NCX was not fully attenuated by OS reduction suggesting that DOX is altering the transporter activity by another mechanism. This increase in NCX activity though not statistically significant may still contribute to the reduction in SR Ca content as NCX has been shown to play a greater role in SR Ca content than SERCA (Bode *et al.*, 2011). Though not investigated in this study the role of L type Ca channel activity and open probability of RyR may be altered by DOX and also contribute to the decrease in SR Ca content through a mechanism that is not OS dependent.

Studies have found myofilament sensitivity to be increased by ROS (Haizlip *et al.*, 2012; Hiranandani & Janssen, 2009; Miura *et al.*, 2015) which supports the evidence in this study that found the increase in myofilament sensitivity due to DOX was attenuated by the reduction in OS (fig 5.8). This suggest that DOX-induced OS is the primary mechanism that leads to an increase in myofilament sensitivity.

The inhibition of NOX showed similar finding to those identified with NAC pre-treatment and fully attenuated various effects of DOX on Ca handling and contractility with partial attenuation of Ca transient amplitude (fig 5.6 & 5.7). Inhibition of XO decreased DOX-induced 174

OS but to a lesser extent than NOX, however still resulted in attenuation of various parameters (fig 5.7). These findings suggest that NOX has a vital role in DOX-induced OS which in turn perturbs various aspects of ECC and that XO may also contribute to the increase in OS. Both enzymes may increase ROS in cardiac myocytes via reduction of DOX resulting in semi-quinone formation and ROS (Bates & Winterbourn, 1982; Huang *et al.*, 2022; Menna *et al.*, 2010). Given that DOX has been shown to accumulate in the mitochondria of cardiac myocytes (Ichikawa *et al.*, 2014a) and this is where NOX4 is expressed (Ichikawa *et al.*, 2014a; Kuroda *et al.*, 2010; Peoples *et al.*, 2019) it is logical that this enzyme is located in a prime position to facilitate the DOX reduction reaction leading to ROS formation.



Figure 7.2: The role of DOX-induced OS on Ca handling and contractility in sheep ventricular myocytes.

7.2.6 Can DAUN-induced oxidative stress account for the alterations to excitation-contraction coupling?

Chapter 3 shows that DAUN reduces contractility of cardiac myocytes which is underpinned by reduction in systolic Ca due to a decrease in SR Ca content. However, OS was only increased at the highest concentration of DAUN. These findings suggest that DAUN alters ECC components mainly via a mechanism other than OS. This is not to say that ROS don't have a role in DAUN cardiotoxicity but that the increase in ROS production does not seem to result in OS in acute exposure at low concentrations.

This is surprising given the similar chemical structures of DOX and DAUN (see fig 1.14). Our studies also identified a potential interaction between DAUN and caffeine which may explained by the binding both compounds have with caffeine (Abramson *et al.*, 1988). This interaction was not evident with DOX which may suggest the two anthracyclines have different binding properties to channels in the cardiac myocytes.

Overall, the mechanism by which DAUN alters ECC remains elusive. Given that many studies focus on the cardiotoxicity of DOX with the assumption that DAUN will behave in a similar manner our study provides evidence that this may not be the case and more biochemical investigation is required to understand the differences in protein interaction. These differences may account for the differences observed in cardiotoxicity between the compounds (Feijen *et al.*, 2015a).

7.3 The effects of doxorubicin and daunorubicin in heart failure

7.3.1 The effect of doxorubicin and daunorubicin on excitation-contraction coupling in a model of heart failure

Both DOX and DAUN reduced Ca transient amplitude in cardiac myocytes from a sheep model of HF (see chapter 6). Unlike in cells from the healthy model this reduction was not reversed upon washout. As discussed in 6.1.1 Ca transient amplitude is known to reduce in HF. (Clarke *et al.*, 2015; Hobai *et al.*, 2004; Piacentino *et al.*, 2003; Sipido *et al.*, 1998). This indicates that the recovery mechanisms of cells are disturbed within the HF model and may indicate a worse outcome in cardiac myocytes from a disease state, due to the lack of recovery following anthracycline treatment. Furthermore, the overall removal of Ca was not altered in the HF model by DOX indicating that the effects on SERCA and NCX were saturated, however SR Ca content was reduced (fig 6.4). While the reduction in SR Ca content from the HF model cannot be explained by alteration in SERCA and NCX activity it may be that other ECC components are altered by DOX and DAUN. For example, as discussed in 3.1.1 L type Ca channel activity as well as the open probability of RyR may be altered by the anthracyclines which can impact SR Ca content.

7.3.2 Are the nature of the effects of doxorubicin and daunorubicin in health different to those in heart failure?

For both DOX and DAUN the percentage change for Ca transient amplitude and sarcomere length shortening were lower in cells isolated form HF model than the healthy model (see fig 6.8 & 6.10). This may be explained by the already increased state of ROS in HF models (Wagner *et al.*, 2011) meaning the impact of anthracycline-induced ROS is to a lesser extent. Both cells from healthy and HF models showed a decrease in SR Ca content which can account for the decrease in Ca amplitude. However, in cells from healthy animals this change can be accounted for by the increase in NCX activity and decrease in SERCA activity. In the cells from the HF model neither SERCA nor NCX activity was found to be altered by DOX. The altered states of SERCA and NCX in the HF model as discussed in 6.1.1 may be saturated meaning DOX does not induce a further change, however the fact SR Ca content is still reduced suggests other components may be altered such as L-type Ca channels and RyR.

Interestingly oxidative stress is associated with HF (Dubois-Deruy *et al.*, 2020; Grieve & Shah, 2003; Martins *et al.*, 2022). If the cells from the HF model have an increased basal level of OS it may be that these effects are also saturated and that DOX does not increase the levels further. This may explain why the effects on SERCA and NCX are saturated as results from chapter 4 show OS impacts SERCA and NCX activity.

177

7.4 Clinical implications in anthracycline-induced cardiotoxicity

7.4.1 Do the cellular effects of doxorubicin and daunorubicin account for clinical dysfunction?

As discussed in 1.6.1 treatment with anthracyclines can result in a decrease in left ventricular ejection fraction, decreased contractility and patients may develop heart failure (Lipshultz *et al.*, 1991; Nakayama *et al.*, 2020).

Both DOX and DAUN reduced systolic Ca which is the cellular basis of stroke volume/ejection fraction thus inotropy as discussed in 1.2.9. This decrease in systolic Ca was due to a decrease in SR Ca content. DOX resulted in a decrease in SERCA activity and increase in NCX activity which can account for a decrease in SR Ca content. The apparent interaction between DAUN and caffeine prevented exploration of channel activity. These cellular changes in Ca handling and contractility by DOX and DAUN provide a basis for systolic dysfunction and may contribute to the reduction in left ventricular contractility and overall function that is identified in anthracycline cardiotoxicity.

As shown in figure 1.11 the acute reduction in contractility due to DOX recovered only to decline again years after diagnosis (Lipshultz *et al.*, 2005). The experiments in this study were acute with the effects on Ca handling and contractility due to both DOX and DAUN recovering fully or partially upon washout. This cellular recovery may occur clinically after anthracycline treatment only to degrade again years later. Patients that suffer from acute cardiotoxicity due to anthracyclines are more likely to suffer from late cardiotoxicity (Grenier & Lipshultz, 1998).

7.4.2 How may the response to anthracycline treatment differ in patients considered high risk of cardiac complications?

Patients are considered high risk of cardiac complications from anthracycline treatment if they suffer from other cardiac conditions or have cardiovascular risk factors such as obesity and diabetes (Armenian *et al.*, 2017). Both the risk of cancer and heart failure increase with age (Laconi *et al.*, 2020; Stewart *et al.*, 2003; Triposkiadis *et al.*, 2019; White *et al.*, 2014). This suggests that there is a risk of comorbidity in adults meaning they could suffer from both diseases (Bertero *et al.*, 2018; Finke *et al.*, 2021) meaning careful consideration of the use of anthracyclines in this patients is required. In children the risk of this comorbidity is lower but the fact that some children may have underlying heart conditions has to be considered. The use of a heart failure model allowed us to investigate the effects of anthracyclines on cardiac myocytes from a disease state and consider how pre-existing heart conditions would be impacted by anthracycline treatment.

When treated with DOX Ca transient amplitude was irreversibly reduced in cells isolated from HF and the percentage decrease was significantly less than from healthy animals at (HF; -23 %, healthy; - 39 %; fig 6.2 & 6.8). The percentage change in sarcomere length shortening as a result of DOX treatment was also lower in HF cells than healthy cells (fig 6.8). Results were similar for DAUN Ca transient amplitude is reduced in cells from the HF model compared to cells from the healthy model (Briston *et al.*, 2011b). These data suggest that the change inflicted by DOX may appear to be to a lesser extent in terms of percentage change from control but that the overall reduction is similar. The cells from HF do not recover in terms of Ca transient amplitude in wash unlike cells from the healthy model suggesting that recovery would be worse in patients with underlying cardiac conditions.

The overall response appears similar in cells from HF compared to healthy cells however many of the effects appear to have been saturated (at least partially) in the HF model prior to DOX or DAUN treatment. However, recovery upon washout was less successful in cells from the HF model suggesting that DOX and DAUN would worsen the heart function.

While these experiments are reductionist and acute the cellular mechanisms can provide insight into the different responses that may occur. These data suggest that pre-existing conditions that result in reduced systolic Ca and contractility may not be worsened acutely however the long-term implications are unknown.

7.5 Implications for alternative treatment strategies

7.5.1 Compensatory treatment strategies for anthracycline-induced cardiotoxicity

Current treatment of anthracycline-induced cardiotoxicity involves the use of angiotensinconverting enzyme inhibitors however there is no specific treatment strategy in terms of length of treatment; particularly following recovery (Henriksen, 2018). These enzymes are used as they have been shown to slow down the progression of ventricular dysfunction following anthracycline treatment (Kobza, 2021).

Our findings show DOX increased NCX activity while reducing SERCA activity therefore reducing systolic Ca and contractility. The inotropic state is shown to be altered as is the case in heart failure which can result in a larger amount of Ca being released from the SR via RyR, a decrease re-uptake of Ca into the SR via SERCA and an increase in Ca removal from the cell via NCX therefore reducing SR Ca content (Hasenfuss & Teerlink, 2011). Positive inotropic drugs increase the force of contraction ultimately by increasing free intracellular Ca (Scholz, 1984) and may be one avenue of investigation to combat anthracycline cardiotoxicity.

These agents include cardiac glycosides, β - adrenoreceptor agonists, phosphodiesterase inhibitors and calcium sensitisers (Hasenfuss & Teerlink, 2011). Glycosides such as digitalis raise intracellular Ca, reducing Ca efflux via NCX and favour SR Ca uptake (Reuter *et al.*, 2002). β -adrenoreceptor agonists stimulate β -adrenergic receptors and promotes increased force of contraction (Hasenfuss & Teerlink, 2011). Phosphodiesterase inhibitors prevent degradation of cAMP therefore allowing the downstream effects (see figure 1.10) such as promoting the opening of L type calcium channels (Feneck, 2007) and phosphorylation of SERCA (Rao & Xi, 2009) which promotes reuptake of Ca into the SR. These positive inotropic drugs may combat the effects that anthracyclines have on protein channels and restore SR Ca content therefore restoring Ca amplitude and contractility.

Ca sensitiser are another class of positive inotropic drug and as discussed in 3.4.6, they increase the force of contraction by increasing the affinity of troponin C to Ca or increasing the effectiveness of this attachment (Lee & Allen, 1997). These agents may be used to restore the regulation of ECC, and the alterations seen on NCX and SERCA therefore reducing the toxic cardiac side effects. However, our study shows DOX and DAUN increase myofilament sensitivity but the effects on Ca handling and contractility are still pronounced, which indicates the use of Ca sensitisers would not be beneficial in reducing anthracycline-induced cardiotoxicity.

The use of these positive inotropic agents are only recommended short term due to the potential proarrhythmic effects (Tisdale *et al.*, 1995). While these drugs may be beneficial in treating anthracycline-induced cardiotoxicity the ideal situation would be to prevent the cardiotoxicity from occurring.

7.5.2 Prevention treatment strategies for anthracycline-induced cardiotoxicity Currently, dexrazoxane (DEX) may be given clinically alongside anthracycline treatment in an attempt to reduce the chances of cardiotoxicity (Muthuramalingam *et al.*, 2013). DEX is an iron chelating agent and may reduce mitochondrial iron accumulation therefore preventing ROS production, in addition it may degrade topoisomerase II which is involved in another potential mechanism of action of anthracyclines (Ichikawa *et al.*, 2014a). As anthracyclines are known to increase ROS, it is believed DEX works to combat this production therefore reducing cardiotoxicity. As discussed in 1.6.5 this treatment strategy has been shown to reduce cardiac events due to anthracycline treatment but the long-term implications for children are debated.

Our findings show DOX increase OS in cardiac myocytes. Alternative treatment strategies may involve reducing the production of ROS in the heart. Inhibition of XO and NOX2/NOX4 attenuated many of the effects of DOX on Ca handling and contractility (see chapter 5).

However, DAUN did not appear to increase ROS in the same manner as DOX and so this strategy may not be useful for all anthracycline treatment (see chapter 4). XO inhibitors are used clinically to treat gout and have been shown to reduce the risk of cardiovascular events (Saito *et al.*, 2021), highlighting their potential usefulness for anthracycline-induced cardiotoxicity prevention.

A concern with implementing the use of ROS inhibitors with anthracycline use is reducing the effectivity of the drugs on cancer. However, ROS production is not the only mechanism of action of anthracyclines as discussed in 1.5.2 indicating that overall reduction in anthracycline-induced ROS may be beneficial. Furthermore, cardiac myocytes are abundant in mitochondria and have fewer antioxidant enzymes (Barth et al., 1992; Costa et al., 2013) making them particularly vulnerable to ROS. One study found the use of the generic antioxidant NAC did not reverse the effects of DOX-induced cardiomyopathy, (Dresdale et al., 1982) however this was given after the disease state had occurred which indicates ROS inhibitors would not be useful as a compensatory treatment. However, the use of ergothioneine which has antioxidant properties was found to have cardioprotective effects in mice treated with DOX and did not interfere with the treatment's efficacy (Cheah et al., 2023). This suggest that ROS inhibitors may be useful given alongside anthracycline treatment as a preventative measure. The use of antioxidants with anthracyclines is not well studied and their use to support anthracycline treatment is not well known (Vincent et al., 2013). Further investigation is required to ensure the efficacy of anthracycline treatment is not impacted by co-treatment with antioxidants and that cardioprotective benefits occur.

7.6 Limitations

The methodology used in this study allowed identification of the individual effects of DOX and DAUN on Ca handling and contractility in intact cardiomyocytes in a robust way. While the effects on various ECC components was investigated the use of patch clamping would have allowed further insight into the effects on L-type Ca channels and NCX activity as well as overcoming the caffeine interaction phenomenon with DAUN. The inhibition of OS resulted 182

in a decrease in SR Ca content due to DOX that could not be explained by the relative activity of the components investigated therefore studying activity of the L-type Ca channel may elucidate further mechanisms. Unfortunately, the equipment was not available during the course of these studies.

Furthermore, these studies are acute and while allowing an insight into the mechanistic effects of DOX and DAUN the implications of long-term treatment may differ. As shown in figure 1.15 the effect of anthracycline treatment on contractility recovers only to reduce again years later (Lipshultz *et al.*, 2005). Investigating the implications of treatment on ion channels over longer time periods may elucidate this mechanism. Primary cardiac myocytes are an invaluable resource but unfortunately are not viable long-term.

7.7 Future work

Valuable insight was obtained from these studies including the effect of DOX and DAUN on intracellular Ca and contractility, OS and the potential attenuating effects of ROS inhibitors. Future work will investigate the effects of DOX and DAUN on the other components of ECC such as LTCC using patch clamping. Furthermore, we will co-load cells with Ca and OSsensitive fluorophores to more directly correlate the changes of OS to those of ECC. This will support that the effects on Ca handling due to DOX are ROS induced due to the bidirectional nature of this relationship. It would also be advantageous to investigate mitochondrial ROS production for which specific fluorophores exist. The effect by which DAUN evokes cardiotoxicity requires further investigation as does the interaction with caffeine which may elucidate mechanistic effects. We have characterised the effect of DOX and DAUN on various aspects of ECC (e.g., SERCA and NCX), however we don't know how the anthracyclines interact with these proteins. While our OS work suggests this may be through ROS signalling the interaction of DAUN and caffeine may suggest there is direct binding occurring with the RyR. To investigate this a collaboration has been secured which will incorporate the use of solidstate NMR to examine the molecular dynamics between anthracyclines and cardiac calcium handling proteins.

In terms of establishing if the long-term effects of anthracyclines are induced in a similar manner to the acute, a cultured cell line will be utilised. As previously mentioned, primary cardiac myocytes are not suitable for long-term experiments due to the reduction in viability. The HL-1 cell line have similar characteristics to primary cardiac myocytes and allow for electrophysiological studies while having the ability to be kept in culture (White *et al.*, 2004). This means that cells can be treated with anthracyclines for a longer period and the effects on Ca handling explored.

7.8 Summary

1 nM DOX reduced the activity of SERCA and increased the activity of NCX resulting in a reduction in SR Ca content. 1 nM DAUN also reduced SR Ca, but apparent interactions of DAUN with caffeine meant the mechanisms could not be fully elucidated. The reduction in SR Ca content underpinned a reduction in Ca transient amplitude by both DOX and DAUN which resulted in a decrease in contractility. DOX and DAUN did not appear to alter diastolic Ca acutely. Both DOX and DAUN increased myofilament sensitivity to Ca which may have potentially offset the effects seen to contractility.

DOX increased cytoplasmic OS in a concentration and time-dependent manner. DAUN also increased OS at higher concentrations (10 μ M). This indicates potential differences in the mechanism of cardiac toxicity and may also account for DAUN being associated with reduced cardiotoxicity.

Removal of OS using NAC attenuated the majority of the effects of DOX on Ca handling and contractility however SR Ca content was still reduced but attenuated by ~45 %. This suggest that ~50 % of the effects of DOX on heart cells are OS-dependent. Further investigation showed NOX plays a vital role in ROS production associated with the increase in OS and that XO has a partial role. Inhibition of both enzymes attenuated various effects of DOX on Ca handling and contractility with the exception of Ca transient amplitude.

In cells from heart failure DOX and DAUN produced similar effects however the impact on Ca transient amplitude was not reversible upon wash. Furthermore, the Ca transient amplitude appeared to be reduced to a lesser extent in cells from a HF model compared to cells from the healthy model. While the SR Ca content was reduced by DOX in cells from HF the activity of NCX and SERCA was not altered (unlike in the healthy model), this suggests the effects on NCX and SERCA are saturated in the HF model and other channels such as L-type Ca channels may have a role in the reduction of SR Ca content.

This study considerably advances our understanding of the cellular basis of HF that is associated with the use of anthracyclines for cancer treatment. We are the first to show that elevations in OS contribute to DOX (and to potentially DAUN) mediated changes to ECC. NOX and XO were revealed as key sources of OS and may present as novel therapeutic targets.

Chapter 8 Bibliography

- Abramson, J. J., Buck, E., Salama, G., Casida, J. E., & Pessah, I. N. (1988). Mechanism of anthraquinone-induced calcium release from skeletal muscle sarcoplasmic reticulum. *J Biol Chem*, 263(35), 18750-18758.
- Akin, B. L., Hurley, T. D., Chen, Z., & Jones, L. R. (2013). The structural basis for phospholamban inhibition of the calcium pump in sarcoplasmic reticulum. *The Journal of biological chemistry, 288*(42), 30181-30191. doi:10.1074/jbc.M113.501585
- Akki, A., Zhang, M., Murdoch, C., Brewer, A., & Shah, A. M. (2009). NADPH oxidase signaling and cardiac myocyte function. *Journal of Molecular and Cellular Cardiology*, 47(1), 15-22. doi:<u>https://doi.org/10.1016/j.yjmcc.2009.04.004</u>
- Al-Aamri, H. M., Ku, H., Irving, H. R., Tucci, J., Meehan-Andrews, T., & Bradley, C. (2019). Time dependent response of daunorubicin on cytotoxicity, cell cycle and DNA repair in acute lymphoblastic leukaemia. *BMC Cancer, 19*(1), 179. doi:10.1186/s12885-019-5377-y
- Altenhöfer, S., Radermacher, K. A., Kleikers, P. W. M., Wingler, K., & Schmidt, H. H. H. W. (2014). Evolution of NADPH Oxidase Inhibitors: Selectivity and Mechanisms for Target Engagement. *Antioxidants & Redox Signaling, 23*(5), 406-427. doi:10.1089/ars.2013.5814
- Altomare, C., Lodrini, A. M., Milano, G., Biemmi, V., Lazzarini, E., Bolis, S., . . . Vassalli, G. (2021). Structural and Electrophysiological Changes in a Model of Cardiotoxicity Induced by Anthracycline Combined With Trastuzumab. *Frontiers in physiology, 12*, 658790. doi:10.3389/fphys.2021.658790
- Alves, A. C., Nunes, C., Lima, J., & Reis, S. (2017). Daunorubicin and doxorubicin molecular interplay with 2D membrane models. *Colloids and Surfaces B: Biointerfaces, 160*, 610-618. doi:<u>https://doi.org/10.1016/j.colsurfb.2017.09.058</u>
- Ana María, G.-M., Hilda, V.-R., Amelia, R., & Bruno, E. (2015). Oxidative Stress-Dependent Coronary Endothelial Dysfunction in Obese Mice. *PLOS ONE, 10*(9), e0138609. doi:10.1371/journal.pone.0138609
- Anzai, K., Ogawa, K., Kuniyasu, A., Ozawa, T., Yamamoto, H., & Nakayama, H. (1998). Effects of hydroxyl radical and sulfhydryl reagents on the open probability of the purified 187

cardiac ryanodine receptor channel incorporated into planar lipid bilayers. *Biochem Biophys Res Commun, 249*(3), 938-942. doi:10.1006/bbrc.1998.9244

- Arai, M., Tomaru, K., Takizawa, T., Sekiguchi, K., Yokoyama, T., Suzuki, T., & Nagai, R. (1998).
 Sarcoplasmic Reticulum Genes are Selectively Down-regulated in Cardiomyopathy
 Produced by Doxorubicin in Rabbits. *Journal of Molecular and Cellular Cardiology,* 30(2), 243-254. doi:https://doi.org/10.1006/jmcc.1997.0588
- Arcamone, F., Franceschi, G., Penco, S., & Selva, A. (1969). Adriamycin (14hydroxydaunomycin), a novel antitumor antibiotic. *Tetrahedron Letters*, 10(13), 1007-1010. doi:<u>https://doi.org/10.1016/S0040-4039(01)97723-8</u>
- Armenian, S., & Bhatia, S. (2018). Predicting and Preventing Anthracycline-Related Cardiotoxicity. American Society of Clinical Oncology Educational Book(38), 3-12. doi:10.1200/edbk_100015
- Armenian, S. H., Hudson, M. M., Mulder, R. L., Chen, M. H., Constine, L. S., Dwyer, M., . . . Kremer, L. C. M. (2015). Recommendations for cardiomyopathy surveillance for survivors of childhood cancer: a report from the International Late Effects of Childhood Cancer Guideline Harmonization Group. *The Lancet Oncology*, *16*(3), e123e136. doi:10.1016/S1470-2045(14)70409-7
- Armenian, S. H., Lacchetti, C., Barac, A., Carver, J., Constine, L. S., Denduluri, N., . . . Lenihan,
 D. (2017). Prevention and Monitoring of Cardiac Dysfunction in Survivors of Adult
 Cancers: American Society of Clinical Oncology Clinical Practice Guideline. *Journal of Clinical Oncology*, 35(8), 893-911. doi:10.1200/jco.2016.70.5400
- Asensio-López, M. C., Soler, F., Sánchez-Más, J., Pascual-Figal, D., Fernández-Belda, F., & Lax,
 A. (2016). Early oxidative damage induced by doxorubicin: Source of production,
 protection by GKT137831 and effect on Ca2+ transporters in HL-1 cardiomyocytes.
 Archives of Biochemistry and Biophysics, 594, 26-36.
 doi:<u>https://doi.org/10.1016/j.abb.2016.02.021</u>
- Attanasio, U., Pirozzi, F., Poto, R., Cuomo, A., Carannante, A., Russo, M., . . . Mercurio, V. (2021). Oxidative stress in anticancer therapies-related cardiac dysfunction. *Free*

RadicalBiologyandMedicine,169,410-415.doi:https://doi.org/10.1016/j.freeradbiomed.2021.04.021

- Baartscheer, A., Schumacher, C. A., Belterman, C. N. W., Coronel, R., & Fiolet, J. W. T. (2003a).
 SR calcium handling and calcium after-transients in a rabbit model of heart failure.
 Cardiovascular Research, 58(1), 99-108. doi:10.1016/s0008-6363(02)00854-4
- Baartscheer, A., Schumacher, C. A., van Borren, M. M., Belterman, C. N., Coronel, R., & Fiolet,
 J. W. (2003b). Increased Na+/H+-exchange activity is the cause of increased [Na+]i and
 underlies disturbed calcium handling in the rabbit pressure and volume overload heart
 failure model. *Cardiovasc Res*, 57(4), 1015-1024. doi:10.1016/s0008-6363(02)00809-x
- Baddeley, D., Jayasinghe, I. D., Lam, L., Rossberger, S., Cannell, M. B., & Soeller, C. (2009).
 Optical single-channel resolution imaging of the ryanodine receptor distribution in rat cardiac myocytes. *Proceedings of the National Academy of Sciences of the United States of America*, 106(52), 22275-22280. doi:10.1073/pnas.0908971106
- Barpe, D. R., Rosa, D. D., & Froehlich, P. E. (2010). Pharmacokinetic evaluation of doxorubicin plasma levels in normal and overweight patients with breast cancer and simulation of dose adjustment by different indexes of body mass. *European Journal of Pharmaceutical Sciences*, 41(3), 458-463. doi:<u>https://doi.org/10.1016/j.ejps.2010.07.015</u>
- Barth, E., Stämmler, G., Speiser, B., & Schaper, J. (1992). Ultrastructural quantitation of mitochondria and myofilaments in cardiac muscle from 10 different animal species including man. *Journal of Molecular and Cellular Cardiology*, 24(7), 669-681. doi:https://doi.org/10.1016/0022-2828(92)93381-S
- Bartz, R. R., Suliman, H. B., & Piantadosi, C. A. (2015). Redox mechanisms of cardiomyocyte mitochondrial protection. *Frontiers in physiology, 6*. doi:10.3389/fphys.2015.00291
- Baruscotti, M., & Robinson, R. B. (2007). Electrophysiology and pacemaker function of the developing sinoatrial node. Am J Physiol Heart Circ Physiol, 293(5), H2613-2623. doi:10.1152/ajpheart.00750.2007

- Bassani, R. A., Bassani, J. W., & Bers, D. M. (1992). Mitochondrial and sarcolemmal Ca2+ transport reduce [Ca2+]i during caffeine contractures in rabbit cardiac myocytes. *The Journal of physiology*, 453, 591-608. doi:10.1113/jphysiol.1992.sp019246
- Bates, D. A., & Winterbourn, C. C. (1982). Deoxyribose breakdown by the adriamycin semiquinone and H2O2: evidence for hydroxyl radical participation. *FEBS Lett*, 145(1), 137-142. doi:10.1016/0014-5793(82)81222-2
- Bayir, H. (2005). Reactive oxygen species. *Crit Care Med*, *33*(12 Suppl), S498-501. doi:10.1097/01.ccm.0000186787.64500.12
- Beavers, D. L., Landstrom, A. P., Chiang, D. Y., & Wehrens, X. H. T. (2014). Emerging roles of junctophilin-2 in the heart and implications for cardiac diseases. *Cardiovascular Research*, 103(2), 198-205. doi:10.1093/cvr/cvu151
- Berry, C. E., & Hare, J. M. (2004a). Xanthine oxidoreductase and cardiovascular disease: molecular mechanisms and pathophysiological implications. *The Journal of physiology*, 555(3), 589-606. doi:<u>https://doi.org/10.1113/jphysiol.2003.055913</u>
- Berry, C. E., & Hare, J. M. (2004b). Xanthine oxidoreductase and cardiovascular disease: molecular mechanisms and pathophysiological implications. *The Journal of physiology*, 555(Pt 3), 589-606. doi:10.1113/jphysiol.2003.055913
- Bers, D. (1997). Ca transport during contraction and relaxation in mammalian ventricular muscle. *Basic Research in Cardiology*, *92*(1), 1-10. doi:10.1007/BF00794062
- Bers, D. M. (2001a). *Excitation-contraction coupling and cardiac contractile force* (2nd edition. ed.): Dordrecht, The Netherlands

London : Kluwer Academic Publishers.

- Bers, D. M. (2001b). *Excitation-Contraction Coupling and Cardiac Contractile Force (2nd ed.)*. Dordrecht: Kluwer Academic Publishers.
- Bers, D. M. (2002). Cardiac excitation-contraction coupling. *Nature, 415*(6868), 198-205. doi:<u>http://dx.doi.org/10.1038/415198a</u>

- Bers, D. M., Despa, S., & Bossuyt, J. (2006). Regulation of Ca2+ and Na+ in Normal and Failing Cardiac Myocytes. Annals of the New York Academy of Sciences, 1080(1), 165-177. doi:https://doi.org/10.1196/annals.1380.015
- Bers, D. M., Eisner, D. A., & Valdivia, H. H. (2003). Sarcoplasmic Reticulum Ca²⁺
 and Heart Failure. *Circulation Research, 93*(6), 487-490.
 doi:doi:10.1161/01.RES.0000091871.54907.6B
- Bers, D. M., & Grandi, E. (2009). Calcium/calmodulin-dependent kinase II regulation of cardiac
 ion channels. J Cardiovasc Pharmacol, 54(3), 180-187.
 doi:10.1097/FJC.0b013e3181a25078
- Bertero, E., Canepa, M., Maack, C., & Ameri, P. (2018). Linking Heart Failure to Cancer. *Circulation, 138*(7), 735-742. doi:doi:10.1161/CIRCULATIONAHA.118.033603
- Biesiadecki, B. J., Davis, J. P., Ziolo, M. T., & Janssen, P. M. L. (2014). Tri-modal regulation of cardiac muscle relaxation; intracellular calcium decline, thin filament deactivation, and cross-bridge cycling kinetics. *Biophysical Reviews, 6*(3), 273-289. doi:10.1007/s12551-014-0143-5
- Bode, E. F., Briston, S. J., Overend, C. L., O'Neill, S. C., Trafford, A. W., & Eisner, D. A. (2011).
 Changes of SERCA activity have only modest effects on sarcoplasmic reticulum Ca2+ content in rat ventricular myocytes. *The Journal of physiology, 589*(19), 4723-4729.
 doi:<u>https://doi.org/10.1113/jphysiol.2011.211052</u>
- Bonanni, A., Campanella, L., Gatta, T., Gregori, E., & Tomassetti, M. (2007). Evaluation of the antioxidant and prooxidant properties of several commercial dry spices by different analytical methods. *Food Chemistry*, 102(3), 751-758. doi:<u>https://doi.org/10.1016/j.foodchem.2006.06.030</u>
- Boraso, A., & Williams, A. J. (1994). Modification of the gating of the cardiac sarcoplasmic reticulum Ca(2+)-release channel by H2O2 and dithiothreitol. *Am J Physiol, 267*(3 Pt 2), H1010-1016. doi:10.1152/ajpheart.1994.267.3.H1010
- Borchi, E., Bargelli, V., Stillitano, F., Giordano, C., Sebastiani, M., Nassi, P. A., . . . Nediani, C. (2010). Enhanced ROS production by NADPH oxidase is correlated to changes in antioxidant enzyme activity in human heart failure. *Biochimica et Biophysica Acta* 191

(BBA) - Molecular Basis of Disease, 1802(3), 331-338. doi:https://doi.org/10.1016/j.bbadis.2009.10.014

- Boucek, R. J., Jr., Buck, S. H., Scott, F., Oquist, N. L., Fleischer, S., Olson, R. D., & Okediji, E. (1993). Anthracycline-induced Tension in Permeabilized Cardiac Fibers: Evidence for the Activation of the Calcium Release Channel of Sarcoplasmic Reticulum. *Journal of Molecular and Cellular Cardiology*, 25(3), 249-259. doi:10.1006/jmcc.1993.1032
- Brandes, R. P., Weissmann, N., & Schröder, K. (2014). Nox family NADPH oxidases: Molecular mechanisms of activation. *Free Radical Biology and Medicine*, *76*, 208-226. doi:https://doi.org/10.1016/j.freeradbiomed.2014.07.046
- Brini, M., Calì, T., Ottolini, D., & Carafoli, E. (2013). The plasma membrane calcium pump in health and disease. *The FEBS Journal, 280*(21), 5385-5397. doi:10.1111/febs.12193
- Briston, S. J., Caldwell, J. L., Horn, M. A., Clarke, J. D., Richards, M. A., Greensmith, D. J., ...
 Trafford, A. W. (2011). Impaired β-adrenergic responsiveness accentuates dysfunctional excitation-contraction coupling in an ovine model of tachypacing-induced heart failure. *The Journal of physiology*, *589*(Pt 6), 1367-1382. doi:10.1113/jphysiol.2010.203984
- Brown, P., Inaba, H., Annesley, C., Beck, J., Colace, S., Dallas, M., . . . Ogba, N. (2020). Pediatric Acute Lymphoblastic Leukemia, Version 2.2020, NCCN Clinical Practice Guidelines in Oncology. *Journal of the National Comprehensive Cancer Network J Natl Compr Canc Netw, 18*(1), 81-112. doi:10.6004/jnccn.2020.0001
- Bruce, J. I. E. (2018). Metabolic regulation of the PMCA: Role in cell death and survival. *Cell Calcium, 69*, 28-36. doi:<u>https://doi.org/10.1016/j.ceca.2017.06.001</u>
- Bryant, S., Kimura, T. E., Kong, C. H. T., Watson, J. J., Chase, A., Suleiman, M. S., . . . Orchard,
 C. H. (2014). Stimulation of ICa by basal PKA activity is facilitated by caveolin-3 in cardiac ventricular myocytes. *Journal of Molecular and Cellular Cardiology, 68*, 47-55. doi:https://doi.org/10.1016/j.yjmcc.2013.12.026
- Burgoyne, J. R., Mongue-Din, H., Eaton, P., & Shah, A. M. (2012). Redox Signaling in Cardiac
 Physiology and Pathology. *Circulation Research*, 111(8), 1091-1106.
 doi:10.1161/CIRCRESAHA.111.255216

- Burridge, P. W., Li, Y. F., Matsa, E., Wu, H., Ong, S.-G., Sharma, A., . . . Wu, J. C. (2016). Human induced pluripotent stem cell-derived cardiomyocytes recapitulate the predilection of breast cancer patients to doxorubicin-induced cardiotoxicity. *Nature medicine*, 22(5), 547-556. doi:10.1038/nm.4087
- Burton, G. J., & Jauniaux, E. (2011). Oxidative stress. *Best Practice & Research Clinical Obstetrics & Gynaecology, 25*(3), 287-299. doi:https://doi.org/10.1016/j.bpobgyn.2010.10.016
- Capranico, G., Kohn, K. W., Pommier, Y., & Zunino, F. (1990). Sequence-selective topoisomerase II inhibition by anthracycline derivatives in SV40 DNA: Relationship with DNA binding affinity and cytotoxicity. *Biochemistry, 29*(2). doi:10.1021/bi00454a033
- Carafoli, E. (1994). Biogenesis: Plasma membrane calcium ATPase: 15 years of work on the purified enzyme1. *The FASEB Journal, 8*(13), 993-1002. doi:10.1096/fasebj.8.13.7926378
- Catterall, W. A. (2018). 1 Voltage-Gated Sodium Channels and Electrical Excitability of the Heart. In D. P. Zipes, J. Jalife, & W. G. Stevenson (Eds.), *Cardiac Electrophysiology: From Cell to Bedside (Seventh Edition)* (pp. 1-11): Elsevier.
- Chakouri, N., Farah, C., Matecki, S., Amedro, P., Vincenti, M., Saumet, L., ... Cazorla, O. (2020).
 Screening for in-vivo regional contractile defaults to predict the delayed Doxorubicin
 Cardiotoxicity in Juvenile Rat. *Theranostics, 10*(18), 8130-8142.
 doi:10.7150/thno.47407
- Chance, B., Sies, H., & Boveris, A. (1979). Hydroperoxide metabolism in mammalian organs. *Physiol Rev, 59*(3), 527-605. doi:10.1152/physrev.1979.59.3.527
- Chase, A., Colyer, J., & Orchard, C. H. (2010). Localised Ca channel phosphorylation modulates the distribution of L-type Ca current in cardiac myocytes. *Journal of Molecular and Cellular Cardiology, 49*(1), 121-131. doi:<u>https://doi.org/10.1016/j.yjmcc.2010.02.017</u>
- Chaudhari, U., Nemade, H., Wagh, V., Gaspar, J. A., Ellis, J. K., Srinivasan, S. P., . . . Sachinidis,
 A. (2016). Identification of genomic biomarkers for anthracycline-induced cardiotoxicity in human iPSC-derived cardiomyocytes: an in vitro repeated exposure 193

toxicity approach for safety assessment. *Archives of Toxicology, 90*(11), 2763-2777. doi:10.1007/s00204-015-1623-5

- Chazin, W. J., & Johnson, C. N. (2020). Calmodulin Mutations Associated with Heart Arrhythmia: A Status Report. *International journal of molecular sciences, 21*(4). doi:10.3390/ijms21041418
- Cheah, I. K., Tang, R. M. Y., Wang, X., Sachaphibulkij, K., Chong, S. Y., Lim, L. H. K., . . . Halliwell, B. (2023). Protection against Doxorubicin-Induced Cardiotoxicity by Ergothioneine. *Antioxidants*, 12(2), 320. Retrieved from https://www.mdpi.com/2076-3921/12/2/320
- Chen, E., Chen, B.-M., Su, Y.-C., Chang, Y.-C., Cheng, T.-L., Barenholz, Y., & Roffler, S. R. (2020).
 Premature Drug Release from Polyethylene Glycol (PEG)-Coated Liposomal Doxorubicin via Formation of the Membrane Attack Complex. ACS Nano, 14(7), 7808-7822. doi:10.1021/acsnano.9b07218
- Chen, F., Hadfield, J. M., Berzingi, C., Hollander, J. M., Miller, D. B., Nichols, C. E., & Finkel, M.
 S. (2013). N-acetylcysteine reverses cardiac myocyte dysfunction in a rodent model of behavioral stress. *Journal of Applied Physiology*, 115(4), 514-524. doi:10.1152/japplphysiol.01471.2012
- Chen, F., Haigh, S., Barman, S., & Fulton, D. (2012a). From form to function: the role of Nox4 in the cardiovascular system. *Frontiers in physiology, 3*. doi:10.3389/fphys.2012.00412
- Chen, F., Lewis, W., Hollander, J. M., Baseler, W., & Finkel, M. S. (2012b). N-acetylcysteine reverses cardiac myocyte dysfunction in HIV-Tat proteinopathy. *Journal of Applied Physiology*, 113(1), 105-113. doi:10.1152/japplphysiol.00068.2012
- Chen, M.-m., & Meng, L.-h. (2022). The double faced role of xanthine oxidoreductase in cancer. *Acta Pharmacologica Sinica*, *43*(7), 1623-1632. doi:10.1038/s41401-021-00800-7
- Choi, H. S., Park, E. S., Kang, H. J., Shin, H. Y., Noh, C. I., Yun, Y. S., . . . Choi, J. Y. (2010). Dexrazoxane for Preventing Anthracycline Cardiotoxicity in Children with Solid

Tumors. *Journal of Korean medical science,* 25(9), 1336-1342. doi:10.3346/jkms.2010.25.9.1336

- Clarke, D. M., Maruyama, K., Loo, T. W., Leberer, E., Inesi, G., & Maclennan, D. H. (1989). Functional consequences of glutamate, aspartate, glutamine, and asparagine mutations in the stalk sector of the Ca2+-ATPase of sarcoplasmic reticulum. *The Journal of biological chemistry*, 264(19), 11246. Retrieved from <u>https://www.jbc.org/content/264/19/11246.full.pdf</u>
- Clarke, J. D., Caldwell, J. L., Horn, M. A., Bode, E. F., Richards, M. A., Hall, M. C. S., . . . Trafford,
 A. W. (2015). Perturbed atrial calcium handling in an ovine model of heart failure:
 Potential roles for reductions in the L-type calcium current. *Journal of Molecular and Cellular Cardiology, 79*, 169-179. doi:10.1016/j.yjmcc.2014.11.017
- Cooper, P. J., Soeller, C., & Cannell, M. B. (2010). Excitation–contraction coupling in human heart failure examined by action potential clamp in rat cardiac myocytes.
 Journal of Molecular and Cellular Cardiology, 49(6), 911-917. doi:10.1016/j.yjmcc.2010.04.012
- Costa, V. M., Carvalho, F., Duarte, J. A., Bastos Mde, L., & Remião, F. (2013). The heart as a target for xenobiotic toxicity: the cardiac susceptibility to oxidative stress. *Chem Res Toxicol, 26*(9), 1285-1311. doi:10.1021/tx400130v
- Costantini, D., & Verhulst, S. (2009). Does high antioxidant capacity indicate low oxidative stress? *Functional Ecology*, *23*(3), 506-509. doi:10.1111/j.1365-2435.2009.01546.x
- Curran, J., Hinton, M. J., Ríos, E., Bers, D. M., & Shannon, T. R. (2007). Adrenergic Enhancement of Sarcoplasmic Reticulum Calcium Leak in Cardiac Myocytes Is Mediated by Calcium/Calmodulin-Dependent Protein Kinase. *Circulation Research*, 100(3), 391-398. doi:doi:10.1161/01.RES.0000258172.74570.e6
- Cusack, B. J., Mushlin, P. S., Voulelis, L. D., Li, X. D., Boucek, R. J., & Olson, R. D. (1993). Daunorubicin-Induced Cardiac Injury in the Rabbit: A Role for Daunorubicinol? *Toxicology and Applied Pharmacology, 118*(2), 177-185. doi:<u>https://doi.org/10.1006/taap.1993.1023</u>

- D'Oria, R., Schipani, R., Leonardini, A., Natalicchio, A., Perrini, S., Cignarelli, A., . . . Giorgino, F. (2020). The Role of Oxidative Stress in Cardiac Disease: From Physiological Response to Injury Factor. *Oxidative Medicine and Cellular Longevity, 2020*, 5732956. doi:10.1155/2020/5732956
- Davies, M. J. (2000). The cardiomyopathies: an overview. *Heart, 83*(4), 469. doi:10.1136/heart.83.4.469
- De Falco, E., Carnevale, R., Pagano, F., Chimenti, I., Fianchini, L., Bordin, A., . . . Frati, G. (2016).
 Role of NOX2 in mediating doxorubicin-induced senescence in human endothelial progenitor cells. *Mechanisms of Ageing and Development, 159*, 37-43. doi:https://doi.org/10.1016/j.mad.2016.05.004
- de Tombe, P. P. (2003). Cardiac myofilaments: mechanics and regulation. *Journal of Biomechanics*, *36*(5), 721-730. doi:<u>https://doi.org/10.1016/S0021-9290(02)00450-5</u>
- Demers, M., & Wagner, D. D. (2014). NETosis: a new factor in tumor progression and cancerassociated thrombosis. *Seminars in thrombosis and hemostasis, 40*(3), 277-283. doi:10.1055/s-0034-1370765
- Deng, S., Yan, T., Jendrny, C., Nemecek, A., Vincetic, M., Gödtel-Armbrust, U., & Wojnowski,
 L. (2014). Dexrazoxane may prevent doxorubicin-induced DNA damage via depleting both Topoisomerase II isoforms. *BMC Cancer*, *14*(1), 842-842. doi:10.1186/1471-2407-14-842
- Denvir, M. A., MacFarlane, N. G., Cobbe, S. M., & Miller, D. J. (1995). Sarcoplasmic reticulum and myofilament function in chemically-treated ventricular trabeculae from patients with heart failure. *Cardiovasc Res, 30*(3), 377-385.
- Despa, S., Islam, M. A., Weber, C. R., Pogwizd, S. M., & Bers, D. M. (2002). Intracellular Na⁺ Concentration Is Elevated in Heart Failure But Na/K Pump Function Is Unchanged. *Circulation, 105*(21), 2543-2548. doi:doi:10.1161/01.CIR.0000016701.85760.97
- Dibb, K., Rueckschloss, U., Eisner, D., Isenberg, G., & Trafford, A. (2004). Mechanisms underlying enhanced cardiac excitation contraction coupling observed in the
senescent sheep myocardium. *Journal of Molecular and Cellular Cardiology, 37*(6), 1171-1181.

- Dibb, K. M., Graham, H. K., Venetucci, L. A., Eisner, D. A., & Trafford, A. W. (2007). Analysis of cellular calcium fluxes in cardiac muscle to understand calcium homeostasis in the heart. *Cell Calcium*, 42(4), 503-512. doi:<u>https://doi.org/10.1016/j.ceca.2007.04.002</u>
- Dibb, K. M., Clarke, J. D., Horn, M. A., Richards, M. A., Graham, H. K., Eisner, D. A., & Trafford,
 A. W. (2009). Characterization of an Extensive Transverse Tubular Network in Sheep
 Atrial Myocytes and its Depletion in Heart Failure. *Circulation: Heart Failure, 2*(5), 482-489. doi:doi:10.1161/CIRCHEARTFAILURE.109.852228
- Diez, C., & Simm, A. (1998). Gene expression in rod shaped cardiac myocytes, sorted by flow cytometry. *Cardiovasc Res, 40*(3), 530-537. doi:10.1016/s0008-6363(98)00189-8
- Dolara, P., Agresti, A., Giotti, A., & Pasquini, G. (1973). Effect of taurine on calcium kinetics of guinea-pig heart. *European Journal of Pharmacology, 24*(3), 352-358. doi:<u>https://doi.org/10.1016/0014-2999(73)90162-3</u>
- Dommett, R. M., Redaniel, T., Stevens, M. C., Martin, R. M., & Hamilton, W. (2013). Risk of childhood cancer with symptoms in primary care: a population-based case-control study. *Br J Gen Pract, 63*(606), e22-29. doi:10.3399/bjgp13X660742
- Doroshow, J. H. (2019). Mechanisms of Anthracycline-Enhanced Reactive Oxygen Metabolism in Tumor Cells. *Oxid Med Cell Longev, 2019*, 9474823. doi:10.1155/2019/9474823
- Dresdale, A. R., Barr, L. H., Bonow, R. O., Mathisen, D. J., Myers, C. E., Schwartz, D. E., . . .
 Rosenberg, S. A. (1982). Prospective randomized study of the role of N-acetyl cysteine in reversing doxorubicin-induced cardiomyopathy. *Am J Clin Oncol, 5*(6), 657-663. doi:10.1097/00000421-198212000-00015
- Dubois-Deruy, E., Peugnet, V., Turkieh, A., & Pinet, F. (2020). Oxidative Stress in Cardiovascular Diseases. *Antioxidants (Basel), 9*(9). doi:10.3390/antiox9090864
- Dulhunty, A. F., Beard, N. A., & Hanna, A. D. (2012). Regulation and dysregulation of cardiac ryanodine receptor (RyR2) open probability during diastole in health and disease. *Journal of General Physiology, 140*(2), 87-92. doi:10.1085/jgp.201210862

- Eigel, B. N., Gursahani, H., & Hadley, R. W. (2004). ROS are required for rapid reactivation of Na+/Ca2+ exchanger in hypoxic reoxygenated guinea pig ventricular myocytes. *American Journal of Physiology-Heart and Circulatory Physiology, 286*(3), H955-H963. doi:10.1152/ajpheart.00721.2003
- Eisner, D. A., Choi, H. S., Díaz, M. E., O'Neill, S. C., & Trafford, A. W. (2000). Integrative Analysis of Calcium Cycling in Cardiac Muscle. *Circulation Research*, *87*(12), 1087-1094. doi:doi:10.1161/01.RES.87.12.1087
- Eisner, D. (2014). Calcium in the heart: from physiology to disease. *Experimental Physiology*, 99(10), 1273-1282. doi:<u>https://doi.org/10.1113/expphysiol.2013.077305</u>
- Eisner, D. A., Caldwell, J. L., Kistamás, K., & Trafford, A. W. (2017). Calcium and Excitation-Contraction Coupling in the Heart. *Circulation Research*, 121(2), 181-195. doi:doi:10.1161/CIRCRESAHA.117.310230
- Eisner, D. A. (2018). Ups and downs of calcium in the heart. *The Journal of physiology, 596*(1), 19-30. doi:10.1113/JP275130
- Ekelund, U. E., Harrison, R. W., Shokek, O., Thakkar, R. N., Tunin, R. S., Senzaki, H., . . . Hare, J. M. (1999). Intravenous allopurinol decreases myocardial oxygen consumption and increases mechanical efficiency in dogs with pacing-induced heart failure. *Circ Res,* 85(5), 437-445. doi:10.1161/01.res.85.5.437
- El Dor, M., Dakik, H., Polomski, M., Haudebourg, E., Brachet, M., Gouilleux, F., . . . Mazurier,
 F. (2020). VAS3947 Induces UPR-Mediated Apoptosis through Cysteine Thiol
 Alkylation in AML Cell Lines. *International journal of molecular sciences, 21*(15).
 doi:10.3390/ijms21155470
- Fanous, I., & Dillon, P. (2016). Cancer treatment-related cardiac toxicity: prevention, assessment and management. *Medical Oncology*, 33(8), 1-11. doi:10.1007/s12032-016-0801-5
- Fedorov, V. V., Glukhov, A. V., & Chang, R. (2012). Conduction barriers and pathways of the sinoatrial pacemaker complex: their role in normal rhythm and atrial arrhythmias. *American Journal of Physiology-Heart and Circulatory Physiology, 302*(9), H1773-H1783. doi:10.1152/ajpheart.00892.2011

- Feijen, E. A. M., Leisenring, W. M., Stratton, K. L., Ness, K. K., Pal, H. J. H. v. d., Caron, H. N., . .
 . Chow, E. J. (2015a). Equivalence Ratio for Daunorubicin to Doxorubicin in Relation to Late Heart Failure in Survivors of Childhood Cancer. *Journal of Clinical Oncology*, 33(32), 3774-3780. doi:10.1200/jco.2015.61.5187
- Feijen, E. A. M., Leisenring, W. M., Stratton, K. L., Ness, K. K., van der Pal, H. J. H., Caron, H. N.,
 ... Chow, E. J. (2015b). Equivalence Ratio for Daunorubicin to Doxorubicin in Relation to Late Heart Failure in Survivors of Childhood Cancer. *Journal of Clinical Oncology,* 33(32), 3774-3780. doi:10.1200/JCO.2015.61.5187
- Feijen, E. A. M., Leisenring, W. M., Stratton, K. L., Ness, K. K., van der Pal, H. J. H., van Dalen,
 E. C., . . . Chow, E. J. (2019). Derivation of Anthracycline and Anthraquinone
 Equivalence Ratios to Doxorubicin for Late-Onset Cardiotoxicity. *JAMA Oncol, 5*(6), 864-871. doi:10.1001/jamaoncol.2018.6634
- Feneck, R. (2007). Phosphodiesterase inhibitors and the cardiovascular system. *Continuing Education in Anaesthesia Critical Care & Pain, 7*(6), 203-207. doi:10.1093/bjaceaccp/mkm039
- Findlay, I. (2002). Voltage- and cation-dependent inactivation of L-type Ca2+ channel currents in guinea-pig ventricular myocytes. *The Journal of physiology*, 541(3), 731-740. doi:10.1113/jphysiol.2002.019729
- Finke, D., Heckmann, M. B., Frey, N., & Lehmann, L. H. (2021). Cancer-A Major Cardiac Comorbidity With Implications on Cardiovascular Metabolism. *Frontiers in physiology*, 12, 729713. doi:10.3389/fphys.2021.729713
- Finkel, T. (2011). Signal transduction by reactive oxygen species. *Journal of Cell Biology,* 194(1), 7-15. doi:10.1083/jcb.201102095
- Frampton, J. E. (2015). Febuxostat: A Review of Its Use in the Treatment of Hyperuricaemia in Patients with Gout. *Drugs*, *75*(4), 427-438. doi:10.1007/s40265-015-0360-7
- Franzini-Armstrong, C., Protasi, F., & Ramesh, V. (1999). Shape, Size, and Distribution of Ca2+ Release Units and Couplons in Skeletal and Cardiac Muscles. *Biophysical Journal*, 77(3), 1528-1539. doi:<u>https://doi.org/10.1016/S0006-3495(99)77000-1</u>

- Galán-Arriola, C., Vílchez-Tschischke, J. P., Lobo, M., López, G. J., de Molina-Iracheta, A., Pérez-Martínez, C., . . . Ibanez, B. (2021). Coronary microcirculation damage in anthracycline cardiotoxicity. *Cardiovascular Research*, *118*(2), 531-541. doi:10.1093/cvr/cvab053
- Gambliel, H. A., Burke, B. E., Cusack, B. J., Walsh, G. M., Zhang, Y. L., Mushlin, P. S., & Olson,
 R. D. (2002). Doxorubicin and C-13 Deoxydoxorubicin Effects on Ryanodine Receptor
 Gene Expression. *Biochemical and Biophysical Research Communications, 291*(3), 433438. doi:<u>https://doi.org/10.1006/bbrc.2002.6380</u>
- Gammella, E., Maccarinelli, F., Buratti, P., Recalcati, S., & Cairo, G. (2014). The role of iron in anthracycline cardiotoxicity. *Frontiers in Pharmacology*, 5. doi:10.3389/fphar.2014.00025
- García, J. G., Ansorena, E., Izal, I., Zalba, G., de Miguel, C., & Milagro, F. I. (2023). Structure, regulation, and physiological functions of NADPH oxidase 5 (NOX5). *Journal of Physiology and Biochemistry*. doi:10.1007/s13105-023-00955-3
- Giantris, A., Abdurrahman, L., Hinkle, A., Asselin, B., & Lipshultz, S. E. (1998). Anthracyclineinduced cardiotoxicity in children and young adults. *Critical Reviews in Oncology/Hematology,* 27(1), 53-68. doi:<u>https://doi.org/10.1016/S1040-8428(97)10007-5</u>
- Giordano, F. J. (2005). Oxygen, oxidative stress, hypoxia, and heart failure. *The Journal of Clinical Investigation*, *115*(3), 500-508. doi:10.1172/JCI24408
- Goldhaber, J. I., & Liu, E. (1994). Excitation-contraction coupling in single guinea-pig ventricular myocytes exposed to hydrogen peroxide. *The Journal of physiology*, 477(1), 135-147. doi:<u>https://doi.org/10.1113/jphysiol.1994.sp020178</u>
- Goldhaber, J. I. (1996). Free radicals enhance Na+/Ca2+ exchange in ventricular myocytes. *Am J Physiol, 271*(3 Pt 2), H823-833. doi:10.1152/ajpheart.1996.271.3.H823
- Gomes, A., Fernandes, E., & Lima, J. L. F. C. (2005). Fluorescence probes used for detection of reactive oxygen species. *Journal of Biochemical and Biophysical Methods, 65*(2-3), 45-80. doi:10.1016/j.jbbm.2005.10.003
- Goodman, J., & Hochstein, P. (1977). Generation of free radicals and lipid peroxidation by redox cycling of adriamycin and daunomycin. *Biochemical and Biophysical Research* 200

Communications, 77(2), 797-803. doi:<u>https://doi.org/10.1016/S0006-291X(77)80048-</u>

- Görlach, A., Bertram, K., Hudecova, S., & Krizanova, O. (2015). Calcium and ROS: A mutual interplay. *Redox Biology, 6*, 260-271. doi:10.1016/j.redox.2015.08.010
- Grabow, D., Kaiser, M., Hjorth, L., Byrne, J., Alessi, D., Allodji, R. S., . . . Kaatsch, P. (2018). The PanCareSurFup cohort of 83,333 five-year survivors of childhood cancer: a cohort from 12 European countries. *Eur J Epidemiol, 33*(3), 335-349. doi:10.1007/s10654-018-0370-3
- Gray, S. P., Shah, A. M., & Smyrnias, I. (2019). NADPH oxidase 4 and its role in the cardiovascular system. *Vasc Biol,* 1(1), H59-h66. doi:10.1530/vb-19-0014
- Greensmith, D. J., Eisner, D. A., & Nirmalan, M. (2010). The effects of hydrogen peroxide on intracellular calcium handling and contractility in the rat ventricular myocyte. *Cell Calcium*, 48(6), 341-351. doi:<u>https://doi.org/10.1016/j.ceca.2010.10.007</u>
- Greensmith, D. J. (2014). Ca analysis: an Excel based program for the analysis of intracellular calcium transients including multiple, simultaneous regression analysis. *Comput Methods Programs Biomed*, *113*(1), 241-250. doi:10.1016/j.cmpb.2013.09.004
- Greensmith, D. J., Galli, G. L. J., Trafford, A. W., & Eisner, D. A. (2014). Direct measurements of SR free Ca reveal the mechanism underlying the transient effects of RyR potentiation under physiological conditions. *Cardiovascular Research*, *103*(4), 554-563. doi:10.1093/cvr/cvu158
- Greenstein, J. L., Hinch, R., & Winslow, R. L. (2006). Mechanisms of excitation-contraction coupling in an integrative model of the cardiac ventricular myocyte. *Biophysical Journal*, *90*(1), 77-91. doi:10.1529/biophysj.105.065169
- Grenier, M. A., & Lipshultz, S. E. (1998). Epidemiology of anthracycline cardiotoxicity in children and adults. *Semin Oncol*, *25*(4 Suppl 10), 72-85.
- Grieve, D. J., & Shah, A. M. (2003). Oxidative stress in heart failure: More than just damage. *European Heart Journal, 24*(24), 2161-2163. doi:10.1016/j.ehj.2003.10.015
- Grynkiewicz, G., Poenie, M., & Tsien, R. Y. (1985). A new generation of Ca2+ indicators with greatly improved fluorescence properties. *J Biol Chem*, *260*(6), 3440-3450.

- Gupta, V., & Poss, K. D. (2012). Clonally dominant cardiomyocytes direct heart morphogenesis. *Nature*, 484(7395), 479-484. doi:http://dx.doi.org/10.1038/nature11045
- Habibalahi, A., Moghari, M. D., Campbell, J. M., Anwer, A. G., Mahbub, S. B., Gosnell, M., . . .
 Goldys, E. M. (2020). Non-invasive real-time imaging of reactive oxygen species (ROS) using auto-fluorescence multispectral imaging technique: A novel tool for redox biology. *Redox Biology*, 34, 101561. doi:<u>https://doi.org/10.1016/j.redox.2020.101561</u>
- Haikala, H., Kaivola, J., Nissinen, E., Wall, P., Levijoki, J., & Lindén, I. B. (1995). Cardiac troponin
 C as a target protein for a novel calcium sensitizing drug, levosimendan. J Mol Cell
 Cardiol, 27(9), 1859-1866. doi:10.1016/0022-2828(95)90009-8
- Haizlip, K. M., Hiranandani, N., Biesiadecki, B. J., & Janssen, P. M. (2012). Impact of hydroxyl radical-induced injury on calcium handling and myofilament sensitivity in isolated myocardium. J Appl Physiol (1985), 113(5), 766-774. doi:10.1152/japplphysiol.01424.2011
- Hajjar, R. J., & Leopold, J. A. (2006). Xanthine Oxidase Inhibition and Heart Failure. *Circulation Research*, *98*(2), 169-171. doi:doi:10.1161/01.RES.0000204557.09610.01
- Hanahan, D., & Weinberg, R. A. (2011). Hallmarks of cancer: the next generation. *Cell, 144*(5), 646-674. doi:10.1016/j.cell.2011.02.013
- Hangas, A., Aasumets, K., Kekäläinen, N. J., Paloheinä, M., Pohjoismäki, J. L., Gerhold, J. M., & Goffart, S. (2018). Ciprofloxacin impairs mitochondrial DNA replication initiation through inhibition of Topoisomerase 2. *Nucleic acids research, 46*(18), 9625. doi:10.1093/nar/gky793
- Hanna, A. D., Janczura, M., Cho, E., Dulhunty, A. F., & Beard, N. A. (2011). Multiple Actions of the Anthracycline Daunorubicin on Cardiac Ryanodine Receptors. *Molecular pharmacology*, 80(3), 538. doi:10.1124/mol.111.073478
- Hanna, A. D., Lam, A., Tham, S., Dulhunty, A. F., & Beard, N. A. (2014). Adverse effects of doxorubicin and its metabolic product on cardiac RyR2 and SERCA2A. *Molecular pharmacology*, 86(4), 438-449. doi:10.1124/mol.114.093849

- Hanna, A. D., Lam, A., Thekkedam, C., Willemse, H., Dulhunty, A. F., & Beard, N. A. (2017). The Anthracycline Metabolite Doxorubicinol Abolishes RyR2 Sensitivity to Physiological Changes in Luminal Ca²⁺ through an Interaction with Calsequestrin. *Molecular pharmacology*, *92*(5), 576. doi:10.1124/mol.117.108183
- Hanson, J., & Huxley, H. E. (1953). Structural Basis of the Cross-Striations in Muscle. *Nature*, *172*(4377), 530-532. doi:10.1038/172530b0
- Harbo, M. B., Nordén, E. S., Narula, J., Sjaastad, I., & Espe, E. K. S. (2020). Quantifying left ventricular function in heart failure: What makes a clinically valuable parameter? *Progress in Cardiovascular Diseases*. doi:<u>https://doi.org/10.1016/j.pcad.2020.05.007</u>
- Harrison, R. (2002). Structure and function of xanthine oxidoreductase: where are we now?
 Free Radical Biology and Medicine, 33(6), 774-797.
 doi:https://doi.org/10.1016/S0891-5849(02)00956-5
- Hasenfuss, G., & Teerlink, J. R. (2011). Cardiac inotropes: current agents and future directions. *European Heart Journal, 32*(15), 1838-1845. doi:10.1093/eurheartj/ehr026
- He, L., He, T., Farrar, S., Ji, L., Liu, T., & Ma, X. (2017). Antioxidants Maintain Cellular Redox
 Homeostasis by Elimination of Reactive Oxygen Species. *Cellular Physiology and Biochemistry*, 44(2), 532-553. doi:10.1159/000485089
- Heitzer, T., Schlinzig, T., Krohn, K., Meinertz, T., & Münzel, T. (2001). Endothelial dysfunction, oxidative stress, and risk of cardiovascular events in patients with coronary artery disease. *Circulation*, *104*(22), 2673-2678. doi:10.1161/hc4601.099485
- Henriksen, P. A. (2018). Anthracycline cardiotoxicity: an update on mechanisms, monitoring and prevention. *Heart*, 104(12), 971. doi:<u>http://dx.doi.org/10.1136/heartjnl-2017-312103</u>
- Hill, M. F., & Singal, P. K. (1996). Antioxidant and oxidative stress changes during heart failure subsequent to myocardial infarction in rats. *Am J Pathol*, *148*(1), 291-300.
- Hiranandani, N., & Janssen, P. M. (2009). Impact of Hydroxyl Radicals on Calcium Handling and Myofilament sensitivity in Isolated Myocardium. *The FASEB Journal*, 23(S1), 988.983-988.983. doi:https://doi.org/10.1096/fasebj.23.1 supplement.988.3

- Hobai, I. A., Maack, C., & O'Rourke, B. (2004). Partial Inhibition of Sodium/Calcium Exchange Restores Cellular Calcium Handling in Canine Heart Failure. *Circulation Research*, 95(3), 292-299. doi:doi:10.1161/01.RES.0000136817.28691.2d
- Hu, Z., Liang, M. C., & Soong, T. W. (2017). Alternative Splicing of L-type CaV1.2 Calcium Channels: Implications in Cardiovascular Diseases. *Genes*, 8(12), 344. Retrieved from <u>https://www.mdpi.com/2073-4425/8/12/344</u>
- Huang, J., Wu, R., Chen, L., Yang, Z., Yan, D., & Li, M. (2022). Understanding Anthracycline
 Cardiotoxicity From Mitochondrial Aspect. *Frontiers in Pharmacology*, 13.
 doi:10.3389/fphar.2022.811406
- Hucker, W. J., Sharma, V., Nikolski, V. P., & Efimov, I. R. (2007). Atrioventricular conduction with and without AV nodal delay: two pathways to the bundle of His in the rabbit heart. Am J Physiol Heart Circ Physiol, 293(2), H1122-1130. doi:10.1152/ajpheart.00115.2007
- Hullin, R., Khan, I. F. Y., Wirtz, S., Mohacsi, P., Varadi, G., Schwartz, A., & Herzig, S. (2003). Cardiac L-type calcium channel beta-subunits expressed in human heart have differential effects on single channel characteristics. *The Journal of biological chemistry*, 278(24), 21623. doi:10.1074/jbc.M211164200
- Huxley, A. F., & Niedergerke, R. (1954). Structural Changes in Muscle During Contraction: Interference Microscopy of Living Muscle Fibres. *Nature*, *173*(4412), 971-973. doi:10.1038/173971a0
- Hyun, S.-W., Kim, B.-R., Hyun, S.-A., & Seo, J.-W. (2017). The assessment of electrophysiological activity in human-induced pluripotent stem cell-derived cardiomyocytes exposed to dimethyl sulfoxide and ethanol by manual patch clamp and multi-electrode array system. *Journal of Pharmacological and Toxicological Methods*, 87, 93-98. doi:<u>https://doi.org/10.1016/j.vascn.2017.03.003</u>
- Ichikawa, Y., Ghanefar, M., Bayeva, M., Wu, R., Khechaduri, A., Naga Prasad, S. V., . . . Ardehali,
 H. (2014a). Cardiotoxicity of doxorubicin is mediated through mitochondrial iron accumulation. *J Clin Invest*, *124*(2), 617-630. doi:10.1172/jci72931

- Ichikawa, Y., Ghanefar, M., Bayeva, M., Wu, R., Khechaduri, A., Prasad, S. V. N., . . . Ardehali,
 H. (2014b). Cardiotoxicity of doxorubicin is mediated through mitochondrial iron accumulation. *The Journal of Clinical Investigation*, 124(2), 617-630. doi:10.1172/JCI72931
- Ide, T., Tsutsui, H., Kinugawa, S., Utsumi, H., Kang, D., Hattori, N., . . . Takeshita, A. (1999).
 Mitochondrial electron transport complex I is a potential source of oxygen free radicals in the failing myocardium. *Circ Res, 85*(4), 357-363. doi:10.1161/01.res.85.4.357
- Ikeda, S., Matsushima, S., Okabe, K., Ikeda, M., Ishikita, A., Tadokoro, T., . . . Tsutsui, H. (2019a). Blockade of L-type Ca2+ channel attenuates doxorubicin-induced cardiomyopathy via suppression of CaMKII-NF-κB pathway. *Scientific Reports, 9*(1), 9850. doi:10.1038/s41598-019-46367-6
- Ikeda, S., Matsushima, S., Okabe, K., Ikeda, M., Ishikita, A., Tadokoro, T., . . . Tsutsui, H. (2019b). Blockade of L-type Ca(2+) channel attenuates doxorubicin-induced cardiomyopathy via suppression of CaMKII-NF-κB pathway. *Sci Rep, 9*(1), 9850. doi:10.1038/s41598-019-46367-6
- Inaba, H., Greaves, M., & Mullighan, C. G. (2013). Acute lymphoblastic leukaemia. *Lancet, 381*(9881), 1943-1955. doi:10.1016/s0140-6736(12)62187-4
- Janssen, P. M. L., & Periasamy, M. (2007). Determinants of frequency-dependent contraction and relaxation of mammalian myocardium. *Journal of Molecular and Cellular Cardiology*, 43(5), 523-531. doi:<u>https://doi.org/10.1016/j.yjmcc.2007.08.012</u>
- Jarasch, E.-D., Grund, C., Bruder, G., Heid, H. W., Keenan, T. W., & Franke, W. W. (1981). Localization of xanthine oxidase in mammary-gland epithelium and capillary endothelium. *Cell*, *25*(1), 67-82. doi:<u>https://doi.org/10.1016/0092-8674(81)90232-4</u>
- Jensen, P. K. (1966). Antimycin-insensitive oxidation of succinate and reduced nicotinamideadenine dinucleotide in electron-transport particles. II. Steroid effects. *Biochim Biophys Acta, 122*(2), 167-174. doi:10.1016/0926-6593(66)90058-0

- Johnson, D. M., & Antoons, G. (2018). Arrhythmogenic Mechanisms in Heart Failure: Linking β-Adrenergic Stimulation, Stretch, and Calcium. *Frontiers in physiology*, *9*. doi:10.3389/fphys.2018.01453
- Jong, J. W. d., Meer, P. v. d., Nieukoop, A. S., Huizer, T., Stroeve, R. J., & Bos, E. (1990). Xanthine oxidoreductase activity in perfused hearts of various species, including humans. *Circulation Research*, *67*(3), 770-773. doi:doi:10.1161/01.RES.67.3.770
- Joukar, S. (2021). A comparative review on heart ion channels, action potentials and electrocardiogram in rodents and human: extrapolation of experimental insights to clinic. *Laboratory Animal Research*, *37*(1), 25. doi:10.1186/s42826-021-00102-3
- Kagan, V. E., Wipf, P., Stoyanovsky, D., Greenberger, J. S., Borisenko, G., Belikova, N. A., . . .
 Bayır, H. (2009). Mitochondrial targeting of electron scavenging antioxidants:
 Regulation of selective oxidation vs random chain reactions. *Advanced Drug Delivery Reviews*, *61*(14), 1375-1385. doi:<u>https://doi.org/10.1016/j.addr.2009.06.008</u>
- Kang, D. H., & Ha, S. K. (2014). Uric Acid Puzzle: Dual Role as Anti-oxidantand Pro-oxidant. *Electrolyte Blood Press, 12*(1), 1-6. doi:10.5049/ebp.2014.12.1.1
- Kattner, P., Strobel, H., Khoshnevis, N., Grunert, M., Bartholomae, S., Pruss, M., . . . Debatin,
 K.-M. (2019). Compare and contrast: pediatric cancer versus adult malignancies. *Cancer and Metastasis Reviews, 38*(4), 673-682. doi:10.1007/s10555-019-09836-y
- Kawakami, M., & Okabe, E. (1998). Superoxide anion radical-triggered Ca2+ release from cardiac sarcoplasmic reticulum through ryanodine receptor Ca2+ channel. *Molecular pharmacology*, 53(3), 497-503. doi:10.1124/mol.53.3.497
- Keith, M., Geranmayegan, A., Sole, M. J., Kurian, R., Robinson, A., Omran, A. S., & Jeejeebhoy,
 K. N. (1998). Increased Oxidative Stress in Patients With Congestive Heart Failure
 11This study was supported by a grant jointly sponsored by the Medical Research
 Council of Canada, Ottawa and Bayer Pharmaceuticals, Etobicoke, Ontario, Canada.
 Journal of the American College of Cardiology, 31(6), 1352-1356.
 doi:https://doi.org/10.1016/S0735-1097(98)00101-6
- Kendall, G. M., Little, M. P., Wakeford, R., Bunch, K. J., Miles, J. C., Vincent, T. J., . . . Murphy, M. F. (2013). A record-based case-control study of natural background radiation and 206

the incidence of childhood leukaemia and other cancers in Great Britain during 1980-2006. *Leukemia*, *27*(1), 3-9. doi:10.1038/leu.2012.151

- Keung, E. C., Toll, L., Ellis, M., & Jensen, R. A. (1991). L-type cardiac calcium channels in doxorubicin cardiomyopathy in rats morphological, biochemical, and functional correlations. *J Clin Invest*, 87(6), 2108-2113. doi:10.1172/jci115241
- Khan, M. A., D'Ovidio, A., Tran, H., & Palaniyar, N. (2019). Anthracyclines Suppress Both NADPH Oxidase- Dependent and -Independent NETosis in Human Neutrophils. *Cancers*, 11(9), 1328. doi:10.3390/cancers11091328
- Khan, M. A., Philip, L. M., Cheung, G., Vadakepeedika, S., Grasemann, H., Sweezey, N., & Palaniyar, N. (2018). Regulating NETosis: Increasing pH Promotes NADPH Oxidase Dependent NETosis. *Front Med (Lausanne), 5*, 19. doi:10.3389/fmed.2018.00019
- Kim, S.-Y., Kim, S.-J., Kim, B.-J., Rah, S.-Y., Chung, S. M., Im, M.-J., & Kim, U.-H. (2006). Doxorubicin-induced reactive oxygen species generation and intracellular Ca2+ increase are reciprocally modulated in rat cardiomyocytes. *Exp Mol Med*, *38*(5), 535-545. doi:10.1038/emm.2006.63
- King, M., Kingery, J., & Casey, B. (2012). Diagnosis and evaluation of heart failure. *Am Fam Physician*, 85(12), 1161-1168.
- Knudson, A. G., Jr. (1971). Mutation and cancer: statistical study of retinoblastoma. Proceedings of the National Academy of Sciences of the United States of America, 68(4), 820-823. doi:10.1073/pnas.68.4.820
- Kobza, C. (2021). Cardiac Toxicity: Using Angiotensin-Converting Enzyme Inhibitors to Prevent Anthracycline-Induced Left Ventricular Dysfunction and Cardiomyopathy. *Clin J Oncol Nurs*, 25(3), 259-266. doi:10.1188/21.Cjon.259-266
- Kosloski, L. M., Bales, I. K., Allen, K. B., Walker, B. L., Borkon, A. M., Stuart, R. S., . . . Wacker,
 M. J. (2009). Purification of cardiac myocytes from human heart biopsies for gene expression analysis. *American Journal of Physiology-Heart and Circulatory Physiology,* 297(3), H1163-H1169. doi:10.1152/ajpheart.00118.2009

- Kranias, E. G., & Hajjar, R. J. (2012). Modulation of Cardiac Contractility by the Phopholamban/SERCA2a Regulatome. *Circulation Research*, 110(12), 1646-1660. doi:doi:10.1161/CIRCRESAHA.111.259754
- Krijnen, P. A. J., Meischl, C., Hack, C. E., Meijer, C. J. L. M., Visser, C. A., Roos, D., & Niessen, H.
 W. M. (2003). Increased Nox2 expression in human cardiomyocytes after acute myocardial infarction. *Journal of Clinical Pathology, 56*(3), 194. doi:10.1136/jcp.56.3.194
- Kubo, H., Margulies, K. B., Piacentino, V., 3rd, Gaughan, J. P., & Houser, S. R. (2001). Patients with end-stage congestive heart failure treated with beta-adrenergic receptor antagonists have improved ventricular myocyte calcium regulatory protein abundance. *Circulation*, 104(9), 1012-1018. doi:10.1161/hc3401.095073
- Kuroda, J., Ago, T., Matsushima, S., Zhai, P., Schneider, M. D., & Sadoshima, J. (2010). NADPH oxidase 4 (Nox4) is a major source of oxidative stress in the failing heart. *Proceedings* of the National Academy of Sciences of the United States of America, 107(35), 15565-15570. doi:10.1073/pnas.1002178107
- Kuroda, J., & Sadoshima, J. (2010). NADPH Oxidase and Cardiac Failure. Journal of Cardiovascular Translational Research, 3(4), 314-320. doi:10.1007/s12265-010-9184 8
- Kuster, G. M., Lancel, S., Zhang, J., Communal, C., Trucillo, M. P., Lim, C. C., . . . Colucci, W. S. (2010). Redox-mediated reciprocal regulation of SERCA and Na+-Ca2+ exchanger contributes to sarcoplasmic reticulum Ca2+ depletion in cardiac myocytes. *Free Radic Biol Med*, 48(9), 1182-1187. doi:10.1016/j.freeradbiomed.2010.01.038
- Kuznetsov, A. V., Kehrer, I., Kozlov, A. V., Haller, M., Redl, H., Hermann, M., . . . Troppmair, J. (2011). Mitochondrial ROS production under cellular stress: comparison of different detection methods. *Analytical and Bioanalytical Chemistry*, 400(8), 2383-2390. doi:10.1007/s00216-011-4764-2
- Lacinová, L., & Hofmann, F. (2005). Ca 2+- and voltage-dependent inactivation of the expressed L-type Ca v1.2 calcium channel. *Archives of Biochemistry and Biophysics*, 437(1), 42-50. doi:10.1016/j.abb.2005.02.025

LaCombe, P., Tariq, M. A., & Lappin, S. L. (2022). Physiology, Afterload Reduction. In *StatPearls*. Treasure Island (FL): StatPearls Publishing

Copyright © 2022, StatPearls Publishing LLC.

- Laconi, E., Marongiu, F., & DeGregori, J. (2020). Cancer as a disease of old age: changing mutational and microenvironmental landscapes. *British Journal of Cancer, 122*(7), 943-952. doi:10.1038/s41416-019-0721-1
- Lederer, W., Cannell, M., Cohen, N., & Berlin, J. (1989). Excitation-contraction coupling in heart muscle. *An International Journal for Chemical Biology in Health and Disease*, *89*(2), 115-119. doi:10.1007/BF00220762
- Lee, J. A. (1981). Sydney Ringer (1834–1910) and Alexis Hartmann (1898–1964). *Anaesthesia,* 36(12), 1115-1121. doi:<u>https://doi.org/10.1111/j.1365-2044.1981.tb08698.x</u>
- Lee, J. A., & Allen, D. G. (1997). Calcium sensitisers: mechanisms of action and potential usefulness as inotropes. *Cardiovascular Research, 36*(1), 10-20. doi:10.1016/S0008-6363(97)00174-0
- Lee, K., Qian, D. Z., Rey, S., Wei, H., Liu, J. O., & Semenza, G. L. (2009). Anthracycline chemotherapy inhibits HIF-1 transcriptional activity and tumor-induced mobilization of circulating angiogenic cells.(MEDICAL SCIENCES)(Hypoxia-inducible factors)(Author abstract). *Proceedings of the National Academy of Sciences of the United States,* 106(7), 2353. Retrieved from

https://www.ncbi.nlm.nih.gov/pmc/articles/PMC2650160/pdf/zpq2353.pdf

- Lee, S., Grafweg, S., Schneider, T., Jimenez, M., Giacobino, J. P., Ghanem, A., . . . Brixius, K. (2010). Total beta-adrenoceptor deficiency results in cardiac hypotrophy and negative inotropy. *Physiol Res*, *59*(5), 679-689.
- Levine, M. (2000). Epirubicin in Breast Cancer: Present and Future. *Clinical Breast Cancer*, 1, S62-S67. doi:<u>https://doi.org/10.3816/CBC.2000.s.012</u>
- Lipscombe, D., Helton, T. D., & Xu, W. (2004). L-Type Calcium Channels: The Low Down. Journal of Neurophysiology, 92(5), 2633-2641. doi:10.1152/jn.00486.2004

- Lipshultz, S. E., Cochran, T. R., Franco, V. I., & Miller, T. L. (2013). Treatment-related cardiotoxicity in survivors of childhood cancer. *Nat Rev Clin Oncol, 10*(12), 697-710. doi:10.1038/nrclinonc.2013.195
- Lipshultz, S. E., Colan, S. D., Gelber, R. D., Perez-Atayde, A. R., Sallan, S. E., & Sanders, S. P. (1991). Late cardiac effects of doxorubicin therapy for acute lymphoblastic leukemia in childhood. *N Engl J Med*, *324*(12), 808-815. doi:10.1056/nejm199103213241205
- Lipshultz, S. E., Lipsitz, S. R., Sallan, S. E., Dalton, V. M., Mone, S. M., Gelber, R. D., & Colan, S.
 D. (2005). Chronic Progressive Cardiac Dysfunction Years After Doxorubicin Therapy for Childhood Acute Lymphoblastic Leukemia. *Journal of Clinical Oncology, 23*(12), 2629-2636. doi:10.1200/JCO.2005.12.121
- Lipshultz, S. E., Scully, R. E., Lipsitz, S. R., Sallan, S. E., Silverman, L. B., Miller, T. L., . . . Colan,
 S. D. (2010). Assessment of dexrazoxane as a cardioprotectant in doxorubicin-treated children with high-risk acute lymphoblastic leukaemia: long-term follow-up of a prospective, randomised, multicentre trial. *The Lancet. Oncology*, *11*(10), 950-961. doi:10.1016/S1470-2045(10)70204-7
- Liu, X., Wang, L., Cai, J., Liu, K., Liu, M., Wang, H., & Zhang, H. (2019). N-acetylcysteine alleviates H2O2-induced damage via regulating the redox status of intracellular antioxidants in H9c2 cells. *Int J Mol Med*, 43(1), 199-208. doi:10.3892/ijmm.2018.3962
- Llach, A., Mazevet, M., Mateo, P., Villejouvert, O., Ridoux, A., Rucker-Martin, C., . . . Gómez,
 A. M. (2019). Progression of excitation-contraction coupling defects in doxorubicin cardiotoxicity. *Journal of Molecular and Cellular Cardiology*, *126*, 129-139. doi:https://doi.org/10.1016/j.yjmcc.2018.11.019
- Lloyd-Jones, D. M., Larson, M. G., Leip, E. P., Beiser, A., D'Agostino, R. B., Kannel, W. B., . . .
 Levy, D. (2002). Lifetime risk for developing congestive heart failure: the Framingham
 Heart Study. *Circulation*, 106(24), 3068-3072.
 doi:10.1161/01.cir.0000039105.49749.6f
- Locke, F. S. (1895). Towards the Ideal Artificial Circulating Fluid for the Isolated Frog's Heart: Preliminary Communication. *The Journal of physiology, 18*(4), 332-333. doi:10.1113/jphysiol.1895.sp000571

- López, J. E., Jaradeh, K., Silva, E., Aminololama-Shakeri, S., & Simpson, P. C. (2017). A method to increase reproducibility in adult ventricular myocyte sizing and flow cytometry: Avoiding cell size bias in single cell preparations. *PLOS ONE, 12*(10), e0186792. doi:10.1371/journal.pone.0186792
- Louisse, J., Wüst, R. C. I., Pistollato, F., Palosaari, T., Barilari, M., Macko, P., . . . Prieto, P. (2017).
 Assessment of acute and chronic toxicity of doxorubicin in human induced pluripotent stem cell-derived cardiomyocytes. *Toxicol In Vitro, 42,* 182-190. doi:10.1016/j.tiv.2017.04.023
- Lytton, J., Westlin, M., Burk, S. E., Shull, G. E., & Maclennan, D. H. (1992). Functional comparisons between isoforms of the sarcoplasmic or endoplasmic reticulum family of calcium pumps. *The Journal of biological chemistry*, *267*(20), 14483. Retrieved from https://www.jbc.org/content/267/20/14483.full.pdf
- Ma, J., Wang, Y., Zheng, D., Wei, M., Xu, H., & Peng, T. (2013). Rac1 signalling mediates doxorubicin-induced cardiotoxicity through both reactive oxygen species-dependent and -independent pathways. *Cardiovascular Research, 97*(1), 77-87. doi:10.1093/cvr/cvs309
- Mackenzie, I. S., Ford, I., Nuki, G., Hallas, J., Hawkey, C. J., Webster, J., . . . Waller, T. (2020).
 Long-term cardiovascular safety of febuxostat compared with allopurinol in patients with gout (FAST): a multicentre, prospective, randomised, open-label, non-inferiority trial. *The Lancet*, *396*(10264), 1745-1757. doi:10.1016/S0140-6736(20)32234-0
- Maeda, A., Honda, M., Kuramochi, T., Tanaka, K., & Takabatake, T. (1997). An angiotensinconverting enzyme inhibitor protects against doxorubicin-induced impairment of calcium handling in neonatal rat cardiac myocytes. *Clin Exp Pharmacol Physiol, 24*(9-10), 720-726. doi:10.1111/j.1440-1681.1997.tb02119.x
- Mahmoudinezhad, M., Abbaszadeh, F., Zarezadeh, M., Bahreini, N., Jamilian, P., Jamilian, P., & Ostadrahimi, A. (2023). N-acetylecysteine, a powerful agent in the reinforcement of anti-oxidant profile: a systematic review and dose-response meta-analysis of controlled clinical trials. *Clinical Nutrition ESPEN*. doi:<u>https://doi.org/10.1016/j.clnesp.2023.01.021</u>

- Malik, U. Z., Hundley, N. J., Romero, G., Radi, R., Freeman, B. A., Tarpey, M. M., & Kelley, E. E. (2011). Febuxostat inhibition of endothelial-bound XO: implications for targeting vascular ROS production. *Free Radic Biol Med*, 51(1), 179-184. doi:10.1016/j.freeradbiomed.2011.04.004
- Mani, S. K., Egan, E. A., Addy, B. K., Grimm, M., Kasiganesan, H., Thiyagarajan, T., . . . Menick,
 D. R. (2010). beta-Adrenergic receptor stimulated Ncx1 upregulation is mediated via a
 CaMKII/AP-1 signaling pathway in adult cardiomyocytes. *Journal of Molecular and Cellular Cardiology*, 48(2), 342-351. doi:10.1016/j.yjmcc.2009.11.007
- Manni, M. E., Zazzeri, M., Musilli, C., Bigagli, E., Lodovici, M., & Raimondi, L. (2013). Exposure of cardiomyocytes to angiotensin II induces over-activation of monoamine oxidase type A: Implications in heart failure. *European Journal of Pharmacology*, 718(1), 271-276. doi:<u>https://doi.org/10.1016/j.ejphar.2013.08.022</u>

Manoil, D., & Bouillaguet, S. (2018). Oxidative Stress in Bacteria Measured by Flow Cytometry.

- Marcel, E., & Ernst, N. (2000). Paradoxical block of the Na+-Ca2+ exchanger by extracellular protons in guinea-pig ventricular myocytes. *The Journal of physiology, 523*(2), 353-366. doi:10.1111/j.1469-7793.2000.t01-1-00353.x
- Marinello, J., Delcuratolo, M., & Capranico, G. (2018). Anthracyclines as Topoisomerase II Poisons: From Early Studies to New Perspectives. *International journal of molecular sciences, 19*(11), 3480. doi:10.3390/ijms19113480
- Marston, S. B., & de Tombe, P. P. (2008). Troponin phosphorylation and myofilament Ca2+sensitivity in heart failure: Increased or decreased? *Journal of Molecular and Cellular Cardiology*, 45(5), 603-607. doi:<u>https://doi.org/10.1016/j.yjmcc.2008.07.004</u>
- Martins, D., Garcia, L. R., Queiroz, D. A. R., Lazzarin, T., Tonon, C. R., Balin, P. d. S., . . . Zornoff,
 L. (2022). Oxidative Stress as a Therapeutic Target of Cardiac Remodeling.
 Antioxidants, 11(12), 2371. Retrieved from https://www.mdpi.com/2076-3921/11/12/2371
- Marzilli, M., Crea, F., Morrone, D., Bonow, R. O., Brown, D. L., Camici, P. G., . . . Boden, W. E.
 (2020). Myocardial ischemia: From disease to syndrome. *International Journal of Cardiology*, *314*, 32-35. doi:10.1016/j.ijcard.2020.04.074

- McCloy, R. A., Rogers, S., Caldon, C. E., Lorca, T., Castro, A., & Burgess, A. (2014). Partial inhibition of Cdk1 in G 2 phase overrides the SAC and decouples mitotic events. *Cell cycle (Georgetown, Tex.), 13*(9), 1400-1412. doi:10.4161/cc.28401
- McGowan, J. V., Chung, R., Maulik, A., Piotrowska, I., Walker, J. M., & Yellon, D. M. (2017).
 Anthracycline Chemotherapy and Cardiotoxicity. *Cardiovasc Drugs Ther, 31*(1), 63-75.
 doi:10.1007/s10557-016-6711-0
- Menna, P., Salvatorelli, E., & Minotti, G. (2010). Anthracycline Degradation in Cardiomyocytes: A Journey to Oxidative Survival. *Chemical Research in Toxicology*, 23(1), 6-10. doi:10.1021/tx9003424
- Mijares, A., & López, J. R. (2001). I-carnitine prevents increase in diastolic [Ca2+] induced by doxorubicin in cardiac cells. *European Journal of Pharmacology, 425*(2), 117-120. doi:<u>https://doi.org/10.1016/S0014-2999(01)01158-X</u>
- Milani-Nejad, N., & Janssen, P. M. (2014). Small and large animal models in cardiac contraction research: advantages and disadvantages. *Pharmacol Ther*, 141(3), 235-249. doi:10.1016/j.pharmthera.2013.10.007
- Mitrofanova, L. B., Gorshkov, A. N., Konovalov, P. V., & Krylova, J. S. (2018). Telocytes in the human sinoatrial node. *Journal of Cellular and Molecular Medicine (Online)*, 22(1), 521-532. doi:<u>http://dx.doi.org/10.1111/jcmm.13340</u>
- Mittmann, C., Eschenhagen, T., & Scholz, H. (1998). Cellular and molecular aspects of contractile dysfunction in heart failure. *Cardiovascular Research*, 39(2), 267-275. doi:10.1016/S0008-6363(98)00139-4
- Miura, M., Taguchi, Y., Nagano, T., Sasaki, M., Handoh, T., & Shindoh, C. (2015). Effect of myofilament Ca(2+) sensitivity on Ca(2+) wave propagation in rat ventricular muscle.
 J Mol Cell Cardiol, 84, 162-169. doi:10.1016/j.yjmcc.2015.04.027
- Miura, Y., & Kimura, J. (1989). Sodium-calcium exchange current. Dependence on internal Ca and Na and competitive binding of external Na and Ca. *The Journal of General Physiology*, *93*(6), 1129-1145. doi:10.1085/jgp.93.6.1129
- Mohamed, T. M. A., Abou-Leisa, R., Baudoin, F., Stafford, N., Neyses, L., Cartwright, E. J., & Oceandy, D. (2013). Development and characterization of a novel fluorescent 213

indicator protein PMCA4-GCaMP2 in cardiomyocytes. *Journal of Molecular and Cellular Cardiology*, *63*, 57-68. doi:<u>https://doi.org/10.1016/j.yjmcc.2013.07.007</u>

- Mollova, M., Bersell, K., Walsh, S., Savla, J., Das, L. T., Park, S. Y., . . . Kühn, B. (2013).
 Cardiomyocyte proliferation contributes to heart growth in young humans.
 Proceedings of the National Academy of Sciences of the United States of America, 110(4), 1446-1451. doi:10.1073/pnas.1214608110
- Moris, D., Spartalis, M., Spartalis, E., Karachaliou, G.-S., Karaolanis, G. I., Tsourouflis, G., . . .
 Theocharis, S. (2017a). The role of reactive oxygen species in the pathophysiology of cardiovascular diseases and the clinical significance of myocardial redox. *Annals of Translational Medicine, 5*(16), 326. Retrieved from https://atm.amegroups.com/article/view/16040
- Moris, D., Spartalis, M., Spartalis, E., Karachaliou, G. S., Karaolanis, G. I., Tsourouflis, G., . . .
 Theocharis, S. (2017b). The role of reactive oxygen species in the pathophysiology of cardiovascular diseases and the clinical significance of myocardial redox. *Ann Transl Med*, *5*(16), 326. doi:10.21037/atm.2017.06.27
- Morris, T. E., & Sulakhe, P. V. (1997). Sarcoplasmic reticulum Ca(2+)-pump dysfunction in rat cardiomyocytes briefly exposed to hydroxyl radicals. *Free Radic Biol Med*, 22(1-2), 37-47. doi:10.1016/s0891-5849(96)00238-9
- Munro, D., & Pamenter, M. E. (2019). Comparative studies of mitochondrial reactive oxygen species in animal longevity: Technical pitfalls and possibilities. *Aging Cell, 18*(5), e13009. doi:<u>https://doi.org/10.1111/acel.13009</u>
- Munshi, N. V. (2012). Gene Regulatory Networks in Cardiac Conduction System Development. *Circulation Research, 110*(11), 1525-1537. doi:doi:10.1161/CIRCRESAHA.111.260026
- Münzel, T., Camici, G. G., Maack, C., Bonetti, N. R., Fuster, V., & Kovacic, J. C. (2017). Impact of Oxidative Stress on the Heart and Vasculature: Part 2 of a 3-Part Series. *Journal of the American College of Cardiology, 70*(2), 212-229. doi:<u>https://doi.org/10.1016/j.jacc.2017.05.035</u>

- Muthuramalingam, S., Gale, J., & Bradbury, J. (2013). Dexrazoxane efficacy for anthracycline extravasation: use in UK clinical practice. *International Journal of Clinical Practice*, *67*(3), 244-249. doi:<u>https://doi.org/10.1111/ijcp.12103</u>
- Nabeebaccus, A., Zhang, M., & Shah, A. M. (2011). NADPH oxidases and cardiac remodelling. *Heart Failure Reviews*, *16*(1), 5-12. doi:10.1007/s10741-010-9186-2
- Nakai, J., Adams, B. A., Imoto, K., & Beam, K. G. (1994). Critical roles of the S3 segment and S3-S4 linker of repeat I in activation of L-type calcium channels. *Proceedings of the National Academy of Sciences PNAS, 91*(3), 1014-1018. doi:10.1073/pnas.91.3.1014
- Nakayama, T., Oshima, Y., Kusumoto, S., Yamamoto, J., Osaga, S., Fujinami, H., . . . Iida, S. (2020). Clinical features of anthracycline-induced cardiotoxicity in patients with malignant lymphoma who received a CHOP regimen with or without rituximab: A single-center, retrospective observational study. *EJHaem*, 1(2), 498-506. doi:10.1002/jha2.110
- Nathan, P. C., Amir, E., & Abdel-Qadir, H. (2016). Cardiac Outcomes in Survivors of Pediatric and Adult Cancers. *Canadian Journal of Cardiology*, *32*(7), 871-880. doi:10.1016/j.cjca.2016.02.065
- Nerbonne, J. M., & Kass, R. S. (2005). Molecular physiology of cardiac repolarization. *Physiol Rev, 85*(4), 1205-1253. doi:10.1152/physrev.00002.2005
- NHS. (2019). Overview: Cancer. Retrieved from https://www.nhs.uk/conditions/cancer/
- Nickel, A. G., Kohlhaas, M., Bertero, E., Wilhelm, D., Wagner, M., Sequeira, V., . . . Maack, C. (2020). CaMKII does not control mitochondrial Ca2+ uptake in cardiac myocytes. *The Journal of physiology*, *598*(7), 1361-1376. doi:<u>https://doi.org/10.1113/JP276766</u>
- Nisimoto, Y., Jackson, H. M., Ogawa, H., Kawahara, T., & Lambeth, J. D. (2010). Constitutive NADPH-dependent electron transferase activity of the Nox4 dehydrogenase domain. *Biochemistry, 49*(11), 2433-2442. doi:10.1021/bi9022285
- Nogrady, B. (2020). How cancer genomics is transforming diagnosis and treatment. *Nature, 579*(7800), S10-s11. doi:10.1038/d41586-020-00845-4
- Nomura, J., Busso, N., Ives, A., Tsujimoto, S., Tamura, M., So, A., & Yamanaka, Y. (2013). Febuxostat, an inhibitor of xanthine oxidase, suppresses lipopolysaccharide-induced 215

MCP-1 production via MAPK phosphatase-1-mediated inactivation of JNK. *PLOS ONE,* 8(9), e75527. doi:10.1371/journal.pone.0075527

- Nordberg, J., & Arnér, E. S. J. (2001). Reactive oxygen species, antioxidants, and the mammalian thioredoxin system1 1This review is based on the licentiate thesis "Thioredoxin reductase—interactions with the redox active compounds 1-chloro-2,4dinitrobenzene and lipoic acid" by Jonas Nordberg, 2001, Karolinska Institute, Stockholm, ISBN 91-631-1064-4. *Free Radical Biology and Medicine, 31*(11), 1287-1312. doi:<u>https://doi.org/10.1016/S0891-5849(01)00724-9</u>
- Noreng, S., Ota, N., Sun, Y., Ho, H., Johnson, M., Arthur, C. P., . . . Koerber, J. T. (2022). Structure of the core human NADPH oxidase NOX2. *Nature Communications, 13*(1), 6079. doi:10.1038/s41467-022-33711-0
- Noujaim, S. F., Lucca, E., Muñoz, V., Persaud, D., Berenfeld, O., Meijler, F. L., & Jalife, J. (2004).
 From mouse to whale: A universal scaling relation for the PR interval of the electrocardiogram of mammals. *Circulation, 110*(18), 2802-2808. doi:10.1161/01.CIR.0000146785.15995.67
- Núñez-Acosta, E., & Sobie, Eric A. (2014). The Ryanodine Receptor Patchwork: Knitting Calcium Spark Dynamics. *Biophysical Journal, 107*(12), 2749-2750. doi:<u>https://doi.org/10.1016/j.bpj.2014.10.068</u>
- Oeffinger, K. C., Mertens, A. C., Sklar, C. A., Kawashima, T., Hudson, M. M., Meadows, A. T., .
 . Robison, L. L. (2006). Chronic health conditions in adult survivors of childhood cancer. *N Engl J Med*, 355(15), 1572-1582. doi:10.1056/NEJMsa060185
- Olson, R. D., Gambliel, H. A., Vestal, R. E., Shadle, S. E., Charlier, H. A., & Cusack, B. J. (2005). Doxorubicin cardiac dysfunction. *Cardiovascular Toxicology*, *5*(3), 269-283. doi:10.1385/CT:5:3:269
- Olson, R. D., Li, X., Palade, P., Shadle, S. E., Mushlin, P. S., Gambliel, H. A., . . . Cusack, B. J. (2000). Sarcoplasmic Reticulum Calcium Release Is Stimulated and Inhibited by Daunorubicin and Daunorubicinol. *Toxicology and Applied Pharmacology*, *169*(2), 168-176. doi:<u>https://doi.org/10.1006/taap.2000.9065</u>

- Ondrias, K., Borgatta, L., Kim, D. H., & Ehrlich, B. E. (1990). Biphasic effects of doxorubicin on the calcium release channel from sarcoplasmic reticulum of cardiac muscle. *Circ Res, 67*(5), 1167-1174. doi:10.1161/01.res.67.5.1167
- Oort, R. J. v., Garbino, A., Wang, W., Dixit, S. S., Landstrom, A. P., Gaur, N., . . . Wehrens, X. H.
 T. (2011). Disrupted Junctional Membrane Complexes and Hyperactive Ryanodine
 Receptors After Acute Junctophilin Knockdown in Mice. *Circulation, 123*(9), 979-988.
 doi:doi:10.1161/CIRCULATIONAHA.110.006437
- Panieri, E., Gogvadze, V., Norberg, E., Venkatesh, R., Orrenius, S., & Zhivotovsky, B. (2013). Reactive oxygen species generated in different compartments induce cell death, survival, or senescence. *Free Radical Biology and Medicine*, *57*, 176-187. doi:https://doi.org/10.1016/j.freeradbiomed.2012.12.024
- Pantazis, A., Savalli, N., Sigg, D., Neely, A., & Olcese, R. (2014). Functional heterogeneity of the four voltage sensors of a human L-type calcium channel. *Proceedings of the National Academy of Sciences of the United States of America*, 111(51), 18381-18386. doi:10.1073/pnas.1411127112
- Pedre, B., Barayeu, U., Ezeriņa, D., & Dick, T. P. (2021). The mechanism of action of N-acetylcysteine (NAC): The emerging role of H2S and sulfane sulfur species.
 Pharmacology & *Therapeutics*, 228, 107916.
 doi:https://doi.org/10.1016/j.pharmthera.2021.107916
- Pelzmann, B., Schaffer, P., Bernhart, E., Lang, P., Mächler, H., Rigler, B., & Koidl, B. (1998). Ltype calcium current in human ventricular myocytes at a physiological temperature from children with tetralogy of Fallot. *Cardiovasc Res, 38*(2), 424-432. doi:10.1016/s0008-6363(98)00002-9
- Peng, W., Shen, H., Wu, J., Guo, W., Pan, X., Wang, R., & Chen, S. (2016). Structural basis for the gating mechanism of the type 2 ryanodine receptor RyR2. *Science*, 354(6310), n/a. doi:10.1126/science.aah5324
- Peoples, J. N., Saraf, A., Ghazal, N., Pham, T. T., & Kwong, J. Q. (2019). Mitochondrial dysfunction and oxidative stress in heart disease. *Experimental & Molecular Medicine*, 51(12), 1-13. doi:10.1038/s12276-019-0355-7

- Pérez, N. G., Gao, W. D., & Marbán, E. (1998). Novel myofilament Ca2+-sensitizing property of xanthine oxidase inhibitors. *Circ Res, 83*(4), 423-430. doi:10.1161/01.res.83.4.423
- Pesola, F., Ferlay, J., & Sasieni, P. (2017). Cancer incidence in English children, adolescents and young people: past trends and projections to 2030. *British Journal of Cancer*, 117(12), 1865-1873. doi:10.1038/bjc.2017.341
- Peterson, B. Z., Demaria, C. D., & Yue, D. T. (1999). Calmodulin Is the Ca 2+ Sensor for Ca 2+ Dependent Inactivation of L-Type Calcium Channels. *Neuron*, 22(3), 549-558.
 doi:10.1016/S0896-6273(00)80709-6
- Piacentino, V., Weber, C. R., Chen, X., Weisser-Thomas, J., Margulies, K. B., Bers, D. M., & Houser, S. R. (2003). Cellular Basis of Abnormal Calcium Transients of Failing Human Ventricular Myocytes. *Circulation Research, 92*(6), 651-658. doi:doi:10.1161/01.RES.0000062469.83985.9B
- Pinali, C., Bennett, H., Davenport, J. B., Trafford, A. W., & Kitmitto, A. (2013). Three-Dimensional Reconstruction of Cardiac Sarcoplasmic Reticulum Reveals a Continuous Network Linking Transverse-Tubules. *Circulation Research*, *113*(11), 1219-1230. doi:doi:10.1161/CIRCRESAHA.113.301348
- Pitt, G. S., Zühlke, R. D., Hudmon, A., Schulman, H., Reuter, H., & Tsien, R. W. (2001). Molecular
 Basis of Calmodulin Tethering and Ca 2+ -dependent Inactivation of L-type Ca 2+
 Channels. *Journal of Biological Chemistry, 276*(33), 30794-30802.
 doi:10.1074/jbc.M104959200
- Pitts, B. J. (1979). Stoichiometry of sodium-calcium exchange in cardiac sarcolemmal vesicles.
 Coupling to the sodium pump. *The Journal of biological chemistry*, 254(14), 6232.
 Retrieved from https://www.jbc.org/content/254/14/6232.full.pdf
- Ponikowski, P., Voors, A. A., Anker, S. D., Bueno, H., Cleland, J. G. F., Coats, A. J. S., . . . Group,
 E. S. D. (2016). 2016 ESC Guidelines for the diagnosis and treatment of acute and chronic heart failure: The Task Force for the diagnosis and treatment of acute and chronic heart failure of the European Society of Cardiology (ESC)Developed with the special contribution of the Heart Failure Association (HFA) of the ESC. *European Heart Journal*, *37*(27), 2129-2200. doi:10.1093/eurheartj/ehw128

- Rao, Y. J., & Xi, L. (2009). Pivotal effects of phosphodiesterase inhibitors on myocyte contractility and viability in normal and ischemic hearts. *Acta Pharmacol Sin, 30*(1), 1-24. doi:10.1038/aps.2008.1
- Rattanasopa, C., Kirk, J. A., Bupha-Intr, T., Papadaki, M., de Tombe, P. P., & Wattanapermpool,
 J. (2019). Estrogen but not testosterone preserves myofilament function from doxorubicin-induced cardiotoxicity by reducing oxidative modifications. *Am J Physiol Heart Circ Physiol*, 316(2), H360-h370. doi:10.1152/ajpheart.00428.2018
- Ray, P. D., Huang, B.-W., & Tsuji, Y. (2012). Reactive oxygen species (ROS) homeostasis and redox regulation in cellular signaling. *Cellular signalling*, 24(5), 981-990. doi:10.1016/j.cellsig.2012.01.008
- Reeves, J. P., Bailey, C. A., & Hale, C. C. (1986). Redox modification of sodium-calcium exchange activity in cardiac sarcolemmal vesicles. *J Biol Chem*, *261*(11), 4948-4955.
- Reis, J., Massari, M., Marchese, S., Ceccon, M., Aalbers, F. S., Corana, F., . . . Mattevi, A. (2020).
 A closer look into NADPH oxidase inhibitors: Validation and insight into their mechanism of action. *Redox Biol, 32*, 101466. doi:10.1016/j.redox.2020.101466
- Reulen, R. C., Winter, D. L., Frobisher, C., Lancashire, E. R., Stiller, C. A., Jenney, M. E., . . .
 Hawkins, M. M. (2010). Long-term cause-specific mortality among survivors of childhood cancer. *JAMA*, *304*(2), 172. doi:10.1001/jama.2010.923
- Reuter, H., Henderson, S. A., Han, T., Ross, R. S., Goldhaber, J. I., & Philipson, K. D. (2002). The Na+-Ca2+ exchanger is essential for the action of cardiac glycosides. *Circ Res, 90*(3), 305-308. doi:10.1161/hh0302.104562
- Ringer, S. (1883). A third contribution regarding the Influence of the Inorganic Constituents of the Blood on the Ventricular Contraction. *The Journal of physiology, 4*(2-3), 222. doi:10.1113/jphysiol.1883.sp000127
- Ringer, S. (1885). Regarding the Effect of the Saline Ingredients of the Blood on the Contraction of the Heart. *British Medical Journal,* 1(1267), 731. doi:http://dx.doi.org/10.1136/bmj.1.1267.731
- Robertson, S. P., Johnson, J. D., Holroyde, M. J., Kranias, E. G., Potter, J. D., & Solaro, R. J. (1982). The effect of troponin I phosphorylation on the Ca2+-binding properties of the 219

Ca2+-regulatory site of bovine cardiac troponin. *J Biol Chem, 257*(1), 260-263. Retrieved from <u>https://www.jbc.org/content/257/1/260.full.pdf</u>

- Rossi, D., Barone, V., Giacomello, E., Cusimano, V., & Sorrentino, V. (2008). The Sarcoplasmic Reticulum: An Organized Patchwork of Specialized Domains. *Traffic, 9*(7), 1044-1049. doi:10.1111/j.1600-0854.2008.00717.x
- Rousseau, E., & Pinkos, J. (1990). pH modulates conducting and gating behaviour of single calcium release channels. *Pflügers Archiv*, *415*(5), 645-647. doi:10.1007/BF02583520
- Sag, C. M., Köhler, A. C., Anderson, M. E., Backs, J., & Maier, L. S. (2011). CaMKII-dependent SR Ca leak contributes to doxorubicin-induced impaired Ca handling in isolated cardiac myocytes. *Journal of Molecular and Cellular Cardiology*, *51*(5), 749-759. doi:<u>https://doi.org/10.1016/j.yjmcc.2011.07.016</u>
- Sag, C. M., Wolff, H. A., Neumann, K., Opiela, M. K., Zhang, J., Steuer, F., . . . Maier, L. S. (2013).
 Ionizing radiation regulates cardiac Ca handling via increased ROS and activated
 CaMKII. *Basic Res Cardiol, 108*(6), 385. doi:10.1007/s00395-013-0385-6
- Saito, H., Tanaka, K., Iwasaki, T., Oda, A., Watanabe, S., Kanno, M., . . . Kazama, J. J. (2021).
 Xanthine oxidase inhibitors are associated with reduced risk of cardiovascular disease.
 Scientific Reports, 11(1), 1380. doi:10.1038/s41598-020-80835-8
- Sarvazyan, N. (1996). Visualization of doxorubicin-induced oxidative stress in isolated cardiac myocytes. Am J Physiol, 271(5 Pt 2), H2079-2085. doi:10.1152/ajpheart.1996.271.5.H2079
- Sato, D., & Bers, Donald M. (2011). How Does Stochastic Ryanodine Receptor-Mediated Ca
 Leak Fail to Initiate a Ca Spark? *Biophysical Journal, 101*(10), 2370-2379.
 doi:<u>https://doi.org/10.1016/j.bpj.2011.10.017</u>
- Sato, D., Dixon, R. E., Santana, L. F., & Navedo, M. F. (2018). A model for cooperative gating of L-type Ca2+ channels and its effects on cardiac alternans dynamics. *PLoS Comput Biol*, 14(1), e1005906. doi:10.1371/journal.pcbi.1005906
- Sato, D., Shannon, Thomas R., & Bers, Donald M. (2016). Sarcoplasmic Reticulum Structure and Functional Properties that Promote Long-Lasting Calcium Sparks. *Biophysical Journal*, 110(2), 382-390. doi:<u>https://doi.org/10.1016/j.bpj.2015.12.009</u>

- Sawyer, D. B., Fukazawa, R., Arstall, M. A., & Kelly, R. A. (1999). Daunorubicin-Induced Apoptosis in Rat Cardiac Myocytes Is Inhibited by Dexrazoxane. *Circulation Research*, 84(3), 257-265. doi:doi:10.1161/01.RES.84.3.257
- Scherer, N. M., & Deamer, D. W. (1986). Oxidative stress impairs the function of sarcoplasmic reticulum by oxidation of sulfhydryl groups in the Ca2+-ATPase. Arch Biochem Biophys, 246(2), 589-601. doi:10.1016/0003-9861(86)90314-0
- Schillinger, W., Fiolet, J. W., Schlotthauer, K., & Hasenfuss, G. (2003). Relevance of Na+–Ca2+ exchange in heart failure. *Cardiovascular Research*, 57(4), 921-933. doi:10.1016/s0008-6363(02)00826-x
- Schmidt, A. G., Edes, I., & Kranias, E. G. (2001). Phospholamban: A Promising Therapeutic
 Target in Heart Failure? *Cardiovasc Drugs Ther*, 15(5), 387-396.
 doi:10.1023/A:1013381204658
- Schmiegelow, K., Forestier, E., Hellebostad, M., Heyman, M., Kristinsson, J., Söderhäll, S., . . .
 Oncology. (2010). Long-term results of NOPHO ALL-92 and ALL-2000 studies of childhood acute lymphoblastic leukemia. *Leukemia*, 24(2), 345-354. doi:10.1038/leu.2009.251
- Scholz, H. (1984). Inotropic drugs and their mechanisms of action. *Journal of the American College of Cardiology*, 4(2), 389-397. doi:<u>https://doi.org/10.1016/S0735-1097(84)80231-4</u>
- Schwan, J., & Campbell, S. G. (2015). Article Commentary: Prospects for In Vitro Myofilament
 Maturation in Stem Cell-Derived Cardiac Myocytes:Supplementary Issue: Stem Cell
 Biology. *Biomarker Insights, 10s1*, BMI.S23912. doi:10.4137/bmi.S23912
- Semenza, G. L. (2007). Evaluation of HIF-1 inhibitors as anticancer agents. *Drug Discovery Today*, *12*(19), 853-859. doi:<u>https://doi.org/10.1016/j.drudis.2007.08.006</u>
- Shadle, S. E., Bammel, B. P., Cusack, B. J., Knighton, R. A., Olson, S. J., Mushlin, P. S., & Ison, R.
 D. O. (2000). Daunorubicin cardiotoxicity: Evidence for the importance of the quinone moiety in a free-radical-independent mechanism. *Biochemical Pharmacology, 60*(10), 1435-1444. doi:<u>https://doi.org/10.1016/S0006-2952(00)00458-5</u>

- Shaffer, F., McCraty, R., & Zerr, C. L. (2014). A healthy heart is not a metronome: an integrative review of the heart's anatomy and heart rate variability. *Frontiers in psychology, 5*, 1040-1040. doi:10.3389/fpsyg.2014.01040
- Shanmugavadivel, D., Liu, J. F., Ball-Gamble, A., Polanco, A., Vedhara, K., Walker, D., & Ojha,
 S. (2022). The Childhood Cancer Diagnosis (CCD) Study: a UK observational study to
 describe referral pathways and quantify diagnostic intervals in children and young
 people with cancer. *BMJ Open*, *12*(2), e058744. doi:10.1136/bmjopen-2021-058744
- Sheibani, M., Azizi, Y., Shayan, M., Nezamoleslami, S., Eslami, F., Farjoo, M. H., & Dehpour, A.
 R. (2022). Doxorubicin-Induced Cardiotoxicity: An Overview on Pre-clinical Therapeutic Approaches. *Cardiovascular Toxicology*, 22(4), 292-310. doi:10.1007/s12012-022-09721-1
- Shen, W.-J., Hsieh, C.-Y., Chen, C.-L., Yang, K.-C., Ma, C.-T., Choi, P.-C., & Lin, C.-F. (2013). A modified fixed staining method for the simultaneous measurement of reactive oxygen species and oxidative responses. *Biochemical and Biophysical Research Communications*, 430(1), 442-447. doi:<u>https://doi.org/10.1016/j.bbrc.2012.11.037</u>

Shih, H. T. (1994). Anatomy of the action potential in the heart. *Tex Heart Inst J, 21*(1), 30-41.

- Sipido, K. R., Stankovicova, T., Flameng, W., Vanhaecke, J., & Verdonck, F. (1998). Frequency dependence of Ca2+ release from the sarcoplasmic reticulum in human ventricular myocytes from end-stage heart failure. *Cardiovascular Research, 37*(2), 478-488. doi:10.1016/s0008-6363(97)00280-0
- Skubitz, K. M., Blaes, A. H., Konety, S. H., & Francis, G. S. (2017). Cardiac safety profile of patients receiving high cumulative doses of pegylated-liposomal doxorubicin: use of left ventricular ejection fraction is of unproven value. *Cancer Chemother Pharmacol,* 80(4), 787-798. doi:10.1007/s00280-017-3420-8
- Smittenaar, C. R., Petersen, K. A., Stewart, K., & Moitt, N. (2016). Cancer incidence and mortality projections in the UK until 2035. *British journal of cancer*, *115*(9), 1147-1155. doi:10.1038/bjc.2016.304
- Solaro, R. J., & Rarick, H. M. (1998). Troponin and Tropomyosin. *Circulation Research, 83*(5), 471-480. doi:doi:10.1161/01.RES.83.5.471

- Souza, C., Mônico, D. A., & Tedesco, A. C. (2020). Implications of dichlorofluorescein photoinstability for detection of UVA-induced oxidative stress in fibroblasts and keratinocyte cells. *Photochemical & Photobiological Sciences, 19*(1), 40-48. doi:10.1039/c9pp00415g
- Spurgeon, H. A., duBell, W. H., Stern, M. D., Sollott, S. J., Ziman, B. D., Silverman, H. S., . . . Lakatta, E. G. (1992). Cytosolic calcium and myofilaments in single rat cardiac myocytes achieve a dynamic equilibrium during twitch relaxation. *The Journal of physiology*, 447, 83-102. doi:10.1113/jphysiol.1992.sp018992
- Stewart, S., MacIntyre, K., Capewell, S., & McMurray, J. J. (2003). Heart failure and the aging population: an increasing burden in the 21st century? *Heart, 89*(1), 49-53. doi:10.1136/heart.89.1.49
- Sugamura, K., & Keaney, J. F. (2011a). Reactive oxygen species in cardiovascular disease. *Free Radical Biology and Medicine,* 51(5), 978-992. doi:<u>https://doi.org/10.1016/j.freeradbiomed.2011.05.004</u>
- Sugamura, K., & Keaney, J. F., Jr. (2011b). Reactive oxygen species in cardiovascular disease. *Free Radic Biol Med*, *51*(5), 978-992. doi:10.1016/j.freeradbiomed.2011.05.004
- Sun, Y.-B., & Irving, M. (2009). The molecular basis of the steep force–calcium relation in heart muscle. Journal of Molecular and Cellular Cardiology, 48, 859-865. doi:10.1016/j.yjmcc.2009.11.019
- Szekeres, F. L. M., Walum, E., Wikström, P., & Arner, A. (2021a). A small molecule inhibitor of Nox2 and Nox4 improves contractile function after ischemia-reperfusion in the mouse heart. *Sci Rep, 11*(1), 11970. doi:10.1038/s41598-021-91575-8
- Szekeres, F. L. M., Walum, E., Wikström, P., & Arner, A. (2021b). A small molecule inhibitor of Nox2 and Nox4 improves contractile function after ischemia–reperfusion in the mouse heart. *Scientific Reports*, 11(1), 11970. doi:10.1038/s41598-021-91575-8
- Tabet, F., Savoia, C., Schiffrin, E. L., & Touyz, R. M. (2004). Differential Calcium Regulation by
 Hydrogen Peroxide and Superoxide in Vascular Smooth Muscle Cells from
 Spontaneously Hypertensive Rats. *Journal of Cardiovascular Pharmacology, 44*(2).
 Retrieved

223

https://journals.lww.com/cardiovascularpharm/Fulltext/2004/08000/Differential Ca lcium Regulation by Hydrogen.9.aspx

- Takeshima, H., Komazaki, S., Nishi, M., Iino, M., & Kangawa, K. (2000). Junctophilins: A Novel
 Family of Junctional Membrane Complex Proteins. *Molecular Cell*, 6(1), 11-22.
 doi:10.1016/S1097-2765(05)00005-5
- Tanaka, Y., Nagoshi, T., Yoshii, A., Oi, Y., Takahashi, H., Kimura, H., . . . Yoshimura, M. (2021).
 Xanthine oxidase inhibition attenuates doxorubicin-induced cardiotoxicity in mice. *Free Radical Biology and Medicine, 162,* 298-308.
 doi:https://doi.org/10.1016/j.freeradbiomed.2020.10.303
- Tebbi, C. K., London, W. B., Friedman, D., Villaluna, D., De Alarcon, P. A., Constine, L. S., ...
 Schwartz, C. L. (2007). Dexrazoxane-Associated Risk for Acute Myeloid
 Leukemia/Myelodysplastic Syndrome and Other Secondary Malignancies in Pediatric
 Hodgkin's Disease. *Journal of Clinical Oncology*, 25(5), 493-500.
 doi:10.1200/JCO.2005.02.3879
- Temma, K., Chugun, A., Akera, T., Kondo, H., & Kurebayashi, N. (1996). Doxorubicin alters Ca(2+) transients but fails to change Ca(2+) sensitivity of contractile proteins. *Environ Toxicol Pharmacol, 1*(2), 131-139. doi:10.1016/1382-6689(95)00021-6
- Tew, K. D. (2012). Editorial. *Biochemical Pharmacology, 83*(8), 985-986. doi:<u>https://doi.org/10.1016/j.bcp.2011.11.023</u>
- Thayer, W. S. (1986). Role of catalase in metabolism of hydrogen peroxide by the perfused rat heart. *FEBS Lett, 202*(1), 137-140. doi:10.1016/0014-5793(86)80664-0
- Timolati, F., Ott, D., Pentassuglia, L., Giraud, M. N., Perriard, J. C., Suter, T. M., & Zuppinger,
 C. (2006). Neuregulin-1 beta attenuates doxorubicin-induced alterations of excitation contraction coupling and reduces oxidative stress in adult rat cardiomyocytes. *J Mol Cell Cardiol, 41*(5), 845-854. doi:10.1016/j.yjmcc.2006.08.002
- Tisdale, J. E., Patel, R., Webb, C. R., Borzak, S., & Zarowitz, B. J. (1995). Electrophysiologic and proarrhythmic effects of intravenous inotropic agents. *Prog Cardiovasc Dis, 38*(2), 167-180. doi:10.1016/s0033-0620(05)80005-2

- Toyoshima, C., Nakasako, M., Nomura, H., & Ogawa, H. (2000). Crystal structure of the calcium pump of sarcoplasmic reticulum at 2.6 Angstrom resolution. *Nature, 405*(6787), 647-655. doi:10.1038/35015017
- Trachtenberg, B., Landy, D., Franco, V., Henkel, J., Pearson, E., Miller, T., & Lipshultz, S. (2011). Anthracycline-Associated Cardiotoxicity in Survivors of Childhood Cancer. *Pediatric Cardiology*, 32(3), 342-353. doi:10.1007/s00246-010-9878-3
- Trafford, A. W., Díaz, M. E., & Eisner, D. A. (1999a). A novel, rapid and reversible method to measure Ca buffering and time-course of total sarcoplasmic reticulum Ca content in cardiac ventricular myocytes. *Pflügers Archiv, 437*(3), 501-503. doi:10.1007/s004240050808
- Trafford, A. W., Díaz, M. E., & Eisner, D. A. (1999b). A novel, rapid and reversible method to measure Ca buffering and time-course of total sarcoplasmic reticulum Ca content in cardiac ventricular myocytes. *Pflugers Arch, 437*(3), 501-503. doi:10.1007/s004240050808
- Trafford, A. W., Díaz, M. E., Negretti, N., & Eisner, D. A. (1997). Enhanced Ca²⁺ Current and Decreased Ca²⁺ Efflux Restore Sarcoplasmic Reticulum Ca²⁺ Content After Depletion. *Circulation Research, 81*(4), 477-484. doi:doi:10.1161/01.RES.81.4.477
- Trafford, A. W., Díaz, M. E., Sibbring, G. C., & Eisner, D. A. (2000). Modulation of CICR has no maintained effect on systolic Ca2+: simultaneous measurements of sarcoplasmic reticulum and sarcolemmal Ca2+ fluxes in rat ventricular myocytes. *The Journal of physiology, 522*(2), 259-270. doi:<u>https://doi.org/10.1111/j.1469-7793.2000.t01-2-</u> <u>00259.x</u>
- Triposkiadis, F., Xanthopoulos, A., & Butler, J. (2019). Cardiovascular Aging and Heart Failure: JACC Review Topic of the Week. *Journal of the American College of Cardiology*, 74(6), 804-813. doi:<u>https://doi.org/10.1016/j.jacc.2019.06.053</u>
- Tscheschner, H., Meinhardt, E., Schlegel, P., Jungmann, A., Lehmann, L. H., Müller, O. J., . . . Raake, P. W. (2019). CaMKII activation participates in doxorubicin cardiotoxicity and

is attenuated by moderate GRP78 overexpression. *PLOS ONE, 14*(4), e0215992-e0215992. doi:10.1371/journal.pone.0215992

- Ukai, T., Cheng, C., Tachibana, H., Igawa, A., Zhang, Z.-s., Cheng, H.-J., & Little, W. C. (2001).
 Allopurinol Enhances the Contractile Response to Dobutamine and Exercise in Dogs
 With Pacing-Induced Heart Failure. *Circulation: Journal of the American Heart* Association, 103, 750-755.
- Unverferth, B. J., Magorien, R. D., Balcerzak, S. P., Leier, C. V., & Unverferth, D. V. (1983). Early changes in human myocardial nuclei after doxorubicin. *Cancer, 52*(2), 215-221. doi:10.1002/1097-0142(19830715)52:2<215::aid-cncr2820520206>3.0.co;2-f
- Ushio-Fukai, M., Zafari, A. M., Fukui, T., Ishizaka, N., & Griendling, K. K. (1996). p22phox is a critical component of the superoxide-generating NADH/NADPH oxidase system and regulates angiotensin II-induced hypertrophy in vascular smooth muscle cells. *J Biol Chem*, 271(38), 23317-23321. doi:10.1074/jbc.271.38.23317
- Valdés, Á., Treuer, A. V., Barrios, G., Ponce, N., Fuentealba, R., Dulce, R. A., & González, D. R. (2018). NOX Inhibition Improves β-Adrenergic Stimulated Contractility and Intracellular Calcium Handling in the Aged Rat Heart. *International journal of molecular sciences, 19*(8). doi:10.3390/ijms19082404
- Valdivia, H. H., Kaplan, J. H., Ellis-Davies, G. C. R., & Lederer, W. J. (1995). Rapid adaptation of cardiac ryanodine receptors: modulation by Mg2+ and phosphorylation. (magnesium). *Science*, 267(5206), 1997. doi:10.1126/science.7701323
- van Der Velden, J., Klein, L. J., Zaremba, R., Boontje, N. M., Huybregts, M. A., Stooker, W., . . . Stienen, G. J. (2001). Effects of calcium, inorganic phosphate, and pH on isometric force in single skinned cardiomyocytes from donor and failing human hearts. *Circulation, 104*(10), 1140-1146. doi:10.1161/hc3501.095485
- van der Velden, J., Papp, Z., Zaremba, R., Boontje, N. M., de Jong, J. W., Owen, V. J., ... Stienen,
 G. J. (2003). Increased Ca2+-sensitivity of the contractile apparatus in end-stage human heart failure results from altered phosphorylation of contractile proteins. *Cardiovasc Res*, *57*(1), 37-47. doi:10.1016/s0008-6363(02)00606-5

- Vanburen, P., & Palmer, B. M. (2010). Cooperative activation of the cardiac myofilament: the pivotal role of tropomyosin. *Circulation*, 121(3), 351-353. doi:10.1161/CIR.0b013e3181d0b947
- Varro, A., Negretti, N., Hester, S. B., & Eisner, D. A. (1993). An estimate of the calcium content of the sarcoplasmic reticulum in rat ventricular myocytes. *Pflügers Archiv*, 423(1), 158-160. doi:10.1007/BF00374975
- Vassalle, C., Sciarrino, R., Bianchi, S., Battaglia, D., Mercuri, A., & Maffei, S. (2012). Sex-related differences in association of oxidative stress status with coronary artery disease.
 Fertility and Sterility, 97(2), 414-419.e412.
 doi:https://doi.org/10.1016/j.fertnstert.2011.11.045
- Vavrova, A., Jansova, H., Mackova, E., Machacek, M., Haskova, P., Tichotova, L., . . . Simunek,
 T. (2013). Catalytic inhibitors of topoisomerase II differently modulate the toxicity of anthracyclines in cardiac and cancer cells. *PLOS ONE, 8*(10), e76676-e76676. doi:10.1371/journal.pone.0076676
- Ventura-Clapier, R., Garnier, A., Veksler, V., & Joubert, F. (2011). Bioenergetics of the failing heart. *Biochim Biophys Acta*, *1813*(7), 1360-1372. doi:10.1016/j.bbamcr.2010.09.006
- Vincent, D. T., Ibrahim, Y. F., Espey, M. G., & Suzuki, Y. J. (2013). The role of antioxidants in the era of cardio-oncology. *Cancer Chemother Pharmacol*, 72(6), 1157-1168. doi:10.1007/s00280-013-2260-4

Vincent, J. L. (2008). Understanding cardiac output. Crit Care, 12(4), 174. doi:10.1186/cc6975

- Volkova, M., & Russell, R., 3rd. (2011). Anthracycline cardiotoxicity: prevalence, pathogenesis and treatment. *Current cardiology reviews*, 7(4), 214-220. doi:10.2174/157340311799960645
- Wagner, S., Ruff, H. M., Weber, S. L., Bellmann, S., Sowa, T., Schulte, T., . . . Maier, L. S. (2011).
 Reactive oxygen species-activated Ca/calmodulin kinase IIδ is required for late I(Na) augmentation leading to cellular Na and Ca overload. *Circ Res, 108*(5), 555-565. doi:10.1161/circresaha.110.221911
- Walker, D. A. (2021). Helping GPs to diagnose children's cancer. *Br J Gen Pract,* 71(705), 151-152. doi:10.3399/bjgp21X715241

- Wang, M.-C., Dolphin, A., & Kitmitto, A. (2004). L-type voltage-gated calcium channels: understanding function through structure. *FEBS Letters*, 564(3), 245-250. doi:https://doi.org/10.1016/S0014-5793(04)00253-4
- Wang, S., Li, Y., Song, X., Wang, X., Zhao, C., Chen, A., & Yang, P. (2015). Febuxostat pretreatment attenuates myocardial ischemia/reperfusion injury via mitochondrial apoptosis. *Journal of Translational Medicine*, *13*(1), 209. doi:10.1186/s12967-015-0578-x
- Wehner, G. J., Jing, L., Haggerty, C. M., Suever, J. D., Leader, J. B., Hartzel, D. N., . . . Fornwalt,
 B. K. (2019). Routinely reported ejection fraction and mortality in clinical practice:
 where does the nadir of risk lie? *European Heart Journal*, 41(12), 1249-1257.
 doi:10.1093/eurheartj/ehz550
- Weiss, R. B., Sarosy, G., Clagett-Carr, K., Russo, M., & Leyland-Jones, B. (1986). Anthracycline analogs: the past, present, and future. *Cancer Chemother Pharmacol*, 18(3), 185-197. doi:10.1007/bf00273384
- Whitaker, R. H. (2018). Anatomy of the heart. *Medicine*, *46*(8), 423-426. doi:10.1016/j.mpmed.2018.05.010
- White, M. C., Holman, D. M., Boehm, J. E., Peipins, L. A., Grossman, M., & Jane Henley, S. (2014). Age and Cancer Risk: A Potentially Modifiable Relationship. *American Journal of Preventive Medicine*, 46(3, Supplement 1), S7-S15. doi:<u>https://doi.org/10.1016/j.amepre.2013.10.029</u>
- White, S. M., Constantin, P. E., & Claycomb, W. C. (2004). Cardiac physiology at the cellular level: use of cultured HL-1 cardiomyocytes for studies of cardiac muscle cell structure and function. *American Journal of Physiology-Heart and Circulatory Physiology*, 286(3), H823-H829. doi:10.1152/ajpheart.00986.2003
- White, W. B. (2018). Gout, Xanthine Oxidase Inhibition, and Cardiovascular Outcomes. *Circulation, 138*(11), 1127-1129. doi:doi:10.1161/CIRCULATIONAHA.118.036148
- White, W. B., Saag, K. G., Becker, M. A., Borer, J. S., Gorelick, P. B., Whelton, A., . . .
 Gunawardhana, L. (2018). Cardiovascular Safety of Febuxostat or Allopurinol in
 Patients with Gout. N Engl J Med, 378(13), 1200-1210. doi:10.1056/NEJMoa1710895
 228

- Wilcken, D. E. L. (2012). Physiology of the normal heart. *Surgery (Oxford), 30*(1), 1-4. doi:https://doi.org/10.1016/j.mpsur.2011.10.005
- Wind, S., Beuerlein, K., Eucker, T., Müller, H., Scheurer, P., Armitage, M., . . . Wingler, K. (2010a). Comparative pharmacology of chemically distinct NADPH oxidase inhibitors. British Journal of Pharmacology, 161(4), 885-898. doi:<u>https://doi.org/10.1111/j.1476-5381.2010.00920.x</u>
- Wind, S., Beuerlein, K., Eucker, T., Müller, H., Scheurer, P., Armitage, M. E., . . . Wingler, K. (2010b). Comparative pharmacology of chemically distinct NADPH oxidase inhibitors.
 Br J Pharmacol, 161(4), 885-898. doi:10.1111/j.1476-5381.2010.00920.x
- Winslow, R. L., Tanskanen, A., Chen, M., & Greenstein, J. L. (2006). Multiscale Modeling of Calcium Signaling in the Cardiac Dyad. *Annals of the New York Academy of Sciences*, 1080(1), 362-375. doi:10.1196/annals.1380.027
- Winterbourn, C. C., Kettle, A. J., & Hampton, M. B. (2016). Reactive Oxygen Species and Neutrophil Function. *Annual Review of Biochemistry*, 85(1), 765-792. doi:10.1146/annurev-biochem-060815-014442
- Wolff, M. R., Buck, S. H., Stoker, S. W., Greaser, M. L., & Mentzer, R. M. (1996). Myofibrillar calcium sensitivity of isometric tension is increased in human dilated cardiomyopathies: role of altered beta-adrenergically mediated protein phosphorylation. J Clin Invest, 98(1), 167-176. doi:10.1172/jci118762
- Xu, L., Chen, J., Li, X.-Y., Ren, S., Huang, C.-X., Wu, G., . . Jiang, X.-J. (2012). Analysis of Na+/Ca2+ exchanger (NCX) function and current in murine cardiac myocytes during heart failure. *Molecular Biology Reports, 39*(4), 3847-3852. doi:10.1007/s11033-011-1163-x
- Yamamoto, J. F., & Goodman, M. T. (2008). Patterns of leukemia incidence in the United States by subtype and demographic characteristics, 1997-2002. *Cancer Causes Control*, 19(4), 379-390. doi:10.1007/s10552-007-9097-2
- Yousefian, M., Hosseinzadeh, H., Hayes, A. W., Hadizadeh, F., & Karimi, G. (2021). The Protective Effect of Natural Compounds on Doxorubicin-Induced Cardiotoxicity via

Nicotinamide Adenine Dinucleotide Phosphate Oxidase Inhibition. *Journal of Pharmacy and Pharmacology*, 74(3), 351-359. doi:10.1093/jpp/rgab109

- Yuchi, Z., Lau, K., & Van Petegem, F. (2012). Disease mutations in the ryanodine receptor central region: crystal structures of a phosphorylation hot spot domain. *Structure*, 20(7), 1201-1211. doi:10.1016/j.str.2012.04.015
- Zahradníková, A., & Zahradník, I. (2012). Construction of calcium release sites in cardiac myocytes. *Frontiers in physiology, 3*, 322-322. doi:10.3389/fphys.2012.00322
- Zhang, G. Q., & Zhang, W. (2009). Heart rate, lifespan, and mortality risk. *Ageing Research Reviews, 8*(1), 52-60. doi:<u>https://doi.org/10.1016/j.arr.2008.10.001</u>
- Zhao, L., & Zhang, B. (2017). Doxorubicin induces cardiotoxicity through upregulation of death receptors mediated apoptosis in cardiomyocytes. *Scientific Reports*, 7, 44735. doi:10.1038/srep44735
- Zheng, J., Lee, H. C. M., bin Sattar, M. M., Huang, Y., & Bian, J.-S. (2011). Cardioprotective effects of epigallocatechin-3-gallate against doxorubicin-induced cardiomyocyte injury. *European Journal of Pharmacology, 652*(1), 82-88. doi:<u>https://doi.org/10.1016/j.ejphar.2010.10.082</u>
- Zhihao, L., Jingyu, N., Lan, L., Michael, S., Rui, G., Xiyun, B., . . . Guanwei, F. (2020). SERCA2a:
 a key protein in the Ca(2+) cycle of the heart failure. *Heart Fail Rev, 25*(3), 523-535.
 doi:10.1007/s10741-019-09873-3
- Zhou, S., Palmeira, C. M., & Wallace, K. B. (2001). Doxorubicin-induced persistent oxidative stress to cardiac myocytes. *Toxicology Letters*, 121(3), 151-157. doi:<u>https://doi.org/10.1016/S0378-4274(01)00329-0</u>
- Zhou, X., Sheng, Y., Yang, R., & Kong, X. (2010). Nicotine Promotes Cardiomyocyte Apoptosis via Oxidative Stress and Altered Apoptosis-Related Gene Expression. *Cardiology*, 115(4), 243-250. doi:10.1159/000301278
- Zorov, D. B., Juhaszova, M., & Sollott, S. J. (2006). Mitochondrial ROS-induced ROS release: An update and review. *Biochimica et Biophysica Acta (BBA) - Bioenergetics*, 1757(5), 509-517. doi:<u>https://doi.org/10.1016/j.bbabio.2006.04.029</u>

- Zühlke, R. D., Pitt, G. S., Deisseroth, K., Tsien, R. W., & Reuter, H. (1999). Calmodulin supports both inactivation and facilitation of L-type calcium channels. *Nature, 399*(6732), 159. doi:10.1038/20200
- Zühlke, R. D., & Reuter, H. (1998). Ca2+-sensitive inactivation of L-type Ca2+ channels depends on multiple cytoplasmic amino acid sequences of the alpha1C subunit. *Proceedings of the National Academy of Sciences of the United States of America,* 95(6), 3287-3294. doi:10.1073/pnas.95.6.3287
- Zuna, J., Madzo, J., Krejci, O., Zemanova, Z., Kalinova, M., Muzikova, K., . . . Trka, J. (2011). ETV6/RUNX1 (TEL/AML1) is a frequent prenatal first hit in childhood leukemia. *Blood*, *117*(1), 368-369. doi:10.1182/blood-2010-09-309070